UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT
Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): April 28, 2023

IVERIC bio, Inc.

(Exact name of registrant as specified in its charter)

Delaware	001-36080	20-8185347
(State or other jurisdiction	(Commission	(I.R.S. Employer
of incorporation)	File Number)	Identification No.)
	8 Sylvan Way Parsippany, NJ 07054	
(Addre	ss of Principal Executive Offices) (Z	ip Code)
Registrant's tel	ephone number, including area code:	(609) 474-6755
Check the appropriate box below if the Form 8-K filing is following provisions (<i>see</i> General Instruction A.2. below):	intended to simultaneously satisfy the	e filing obligation of the registrant under any of the
☐ Written communications pursuant to Rule 425 under the	ne Securities Act (17 CFR 230.425)	
⊠ Soliciting material pursuant to Rule 14a-12 under the l	Exchange Act (17 CFR 240.14a-12)	
☐ Pre-commencement communications pursuant to Rule	14d-2(b) under the Exchange Act (1	7 CFR 240.14d-2(b))
☐ Pre-commencement communications pursuant to Rule	13e-4(c) under the Exchange Act (17	7 CFR 240.13e-4(c))
Securities registered pursuant to Section 12(b) of the Act:		
Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$0.001 per share	ISEE	The Nasdaq Global Select Market
Indicate by check mark whether the registrant is an emergin chapter) or Rule 12b-2 of the Securities Exchange Act of 1 ☐ Emerging Growth Company ☐ If an emerging growth company, indicate by check any new or revised financial accounting standards	934 (§240.12b-2 of this chapter). k mark if the registrant has elected r	not to use the extended transition period for complying with

Item 1.01 Entry into a Material Definitive Agreement.

On April 28, 2023, IVERIC bio, Inc. (the "<u>Company</u>") entered into an Agreement and Plan of Merger (the "<u>Merger Agreement</u>") with Astellas US Holding, Inc., a Delaware corporation ("<u>Parent</u>"), Berry Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("<u>Merger Sub</u>"), and solely as provided by Section 8.10(b) of the Merger Agreement, Astellas Pharma Inc., a company organized under the laws of Japan ("<u>Guarantor</u>"), providing for the merger of Merger Sub with and into the Company (the "<u>Merger</u>"), with the Company surviving the Merger as a wholly owned subsidiary of Parent. Capitalized terms used herein and not otherwise defined herein have the meanings set forth in the Merger Agreement.

At the Effective Time of the Merger, each share of common stock, par value \$0.001 per share, of the Company issued and outstanding as of immediately prior to the Effective Time (other than Excluded Shares and Dissenting Shares) will be cancelled and automatically converted into the right to receive cash in an amount equal to \$40.00, without interest (the "Merger Consideration") and subject to any withholding of taxes. The Merger Agreement requires Guarantor to guarantee the covenants, obligations (including payment obligations) and liabilities applicable to Parent, Merger Sub or the Surviving Corporation, as applicable, under the Merger Agreement.

Consummation of the Merger is subject to customary closing conditions, including, without limitation, the absence of certain legal impediments, no Material Adverse Effect having occurred since the signing of the Merger Agreement, the expiration or termination of the required waiting periods under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and any other Specified Antitrust Authority, solely to the extent the approval of any such authority is required in connection with the Merger, and approval by the Company's stockholders. The parties expect the Merger and the other transactions contemplated by the Merger Agreement to close in the third calendar quarter of 2023.

The Company has made customary representations and warranties in the Merger Agreement and has agreed to customary covenants regarding the operation of the business of the Company and its Subsidiaries prior to the Effective Time. The Company is also subject to customary restrictions on its ability to solicit alternative acquisition proposals from third parties and to provide non-public information to, and participate in discussions and engage in negotiations with, third parties regarding alternative acquisition proposals, with customary exceptions to allow the Board of Directors to exercise its fiduciary duties.

The Merger Agreement contains certain termination rights for the Company and Parent. Subject to certain limitations, the Company or Parent may terminate the Merger Agreement if the Merger is not consummated by midnight Eastern Time, on October 27, 2023, subject to an automatic 90-day extension in the event that all conditions other than antitrust approval are not obtained or waived as of such date and a second automatic 90-day extension thereafter in the event that all conditions other than the required antitrust approvals have been obtained or waived as of such extended date (the "End Date").

Upon termination of the Merger Agreement, under specified circumstances, the Company will be required to pay Parent a termination fee of \$222,370,000. Such circumstances include where the Merger Agreement is terminated (i) in connection with the Company accepting a Superior Offer approved by the Board of Directors, (ii) due to the Board of Directors' change or withdrawal of, or failure to reaffirm, its recommendation of the Merger, or (iii) because the Board of Directors or the Company intentionally breach their non-solicit obligations under the Merger Agreement in any material respects. This termination fee will also be payable if the Merger Agreement is terminated because the Company's stockholders did not vote to adopt the Merger Agreement, the Merger is not consummated before the End Date, or the Company breaches its representations, warranties or covenants in a manner that would cause the related closing conditions to not be met, and prior to any such termination, a proposal to acquire at least 50% of the Company's stock or assets is communicated to the Board of Directors or publicly disclosed and the Company enters into an agreement for, or consummates, a transaction contemplated by such proposal within one year of termination.

At the Effective Time, each Company Option that is then outstanding and unexercised, whether or not vested and which has a per share exercise price that is less than the Merger Consideration (each, an "In the Money Option"), will be cancelled and converted into the right to receive a cash payment equal to (A) the excess of (x) the Merger Consideration over (y) the exercise price payable per share under such In the Money Option, multiplied by (B) the total number of shares subject to such In the Money Option immediately prior to the Effective Time (without regard to vesting). In addition, at the Effective Time, each Company Option other than an In the Money Option that is then outstanding and unexercised, whether or not vested, will be cancelled with no consideration payable in respect thereof. At the Effective Time, each then outstanding Company RSU will be canceled and the holder thereof will be entitled to receive a cash payment equal to the product of (x) the Merger Consideration and (y) the number of shares subject to such Company RSU. At the Effective Time, each then outstanding Company PSU will be canceled and converted into a cash-based award, which will entitle the holder thereof to receive a cash payment equal to the product of (x) the Merger Consideration, and (y) the number of shares subject to such Company PSU, subject to the same terms and conditions (including vesting, forfeiture and acceleration provisions) that were applicable to the corresponding Company PSU immediately prior to the Effective Time.

The representations, warranties and covenants of the Company contained in the Merger Agreement have been made solely for the benefit of Parent and Merger Sub. In addition, such representations, warranties and covenants (i) have been made only for purposes of the Merger Agreement, (ii) have been qualified by (a) subject to certain terms and conditions, matters specifically disclosed in the Company's filings with the United States Securities and Exchange Commission ("SEC") prior to the date of the Merger Agreement and (b) confidential disclosures made to Parent and Merger Sub in the disclosure letter delivered in connection with the Merger Agreement, (iii) are subject to materiality qualifications contained in the Merger Agreement which may differ from what may be viewed as material by investors, and (iv) have been included in the Merger Agreement for the purpose of allocating risk between the contracting parties rather than establishing matters as fact. Accordingly, the Merger Agreement is included with this filing only to provide investors with information regarding the terms of the Merger Agreement, and not to provide investors with any other factual information regarding the Company or its business.

Investors should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of the Company or any of its subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures. The Merger Agreement should not be read alone, but should instead be read in conjunction with the other information regarding the Company that is or will be contained in, or incorporated by reference into, the Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and other documents that the Company files with the SEC.

The foregoing description of the Merger Agreement and the transactions contemplated thereby does not purport to be complete and is subject to, and qualified in its entirety by, the full text of the Merger Agreement a copy of which is attached hereto as Exhibit 2.1, and the terms of which are incorporated herein by reference.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

On April 28, 2023, the Board approved an amendment to the Company's existing Amended and Restated Bylaws (the "Bylaws") to add a new Article V, Section 9 forum selection provision (the "Forum Selection Amendment").

The Forum Selection Amendment provides that, unless the Company consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware will be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Company; (ii) any action asserting a claim of breach of a duty (including any fiduciary duty) owed by any current or former director, officer, stockholder, employee or agent of the Company or the Company's stockholders, (iii) any action asserting a claim against the Company or any current or former director, officer, stockholder, employee or agent of the Company arising out of or relating to any provision of the General Corporation Law of the State of Delaware, the Certificate of Incorporation of the Company or the Bylaws (each, as in effect from time to time), or (iv) any action asserting a claim against the Company or any current or former director, officer, stockholder, employee or agent of the Company governed by the internal affairs doctrine of the State of Delaware; provided, however, that, in the event that the Court of Chancery of the State of Delaware lacks subject matter jurisdiction over any such action or proceeding, the sole and exclusive forum for such action or proceeding will be another state or federal court located within the State of Delaware, in each such case, unless the Court of Chancery (or such other state or federal court located within the State of Delaware, as applicable) has dismissed a prior action by the same plaintiff asserting the same claims because such court lacked personal jurisdiction over an indispensable party named as a defendant therein.

This summary is qualified in its entirety by reference to the Forum Selection Amendment, which is filed as Exhibit 3.1 hereto and incorporated by reference herein.

Item 8.01 Other Events.

On April 30, 2023, the Company and Guarantor issued a joint press release announcing entry into the Merger Agreement, a copy of which is attached as Exhibit 99.1 to the Current Report on Form 8-K and is incorporated herein by reference.

Item 9.01. Financial Statements and Exhibits

(d) Exhibits.

Exhibit No.	Description
<u>2.1</u>	Agreement and Plan of Merger, dated as of April 28, 2023, by and among Astellas US Holding, Inc., Berry Merger Sub, Inc., IVERIC bio, Inc. and, solely as provided by Section 8.10(b) therein, Astellas Pharma Inc.*
<u>3.1</u>	Amendment to IVERIC bio, Inc.'s Amended and Restated Bylaws, dated April 28, 2023
<u>99.1</u>	Joint Press Release issued by IVERIC bio, Inc. and Astellas Pharma Inc. on April 30, 2023

^{*} Schedules have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Company hereby undertakes to furnish supplemental copies of any of the omitted schedules upon request by the U.S. Securities and Exchange Commission; provided, that the Company may request confidential treatment pursuant to Rule 24b-2 of the Securities Exchange Act of 1934, as amended, for any schedules so furnished.

* * *

Forward Looking Statements

All statements in this Current Report on Form 8-K, other than statements of historical fact, are statements that could be deemed "forward-looking statements." In some cases, forward-looking statements may be identified by terminology such as "believe," "may," "will," "should", "predict", "goal", "strategy", "potentially," "estimate," "continue," "anticipate," "intend," "could," "would," "project," "plan," "expect," "seek" and similar expressions and variations thereof. The Company intends these forward-looking statements to be covered by the safe harbor provisions for forward-looking statements in the U.S. Private Securities Litigation Reform Act of 1995.

This Current Report on Form 8-K contains "forward-looking statements" relating to, among other things, the proposed acquisition of the Company, by way of a merger of a subsidiary of Guarantor (together with its subsidiaries, "Astellas") with and into the Company and the objectives of such proposed acquisition, Astellas' and the Company's beliefs and expectations regarding the potential benefits sought to be achieved by Astellas' proposed acquisition of the Company, the potential effects of the proposed acquisition on both Astellas and the Company, the expected benefits and success of the Company's product candidates, the anticipated timing for approval of ACP, the anticipated financing of the proposed acquisition, and the anticipated timing of completion of the proposed acquisition, each of which involves substantial risks and uncertainties that could cause actual results to differ materially from those expressed or implied by such statements.

Risks and uncertainties include, among other things, risks related to the ability of the Company and Astellas to complete the transactions contemplated by the Merger Agreement; the satisfaction or waiver of the conditions to closing the proposed acquisition set forth in the Merger Agreement (including the failure to obtain necessary regulatory approvals and failure to obtain the requisite vote by the Company's stockholders) in the anticipated timeframe or at all, including the possibility that the proposed acquisition does not close; the timing and nature of regulatory filings for the Company's product candidates, and the possibility of a termination of the Merger Agreement; the possibility that competing offers to acquire the Company may be made; risks related to the ability to realize the anticipated benefits of the proposed acquisition, including the possibility that the expected benefits from the acquisition will not be realized or will not be realized within the expected time period; the risk that the Company's business and products will not be integrated with those of Astellas successfully; the effects of disruption from the transactions contemplated by the Merger Agreement on the Company's business and the fact that the announcement and pendency of the transactions may make it more difficult to establish or maintain relationships with employees, suppliers and other business partners; negative effects of this announcement or the consummation of the proposed acquisition on the market price of Astellas' or the Company's common stock and/or operating results; significant transaction costs; unknown liabilities; the risk of litigation and/or regulatory actions related to the proposed acquisition or the Company's business; risks related to the financing of the acquisition; other business effects and uncertainties, including the effects of industry, market, business, economic, political or regulatory conditions; future exchange and interest rates; changes in tax and other laws, regulations, rates and policies; future business combinations or disposals; the uncertainties inherent in research and development, including the ability to meet anticipated clinical endpoints, commencement and/or completion dates for clinical trials, regulatory submission dates, regulatory approval dates and/or launch dates, as well as the possibility of unfavorable new clinical data and further analyses of existing clinical data; risks associated with interim data; the risk that clinical trial data is subject to differing interpretations and assessments by regulatory authorities; whether regulatory authorities will be satisfied with the design of and results from the clinical studies; whether and when drug applications may be filed in any jurisdictions for the Company's pipeline products; whether and when any such applications may be approved by regulatory authorities, which will depend on myriad factors, including making a determination as to whether the product's benefits outweigh its known risks and determination of the product's efficacy and, if approved, whether any such products will be commercially successful; decisions by regulatory authorities impacting labeling, manufacturing processes, safety or other matters that could affect the availability or commercial potential of such products; expectations for personnel and human capital matters; and competitive developments.

Moreover, Astellas and the Company operate in very competitive and rapidly changing environments, and new risks emerge from time to time. Astellas and the Company have based these forward-looking statements on their current expectations and projections about future events and trends that they believe may affect the financial condition, results of operations, business strategy, short-term and long-term business operations and objectives and financial needs of Astellas and the Company, but they cannot guarantee future events, results, actions, levels of activity, performance or achievements, business and market conditions, the timing and results of biotechnology development and potential regulatory approval. The foregoing factors are not exhaustive. You should also carefully consider other risks and uncertainties that may affect the business of the Company, including those described in the "Forward-Looking Statements", "Summary of Principal Risk Factors", and "Risk Factors" sections of the Company's Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and other documents filed from time to time with the SEC, all of which are available on the SEC's website at www.sec.gov. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements. Forward-looking statements speak only as of the date they are made. Readers are cautioned not to put undue reliance on forward-looking statements and Astellas and the Company assume no obligation to, and do not intend to, update or revise these forward-looking statements, whether as a result of new information, future events, or otherwise, unless required by applicable law.

Additional Information and Where to Find It

In connection with the proposed acquisition, the Company will be filing documents with the SEC, including preliminary and definitive proxy statements relating to the proposed acquisition. This document is not a substitute for the proxy statement or any other document which the Company may file with the SEC. The definitive proxy statement will be mailed to the Company's stockholders in connection with the proposed acquisition. This Current Report on Form 8-K is not a substitute for the proxy statement or any other document that may be filed by the Company with the SEC. BEFORE MAKING ANY VOTING DECISION, THE COMPANY'S INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE PRELIMINARY AND DEFINITIVE PROXY STATEMENTS AND ANY OTHER DOCUMENTS TO BE FILED WITH THE SEC IN CONNECTION WITH THE PROPOSED TRANSACTION OR INCORPORATED BY REFERENCE IN THE PROXY STATEMENT WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED ACQUISITION. Any vote in respect of resolutions to be proposed at the Company's stockholder meeting to approve the proposed transaction or other responses in relation to the proposed transaction should be made only on the basis of the information contained in the Company's proxy statement. Investors and security holders may obtain free copies of these documents (when they are available) and other related documents filed with the SEC at the SEC's web site at www.sec.gov, and all documents filed by the Company with the SEC are available to all stockholders of the Company free of charge at https://investors.ivericbio.com/financial-information/sec-filings or by contacting the Company's investor relations department at the following:

IVERIC bio, Inc. Kathy Galante Senior Vice President, Investor Relations kathy.galante@ivericbio.com

Participants in the Solicitation

The Company, and its directors, executive officers and other members of management and certain other people may be deemed to be participants in the solicitation of proxies in connection with the proposed acquisition. Information about the Company's directors and executive officers is included in the proxy statement for the Company's annual meeting of stockholders for 2023, filed with the SEC on April 5, 2023. Additional information regarding these persons and their interests in the merger will be included in the proxy statement relating to the proposed acquisition when it is filed with the SEC. These documents, when available, can be obtained free of charge from the sources indicated above.

Important Additional Information

This Current Report on Form 8-K is for informational purposes only and is not intended to and does not constitute, or form part of, an offer, invitation or the solicitation of an offer or invitation to purchase, otherwise acquire, subscribe for, sell or otherwise dispose of the Company's common stock or any other securities, or the solicitation of any vote or approval in any jurisdiction, pursuant to the proposed acquisition or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: May 1, 2023

IVERIC bio, Inc.

By: /s/ David F. Carroll

Name: David F. Carroll

Title: Senior Vice President, Chief Financial Officer and Treasurer

AGREEMENT AND PLAN OF MERGER

among:

IVERIC BIO, INC.,

a Delaware corporation;

ASTELLAS US HOLDING, INC.,

a Delaware corporation;

BERRY MERGER SUB, INC.,

a Delaware corporation; and

solely as provided by Section 8.10(b) of this Agreement,

ASTELLAS PHARMA INC.,

a company organized under the laws of Japan

Dated as of April 28, 2023

Table of Contents

SECTION 1

MERGER TRANSACTION

1.1	Merger of Merger Sub into the Company	2
1.2	Effect of the Merger	2
1.3	Closing; Effective Time	2
1.4	Certificate of Incorporation and Bylaws; Directors and Officers	3
1.5	Conversion of Shares	3
1.6	Surrender of Certificates; Stock Transfer Books	4
1.7	Dissenters' Rights	7
1.8	Treatment of Company Equity Awards	7
1.9	Further Action	9
	SECTION 2	
	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	
2.1	Due Organization; Subsidiaries, Etc.	9
2.2	Certificate of Incorporation and Bylaws	10
2.3	Capitalization, Etc.	10
2.4	SEC Filings; Financial Statements	12
2.5	Absence of Changes; No Material Adverse Effect	14
2.6	Title to Assets	14
2.7	Real Property	14
2.8	Intellectual Property	15
2.9	Contracts	17
2.10	Liabilities	20
2.11	Compliance with Legal Requirements	20
2.12	Regulatory Matters	21
2.13	Certain Business Practices	22
2.14	Governmental Authorizations	23
2.15	Tax Matters	23
2.16	Employee Matters	26
2.17	Benefit Plans	27
2.18	Environmental Matters	30
2.19	Insurance	30
2.20	Legal Proceedings; Orders	30
2.21	Authority; Binding Nature of Agreement	31
2.22	Takeover Laws	31
2.23	Non-Contravention; Consents	32
2.24	Opinion of Financial Advisors	32

2.25	Acknowledgment by Company	33
	SECTION 3	
	REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	
3.1	Due Organization	33
3.2	Merger Sub	33
3.3	Authority; Binding Nature of Agreement	33
3.4	Non-Contravention; Consents	34
3.5	Disclosure	34
3.6	Absence of Litigation	35
3.7	Funds	35
3.8	Ownership of Shares	35
3.9	Acknowledgement by Parent and Merger Sub	35
3.10	Brokers and Other Advisors	36
	SECTION 4	
	CERTAIN COVENANTS OF THE COMPANY	
4.1	Access and Investigation; Notice of Certain Events	36
4.2	Operation of the Acquired Corporations' Business	38
4.3	No Solicitation	42
4.4	Preparation of Proxy Statement; Stockholder Meeting	45
	SECTION 5	
	ADDITIONAL COVENANTS OF THE PARTIES	
5.1	Company Board Recommendation	47
5.2	Filings, Consents and Approvals	49
5.3	Employee Benefits	52
5.4	ESPP	54
5.5	Indemnification of Officers and Directors	55
5.6	Stockholder Litigation	57
5.7	Additional Agreements	57
5.8	Disclosure	58
5.9	Takeover Laws	58
5.10	Section 16 Matters	58
5.11	Credit Agreement	59
5.12	Stock Exchange Delisting; Deregistration	59
	ii	
	п	

SECTION 6

CONDITIONS PRECEDENT TO THE MERGER

6.1 6.2	Conditions to Each Party's Obligation to Effect the Merger Conditions to Obligations of Parent and Merger Sub to Effect the Merger	60 60
6.3	Conditions to Obligations of the Company to Effect the Merger	61
	SECTION 7	
	TERMINATION	
7.1 7.2 7.3	Termination Effect of Termination Expenses; Termination Fees	62 64 64
	SECTION 8	
	MISCELLANEOUS PROVISIONS	
8.1 8.2 8.3 8.4 8.5 8.6 8.7 8.8 8.9 8.10 8.11	Amendment Waiver No Survival of Representations and Warranties Entire Agreement; Counterparts Applicable Legal Requirements; Jurisdiction; Specific Performance; Remedies Assignability No Third Party Beneficiaries Notices Severability Obligation of Parent; Guaranty Construction	66 66 67 67 67 68 68 69 70 70
Exhibit A	Certain Definitions	
Annexes		
Annex I	Form of Certificate of Incorporation of the Surviving Corporation	
	iii	

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER ("<u>Agreement</u>") is made and entered into as of April 28, 2023, by and among: (i) Astellas US Holding, Inc., a Delaware corporation ("<u>Parent</u>"); (ii) Berry Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("<u>Merger Sub</u>"); (iii) IVERIC bio, Inc., a Delaware corporation (the "<u>Company</u>"); and (iv) solely as provided by <u>Section 8.10(b)</u>, Astellas Pharma Inc., a company organized under the laws of Japan ("<u>Guarantor</u>"). Certain capitalized terms used in this Agreement are defined in <u>Exhibit A</u>.

INTRODUCTION

Parent, a wholly owned subsidiary of Guarantor, desires to acquire the Company through the merger of Merger Sub with and into the Company (the "Merger"), with the Company continuing as the surviving corporation in the Merger (the "Surviving Corporation"), on the terms and subject to the conditions set forth in this Agreement, whereby (i) each issued and outstanding share of Company Common Stock (the "Shares") as of the Effective Time (other than Excluded Shares and Dissenting Shares) shall be cancelled and converted into the right to receive \$40.00 per Share, in cash, without interest (the "Merger Consideration"), and subject to any withholding of Taxes, and (ii) the Company shall become a wholly owned Subsidiary of Parent as a result of the Merger.

The board of directors of the Company (the "Board of Directors") has unanimously (i) determined that this Agreement and the Transactions, including the Merger, are advisable and fair to, and in the best interest of, the Company and its stockholders, (ii) declared it advisable to enter into this Agreement, (iii) approved the execution, delivery and performance by the Company of this Agreement and the consummation of the Transactions, including the Merger, (iv) resolved that the Merger shall be governed by Section 251(c) of the DGCL, upon the terms and subject to the conditions set forth in this Agreement, and (v) resolved to recommend that the stockholders of the Company adopt this Agreement at any meeting of the Company's stockholders held for such purpose and any adjournment or postponement thereof (the preceding clauses (i), (ii) and (v) collectively, the "Company Board Recommendation"), in each case, on the terms and subject to the conditions of this Agreement.

The board of directors of Parent and Merger Sub have approved this Agreement and declared it advisable for Parent and Merger Sub, respectively, to enter into this Agreement.

AGREEMENT

The Parties, intending to be legally bound, agree as follows:

SECTION 1

MERGER TRANSACTION

- 1.1 Merger of Merger Sub into the Company. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with the DGCL, at the Effective Time, the Company and Parent shall consummate the Merger, whereby Merger Sub shall be merged with and into the Company, the separate existence of Merger Sub shall cease, and the Company will continue as the Surviving Corporation.
- 1.2 Effect of the Merger. The Merger shall have the effects set forth in this Agreement and in the applicable provisions of the DGCL. Without limiting the generality of the foregoing, at the Effective Time, all of the property, rights, privileges, immunities, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all of the debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.

1.3 Closing; Effective Time.

- (a) Unless this Agreement shall have been terminated pursuant to Section 7, and unless otherwise mutually agreed in writing between the Company, Parent and Merger Sub, the consummation of the Merger (the "Closing") shall take place by means of a virtual closing via the electronic exchange of documents and signatures by the Parties as soon as practicable (and in no event later than three (3) business days) following the satisfaction or, to the extent permitted by applicable Legal Requirements, waiver (by the Party or Parties entitled to the benefits thereof) of the conditions set forth in Section 6 (other than conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or, to the extent permitted by applicable Legal Requirements, waiver of such conditions at the Closing); provided, that if any of the conditions set forth in Section 6 are not satisfied or, to the extent permitted by applicable Legal Requirements, waived on such third (3rd) business day, then the Closing shall take place on the first (1st) business day thereafter on which all such conditions shall have been satisfied or, to the extent permitted by applicable Legal Requirements, waived; provided, further that if the Parties mutually agree in writing upon another date, the Closing shall take place on such date. The date on which the Closing occurs is referred to in this Agreement as the "Closing Date".
- (b) Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the Company and Merger Sub shall file or cause to be filed a certificate of merger with the Secretary of State of the State of Delaware with respect to the Merger, in such form reasonably agreed upon between the Parties and as required by, and executed and acknowledged in accordance with, the relevant provisions of the DGCL, and the Parties shall take all such further actions as may be required by applicable Legal Requirements to make the Merger effective. The Merger shall become effective upon the date and time of the filing of that certificate of merger with the Secretary of State of the State of Delaware or such later date and time as is agreed upon in writing by the Parties and specified in the certificate of merger (such date and time the Merger is effective, the "Effective Time").

1.4 Certificate of Incorporation and Bylaws; Directors and Officers.

- (a) As of the Effective Time, the certificate of incorporation of the Company shall by virtue of the Merger and without any further action, be amended and restated to read in its entirety as set forth on Annex I and, as so amended and restated, shall be the certificate of incorporation of the Surviving Corporation until thereafter changed or amended as provided therein or by applicable Legal Requirements, subject to Section 5.5(a).
- (b) As of the Effective Time, the bylaws of the Surviving Corporation shall be amended and restated to conform to the bylaws of Merger Sub as in effect immediately prior to the Effective Time, until thereafter changed or amended as provided therein or by applicable Legal Requirements, subject to Section 5.5(a), except that references to the name of Merger Sub shall be replaced by references to the name of the Surviving Corporation.
- (c) As of the Effective Time, the directors and officers of the Surviving Corporation shall be the respective individuals who served as the directors and officers of Merger Sub as of immediately prior to the Effective Time, until their respective successors are duly elected and qualified, or their earlier death, resignation or removal. Prior to the Effective Time, each director of the Company immediately prior to the Effective Time shall execute and deliver a letter effectuating such director's resignation as a member of the Board of Directors, conditioned upon and effective as of the Effective Time.

1.5 Conversion of Shares.

- (a) At the Effective Time, by virtue of the Merger and without any further action on the part of Parent, Merger Sub, the Company or any stockholder of the Company:
- (i) any Shares held immediately prior to the Effective Time by the Company (or held in the Company's treasury) shall automatically be cancelled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor;
- (ii) any Shares held immediately prior to the Effective Time by Parent, Merger Sub or any other direct or indirect wholly owned Subsidiary of Parent or Merger Sub shall automatically be cancelled and retired and shall cease to exist, and no consideration shall be delivered in exchange therefor;
- (iii) any Shares held immediately prior to the Effective Time by any direct or indirect wholly owned Subsidiary of the Company shall be converted into such number of shares of stock of the Surviving Corporation such that each such Subsidiary shall own the same percentage of the outstanding capital stock of the Surviving Corporation immediately following the Effective Time as such Subsidiary owned in the Company immediately prior to the Effective Time and no other consideration shall be delivered in exchange therefor (such Shares together with the Shares described in the preceding clauses (i) and (ii), the "Excluded Shares");

- (iv) except as provided in <u>clauses (i)</u>, (ii) and (<u>iii)</u> above and subject to <u>Section 1.5(b)</u>, each Share outstanding immediately prior to the Effective Time (other than any Dissenting Shares, which shall have only those rights set forth in <u>Section 1.7</u>) shall be converted into the right to receive the Merger Consideration, in each case without any interest thereon and subject to any withholding of Taxes in accordance with <u>Section 1.6(e)</u>; and
- (v) each share of the common stock, \$0.001 par value per share, of Merger Sub then outstanding shall be converted into one share of common stock of the Surviving Corporation.

From and after the Effective Time, subject to this <u>Section 1.5(a)</u>, all Shares shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and each applicable holder of such Shares (other than Dissenting Shares) shall cease to have any rights with respect thereto, except the right to receive the Merger Consideration therefor upon the surrender of such Shares in accordance with <u>Section 1.6</u> or, in the case of Dissenting Shares, the rights set forth in <u>Section 1.7</u>.

(b) If, between the date of this Agreement and the Effective Time, the outstanding Shares are changed into a different number or class of shares by reason of any stock split, division or subdivision of shares, stock dividend, reverse stock split, consolidation of shares, reclassification, recapitalization or other similar transaction, then the Merger Consideration shall be appropriately adjusted; *provided*, that nothing in this Section 1.5(b) shall be construed to permit the Company to take any action that is prohibited by Section 4.2 or the other terms of this Agreement.

1.6 Surrender of Certificates; Stock Transfer Books.

(a) Prior to the Closing Date, Parent shall designate a bank or trust company reasonably acceptable to the Company to act as agent (the "Paying Agent") for the holders of Shares to receive the aggregate Merger Consideration to which holders of such Shares shall become entitled pursuant to Section 1.5. On or prior to the Closing Date, Parent shall deposit, or shall cause to be deposited, with the Paying Agent cash sufficient to pay the aggregate Merger Consideration payable pursuant to Section 1.5 (the "Payment Fund"). The Payment Fund shall not be used for any purpose other than to pay the aggregate Merger Consideration in the Merger; provided that any interest or other income resulting from investment of the Payment Fund that results in an amount being held in the Payment Fund that is greater than the amount payable pursuant to this Section 1.6 shall be promptly returned to Parent. The Payment Fund shall be invested by the Paying Agent as directed by Parent; provided that such investments shall be (w) in obligations of or guaranteed by the United States of America, (x) in commercial paper obligations rated A-1 or P-1 or better by Moody's Investors Service, Inc. or Standard & Poor's Corporation, respectively, (y) in certificates of deposit, bank repurchase agreements or banker's acceptances of commercial banks with capital exceeding \$1 billion, or (z) in money market funds having a rating in the highest investment category granted by a recognized credit rating agency at the time of acquisition or a combination of the foregoing and, in any such case, no such instrument shall have a maturity exceeding three (3) months. To the extent Parent becomes aware that (i) there are any losses with respect to any such investments or (ii) the Payment Fund has diminished for any reason below the level required for the Paying Agent to make prompt cash payment pursuant to Section 1.5. Parent or the Surviving Corporation shall promptly replace or restore the cash in the Payment Fund, so as to ensure that the Payment Fund is, at

Promptly after the Effective Time (but in no event later than three (3) business days thereafter), the Surviving Corporation shall cause to be delivered to each Person who was, at the Effective Time, a holder of record of (i) Shares represented by a certificate evidencing such Shares ("Certificated Shares") or (ii) Book-Entry Shares, who, in each case of (i) and (ii) was entitled to receive the Merger Consideration pursuant to Section 1.5, (A) a form of letter of transmittal, which shall be in reasonable and customary form and shall specify that delivery shall be effected, and risk of loss and title to the Certificated Shares shall pass, only upon proper delivery of the stock certificates evidencing such Certificated Shares (the "Certificates") (or affidavits of loss in lieu thereof in accordance with Section 1.6(f), if applicable) to the Paying Agent, or a customary agent's message with respect to Book-Entry Shares, and (B) instructions (including the requirement for each holder of Shares to provide IRS Form W-9 or applicable series of IRS Form W-8) for use in effecting the surrender of the Certificates and Book-Entry Shares in exchange for the Merger Consideration payable in respect of such Shares pursuant to Section 1.5. Upon surrender to the Paying Agent of Certificates (or affidavits of loss in lieu thereof in accordance with Section 1.6(f), if applicable) or Book-Entry Shares, together with such letter of transmittal in the case of Certificates, duly completed and validly executed in accordance with the instructions thereto, and such other documents as may be required pursuant to the instructions, the holder of such Certificated Shares or Book-Entry Shares shall be entitled to receive in exchange therefor the Merger Consideration for each Certificated Share or Book-Entry Share, and such Certificates and Book-Entry Shares shall then be cancelled. No interest shall accrue or be paid on the Merger Consideration payable upon the surrender of any Certificates or Book-Entry Shares for the benefit of the holder thereof. If the payment of any Merger Consideration is to be made to a Person other than the Person in whose name the surrendered Certificates formerly evidencing the Certificated Shares is registered on the stock transfer books of the Company, it shall be a condition of payment that the Certificate so surrendered shall be endorsed properly or otherwise be in proper form for transfer and that the Person requesting such payment shall have paid all transfer and other Taxes required by reason of the payment of the Merger Consideration to a Person other than the registered holder of the Certificate surrendered, or shall have established to the satisfaction of Parent that such transfer or other Taxes either have been paid or are not applicable. None of Parent, Merger Sub or the Surviving Corporation shall have any liability for the transfer and other Taxes described in this Section 1.6(b) under any circumstance. Payment of the applicable Merger Consideration with respect to Book-Entry Shares shall only be made to the Person in whose name such Book-Entry Shares are registered. Until surrendered as contemplated by this Section 1.6, each Certificated Share and Book-Entry Share (in each case, other than Dissenting Shares) shall be deemed at any time after the Effective Time to represent only the right to receive the Merger Consideration as contemplated by <u>Section 1.5</u>.

- (c) At any time following twelve (12) months after the Effective Time, Parent shall be entitled to require the Paying Agent to deliver to it any funds (with respect to the aggregate Merger Consideration to which holders of Shares shall become entitled pursuant to Section 1.5) which had been made available to the Paying Agent and not disbursed to holders of Certificated Shares or Book-Entry Shares in accordance with this Section 1.6 (including all interest and other income received by the Paying Agent in respect of all funds made available to it), and, thereafter, such holders shall be entitled to look to the Surviving Corporation (subject to abandoned property, escheat and other similar Legal Requirements) only as general creditors thereof with respect to the Merger Consideration that may be payable upon due surrender of the Certificates or Book-Entry Shares held by them, without any interest thereon. Notwithstanding the foregoing, neither the Surviving Corporation nor the Paying Agent shall be liable to any holder of Certificated Shares or Book-Entry Shares for the Merger Consideration delivered in respect of such Share to a public official pursuant to any abandoned property, escheat or other similar Legal Requirements. Any amounts remaining unclaimed by such holders at such time at which such amounts would otherwise escheat to or become property of any Governmental Body shall become, to the extent permitted by applicable Legal Requirements, the property of the Surviving Corporation or its designee, free and clear of all claims or interest of any Person previously entitled thereto.
- (d) At the close of business on the day of the Effective Time, the stock transfer books of the Company with respect to the Shares shall be closed and thereafter there shall be no further registration of transfers of Shares on the records of the Company. From and after the Effective Time, the holders of the Shares outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such Shares except as otherwise provided herein or by applicable Legal Requirements. If, after the Effective Time, Certificates or Book-Entry Shares are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in this Agreement.
- (e) Each of the Company, the Surviving Corporation, Parent and Merger Sub, and their Affiliates, shall be entitled to deduct and withhold (or cause the Paying Agent to deduct and withhold) from the Merger Consideration payable to any holder of the Shares or any holder of Company Options, Company RSUs, Company PSUs or any other consideration otherwise payable pursuant to this Agreement such amounts as it is required by any Legal Requirement to deduct and withhold with respect to Taxes. Each such withholding agent shall use commercially reasonable efforts to reduce or eliminate any such withholding, including by requesting any necessary Tax forms, including IRS Form W-9 or the appropriate series of IRS Form W-8, as applicable, or any similar information. Each such withholding agent shall take all action that may be necessary to ensure that any such amounts so withheld are timely and properly remitted to the appropriate Governmental Body. To the extent that amounts are so withheld and timely and properly remitted to the appropriate Governmental Body, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of the Shares, holder of Company Options, Company RSUs, Company PSUs or other recipient of consideration hereunder in respect of which such deduction and withholding was made.
- (f) If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the holder of the Certificated Shares formerly represented by that Certificate, or by a representative of that holder, claiming that Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by that holder of a bond, in such reasonable amount as Parent may direct, as indemnity against any claim that may be made against it with respect to such Certificate (which amount shall not exceed the Merger Consideration payable with respect to such Certificated Shares), the Paying Agent will pay (less any amounts entitled to be deducted or withheld pursuant to Section 1.6(e)), in exchange for such lost, stolen or destroyed Certificate, the applicable Merger Consideration to be paid in respect of the Certificated Shares formerly represented by such Certificate, as contemplated by this Section 1.

1.7 Dissenters' Rights. Notwithstanding anything to the contrary in this Agreement, any Shares outstanding immediately prior to the Effective Time which are held by holders who are entitled to appraisal rights under Section 262 of the DGCL and have properly exercised and perfected their respective demands for appraisal of such Shares in the time and manner provided in Section 262 of the DGCL and, as of the Effective Time, have neither effectively withdrawn nor lost their rights to such appraisal and payment under the DGCL (the "Dissenting Shares"), shall not be converted into the right to receive the Merger Consideration, but shall, by virtue of the Merger, be automatically cancelled and no longer outstanding, shall cease to exist and shall be entitled to only such consideration as shall be determined pursuant to Section 262 of the DGCL; provided that if any such holder shall have failed to perfect or shall have effectively withdrawn or lost such holder's right to appraisal and payment under the DGCL, such holder's Shares shall be deemed to have been converted as of the Effective Time into the right to receive the Merger Consideration (less any amounts entitled to be deducted or withheld pursuant to Section 1.6(e)), and such Shares shall no longer be deemed to be Dissenting Shares. The Company shall give prompt notice to Parent and Merger Sub of any demands received by the Company for appraisal of any Dissenting Shares, withdrawals of such demands and any other instruments served pursuant to Section 262 of the DGCL, in each case prior to the Effective Time. Parent and Merger Sub shall have the right to direct and participate in all negotiations and proceedings with respect to such demands, and the Company shall not, without the prior written consent of Parent and Merger Sub, settle or offer to settle, or make any payment with respect to, any such demands, approve any withdrawal of any such demands, or agree or commit to do any of the foregoing.

1.8 Treatment of Company Equity Awards.

- (a) At the Effective Time, each Company Option that is then outstanding and unexercised, whether or not vested and which has a per share exercise price that is less than the Merger Consideration (each, an "In the Money Option"), shall be cancelled and converted into the right to receive a cash payment equal to (A) the excess of (x) the Merger Consideration over (y) the exercise price payable per Share under such In the Money Option, multiplied by (B) the total number of Shares subject to such In the Money Option immediately prior to the Effective Time (without regard to vesting).
- (b) At the Effective Time, each Company Option other than an In the Money Option that is then outstanding and unexercised, whether or not vested, shall be cancelled with no consideration payable in respect thereof.

- (c) At the Effective Time, each then outstanding Company RSU shall be canceled and the holder thereof shall be entitled to receive a cash payment equal to the product of (x) the Merger Consideration and (y) the number of Shares subject to such Company RSU.
- (d) At the Effective Time, each then outstanding Company PSU shall be canceled and converted into a cash-based award (a "Converted PSU"), which shall entitle the holder thereof to receive a cash payment equal to the product of (x) the Merger Consideration, and (y) the number of Shares subject to such Company PSU, subject to the same terms and conditions (including vesting, forfeiture and acceleration provisions) that were applicable to the corresponding Company PSU immediately prior to the Effective Time. Parent shall make (or shall cause the Surviving Corporation to make) a payment in respect of each portion of a Converted PSU that becomes vested on the Surviving Corporation's next regularly scheduled payroll date following the applicable vesting date (but in no event later than ten (10) business days after the applicable vesting date). Following the Effective Time, no Converted PSU may be pledged, encumbered, sold, assigned or transferred (including any transfer by operation of law), by any Person, other than to the Surviving Corporation, or be taken or reached by any legal or equitable process in satisfaction of any liability of such Person.
- (e) As soon as reasonably practicable after the Effective Time (but no later than the second payroll date after the Effective Time), the Surviving Corporation or its Affiliate, as applicable, shall pay the aggregate consideration payable pursuant to Sections 1.8(a) and 1.8(c), net of any applicable withholding Taxes, payable with respect to the In the Money Options and Company RSUs through, to the extent applicable, the Surviving Corporation's or its Affiliate's payroll to the holders of In the Money Options and Company RSUs. Parent shall cause the Surviving Corporation or its Affiliates, as applicable, to pay the aggregate consideration payable pursuant to Section 1.8(d) at the times provided in such Section, net of any applicable withholding Taxes, payable with respect to the Converted PSUs through, to the extent applicable, the Surviving Corporation's or its Affiliate's payroll to the holders of Converted PSUs.
- (f) Prior to the Effective Time, the Company shall take all actions necessary (under the Company Equity Plans and award agreements pursuant to which Company Options, Company RSUs and Company PSUs are outstanding or otherwise) to effect the transactions described in this Section 1.8, including in order to comply with the provisions of the Company Equity Plans, such that immediately following the Effective Time there are no obligations of the Acquired Corporations, Parent, Merger Sub or their Affiliates in respect of Company Options, Company RSUs and Company PSUs, except to make the payments set forth in this Section 1.8. All Company Options, Company RSUs and Company PSUs shall terminate as of the Effective Time, and following the Effective Time, no holder of any Company Options, Company RSUs and Company PSUs will have any right to acquire any equity securities of the Company, the Surviving Corporation or any of their Affiliates as a result of such holder's Company Options, Company RSUs and Company PSUs.
- (g) To the extent a payment made pursuant to the timing set forth in this Section 1.8 would trigger a Tax or penalty under Section 409A of the Code, such payment shall be made on the earliest date that payment would not trigger such Tax or penalty.

1.9 Further Action. The Parties agree to take all necessary action to cause the Merger to become effective in accordance with Section 251(c) of the DGCL and this Section 1 as soon as practicable following the satisfaction or, to the extent permitted by applicable Legal Requirements, waiver of the last of the conditions set forth in Section 6 (other than conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or, to the extent permitted by applicable Legal Requirements, waiver of such conditions). If, at any time after the Effective Time, any further action is reasonably determined by Parent to be necessary or desirable to carry out the purposes of this Agreement or to vest the Surviving Corporation with full right, title and possession of and to all rights and property of Merger Sub and the Company, the officers and directors of the Surviving Corporation and Parent shall be fully authorized (in the name of Merger Sub, in the name of the Company and otherwise) to take such action.

SECTION 2

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company hereby represents and warrants to Parent and Merger Sub as follows, it being understood that each representation and warranty contained in this Section 2 is subject to (a) exceptions and disclosures set forth in the section or subsection of Section 2 of the Company Disclosure Schedule corresponding to the particular section or subsection in this Section 2; (b) any exception or disclosure set forth in any other section or subsection of Section 2 of the Company Disclosure Schedule to the extent it is reasonably apparent on the face of such exception or disclosure that such exception or disclosure is applicable to qualify such section or subsection; and (c) disclosure in the Company SEC Documents filed on or after January 1, 2022 and publicly available prior to the close of business on the second (2nd) business day preceding the date of this Agreement (other than any information in the "Risk Factors" or "Forward-Looking Statements" sections of such Company SEC Documents or other general cautionary, predictive or forward-looking statements in any other sections of such Company SEC Documents); provided that (x) clauses (b) and (c) of this paragraph shall not apply to the representation and warranty set forth in Section 2.5(b) (No Material Adverse Effect) and (y) clause (c) of this paragraph shall not apply to any of the representations and warranties set forth in Section 2.1 (Due Organization; Subsidiaries, Etc.), Section 2.2 (Certificate of Incorporation and Bylaws), Section 2.3 (Capitalization, Etc.), Section 2.21 (Authority; Binding Nature of Agreement), Section 2.22 (Takeover Laws), Section 2.23 (Non-Contravention; Consents), Section 2.24 (Opinion of Financial Advisors) or Section 2.25 (Brokers and Other Advisors):

2.1 Due Organization; Subsidiaries, Etc.

(a) The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, and the Company's only Subsidiaries are set forth on Section 2.1 of the Company Disclosure Schedule (the Company and each such Subsidiary, an "Acquired Corporation" and collectively, the "Acquired Corporations"). Each Acquired Corporation has all necessary power and authority: (i) to conduct its business in the manner in which its business is currently being conducted; and (ii) to own and use its assets in the manner in which its assets are currently owned and used. Each Acquired Corporation is qualified or licensed to do business as a foreign corporation, and is in good standing, in each jurisdiction where the nature of its business requires such qualification or licensing, except where such failure does not have, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

- (b) The Company owns beneficially and of record all of the outstanding shares of capital stock or ordinary shares of the other Acquired Corporations, free and clear of all Encumbrances and transfer restrictions, except for Encumbrances or transfer restrictions of general applicability as may be provided under the Securities Act or applicable securities laws. Except for the shares of capital stock or ordinary shares of the other Acquired Corporations held by the Company, no Acquired Corporation owns, directly or indirectly, any capital stock or equity interests of any nature in, or subscriptions, options, calls, warrants or rights (whether or not currently exercisable) to acquire, or other securities convertible into or exchangeable or exercisable for, any capital stock or equity interests of, any Entity. None of the Acquired Corporations has agreed or is obligated to make, or is bound by any Contract under which it may become obligated to make, any future investment in or capital contribution to any other Entity.
- 2.2 Certificate of Incorporation and Bylaws. The Company has delivered or made available to Parent accurate and complete copies of the certificate of incorporation, bylaws and similar charter and organizational documents of each Acquired Corporation, including all amendments thereto, as in effect on the date hereof.

2.3 Capitalization, Etc.

- (a) The authorized capital stock of the Company consists of: (i) 200,000,000 Shares, of which 137,616,082 Shares had been issued and were outstanding as of the close of business on April 27, 2023 (the "Capitalization Date"); and (ii) 5,000,000 shares of Company Preferred Stock, of which no shares are outstanding. All of the outstanding shares of the capital stock of the Company have been duly authorized and validly issued, and are fully paid and nonassessable. In the period between the Capitalization Date and the execution and delivery of this Agreement, the Company has not issued any new Shares except pursuant to the exercise of the purchase rights under the Company ESPP, the vesting of Company RSUs outstanding as of the Capitalization Date in accordance with their terms or the exercise of Company Options outstanding as of the Capitalization Date in accordance with their terms and, since the Capitalization Date, the Company has not issued any Company Options or other equity based awards, in each case, other than pursuant to any offer of employment or Contract executed on or prior to the date of the Capitalization Date.
- (b) All of the outstanding shares of the capital stock or ordinary shares, as applicable, of the Company's Subsidiaries have been duly authorized and validly issued, and are fully paid and nonassessable.
- (c) (i) None of the outstanding shares of capital stock of the Acquired Corporations are entitled or subject to any preemptive right, right of repurchase or forfeiture, right of participation, right of maintenance or any similar right; (ii) none of the outstanding shares of capital stock of the Acquired Corporations are subject to any right of first refusal in favor of any Acquired Corporation; (iii) there are no outstanding bonds, debentures, notes or other indebtedness of any Acquired Corporation having a right to vote on any matters on which the stockholders of the Acquired Corporations have a right to vote; and (iv) there is no Contract relating to the voting or registration of, or restricting any Person from purchasing, selling, pledging or otherwise disposing of (or from granting any option or similar right with respect to), any shares of capital stock of the Acquired Corporations. No Acquired Corporation is under any obligation, or bound by any Contract pursuant to which it may become obligated, to repurchase, redeem or otherwise acquire any outstanding shares of capital stock of the Acquired Corporations. The Shares constitute the only outstanding class of securities of the Company registered under the Securities Act. There are no voting trusts or other Contracts with respect to the voting of any Shares.

- (d) As of the close of business on the Capitalization Date: (i) 12,729,560 Shares were subject to issuance pursuant to Company Options granted and outstanding under the Company Equity Plans, (ii) 2,497,540 Shares were subject to issuance pursuant to Company RSUs and 198,432 Shares were subject to issuance pursuant to Company PSUs (determined at the maximum level of performance) granted and outstanding under the Company Equity Plans, (iii) 4,561,362 Shares were reserved for future issuance under Company Equity Plans and (iv) 676,853 Shares were reserved for future issuance under the Company ESPP. Other than as set forth in this Section 2.3(d), there are no issued, reserved for issuance, outstanding or authorized stock option, stock appreciation, phantom stock, stock unit, restricted stock unit, stock-based performance unit, profit participation or similar rights or equity-based awards with respect to the Company.
- (e) There are no issued, reserved for issuance, outstanding or authorized stock option, stock appreciation, phantom stock, stock unit, profit participation or similar rights or equity-based awards with respect to any Subsidiary of the Company.
- (f) Except (y) as set forth in this Section 2.3 and (z) for Company Options, Company RSUs and Company PSUs outstanding as of the date of this Agreement, there are no: (i) outstanding shares of capital stock of or other securities of any Acquired Corporation; (ii) outstanding subscriptions, options, calls, warrants or rights (whether or not currently exercisable) to acquire any shares of the capital stock, restricted stock unit, stock-based performance unit or any other right that is linked to, or the value of which is in any way based on or derived from the value of any shares of capital stock or other securities of any Acquired Corporation, in each case other than derivative securities not issued by an Acquired Corporation; (iii) outstanding securities, instruments, bonds, debentures, notes or obligations that are or may become convertible into or exchangeable for any shares of the capital stock or other securities of any Acquired Corporation; or (iv) stockholder rights plans (or similar plans commonly referred to as a "poison pill") or Contracts under which any Acquired Corporation is or may become obligated to sell or otherwise issue any shares of its capital stock or any other securities.
- (g) Section 2.3(g) of the Company Disclosure Schedule sets forth a listing of all Persons (either by name or employee identification number) who hold outstanding Company Options, Company RSUs and Company PSUs as of the close of business on the Capitalization Date, indicating, with respect to each Company Option, Company RSU and Company PSU, the number of Shares subject thereto, the date of grant, the vesting schedule, the per Share exercise price (if applicable), the expiration date and, with respect to Company Options, whether such Company Option is intended to be an incentive stock option (within the meaning of Section 422 of the Code). All Company Options, Company RSUs and Company PSUs are evidenced by award agreements in the forms that have been provided to Parent.

(h) All outstanding Shares have been offered and issued in compliance in all material respects with all applicable securities Legal Requirements, including the Securities Act and "blue sky" Legal Requirements.

2.4 SEC Filings; Financial Statements.

- (a) Since January 1, 2020 (the "Applicable Date"), the Company has filed or furnished on a timely basis all reports, schedules, forms, statements and other documents (including exhibits and all other information incorporated therein) required to be filed or furnished by the Company with the SEC (as supplemented, modified or amended since the time of filing, the "Company SEC Documents"). As of their respective dates, or, if amended prior to the date of this Agreement, as of the date of (and giving effect to) the last such amendment (and, in the case of registration statements and proxy statements, on the date of effectiveness and the dates of the relevant meetings, respectively), the Company SEC Documents complied in all material respects with the requirements of the Securities Act, the Exchange Act or the Sarbanes-Oxley Act of 2002, as amended (the "Sarbanes-Oxley Act"), as the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to those Company SEC Documents, and, except to the extent that information contained in such Company SEC Document has been revised, amended, modified or superseded (prior to the date of this Agreement) by a later filed Company SEC Document, none of the Company SEC Documents when filed or furnished contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. No executive officer of the Company has failed to make the certifications required of him or her under Section 302 or 906 of the Sarbanes Oxley Act with respect to any Company SEC Documents.
- (b) The financial statements (including any related notes and schedules) contained or incorporated by reference in the Company SEC Documents: (i) complied as to form in all material respects with the published rules and regulations of the SEC applicable thereto; (ii) were prepared in accordance with United States generally accepted accounting principles ("GAAP") applied on a consistent basis throughout the periods covered (except as may be indicated in the notes to such financial statements or, in the case of unaudited interim financial statements, as may be permitted by the SEC on Form 10-Q, Form 8-K or any successor form under the Exchange Act); and (iii) fairly presented, in all material respects, the financial position of the Company as of the respective dates thereof and the results of operations and cash flows of the Company for the periods covered thereby (subject, in the case of the unaudited financial statements, to the absence of notes and to normal and recurring year-end adjustments that are not individually or in the aggregate material).

- (c) The Company maintains, and at all times since the Applicable Date has maintained, a system of internal control over financial reporting (as defined in Rule 13a-15 under the Exchange Act) which is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP, and includes those policies and procedures that:
 (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of the Company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and that receipts and expenditures are being made only in accordance with authorizations of management and the Board of Directors; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the assets of the Company that could have a material effect on the financial statements. The Company's management has completed an assessment of the effectiveness of the Company internal control over financial reporting in compliance with the requirements of Section 404 of the Sarbanes-Oxley Act for the fiscal year ended December 31, 2022, and, except as set forth in the Company SEC Documents filed prior to the date of this Agreement, that assessment concluded that those controls were effective. To the knowledge of the Company, since January 1, 2023, neither the Company or the Company's independent registered accountant has identified or been made aware of: (1) any significant deficiency or material weakness in the design or operation of the internal control over financial reporting utilized by the Company, which is reasonably likely to adversely affect the Company's ability to record, process, summarize and report financial information; (2) any illegal act or fraud, whether or not material, that involves the management or other
- (d) The Company maintains, and at all times since the Applicable Date has maintained, disclosure controls and procedures as defined in and required by Rule 13a-15 or 15d-15 under the Exchange Act that are reasonably designed to ensure that all information required to be disclosed in the Company's reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC and that all such information is accumulated and communicated to the Company's management as appropriate to allow timely decisions regarding required disclosure and to enable the principal executive officer of the Company and the principal financial officer of the Company to make the certifications required under the Exchange Act with respect to such reports. The Company is, and at all times since the Applicable Date has been, in compliance in all material respects with all current listing and corporate governance requirements of Nasdaq.
- (e) Since January 1, 2023, (i) no Acquired Corporation has received any material written complaint, allegation, assertion or claim regarding the accounting or auditing practices, procedures, methodologies or methods of any of the Acquired Corporations or their respective internal accounting controls, including any credible complaint, allegation, assertion or claim that any of the Acquired Corporations has engaged in questionable accounting or auditing practices, and (ii) no attorney representing any of the Acquired Corporations (whether or not employed by any Acquired Corporation) has reported evidence of a material violation of the applicable Legal Requirements, breach of fiduciary duty or similar violation by any of the Acquired Corporations or their respective officers, directors, employees or agents to the Board of Directors or any committee thereof or to any director or officer of the Acquired Corporations pursuant to the rules of the SEC adopted under Section 307 of the Sarbanes Oxley Act.

- (f) None of the Acquired Corporations is a party to, nor does any Acquired Corporation have any obligation or other commitment to become a party to, "off-balance sheet arrangements" (as defined in Item 303(a) of Regulation S-K under the Exchange Act) where the result, purpose or intended effect of such Contract is to avoid disclosure of any material transaction involving, or material liabilities of, such Acquired Corporation in the Company SEC Documents.
- (g) As of the date of this Agreement, there are no outstanding or unresolved comments in comment letters received from the SEC with respect to the Company SEC Documents. None of the Company SEC Documents is the subject of ongoing SEC review and there are no inquiries or investigations by the SEC or any internal investigations pending or threatened, in each case regarding any accounting practices of the Company.
- **2.5 Absence of Changes; No Material Adverse Effect**. Except as expressly contemplated by this Agreement, from January 1, 2023 through the date of this Agreement:
- (a) except for discussions, negotiations and activities related to this Agreement or other potential strategic transactions, the Acquired Corporations have operated in all material respects in the ordinary course of business consistent with past practice;
 - (b) there has not occurred a Material Adverse Effect; and
- (c) there has not been any action taken by any of the Acquired Corporations that, if taken during the Pre-Closing Period without Parent's prior written consent, would require Parent's consent pursuant to <u>Section 4.2</u>.
- 2.6 Title to Assets. Each Acquired Corporation has good and valid title to all assets (excluding Intellectual Property Rights) owned or purported to be owned by it as of the date of this Agreement and necessary for the conduct of the business of the Acquired Corporations, taken as a whole, and such assets are owned by the Acquired Corporations free and clear of any Encumbrances (other than Permitted Encumbrances).

2.7 Real Property.

- (a) The Acquired Corporations do not own any real property.
- (b) The Acquired Corporations hold valid and existing leasehold interests in the real property that is leased or subleased by the Acquired Corporations from another Person (the "Leased Real Property"), free and clear of all Encumbrances other than Permitted Encumbrances. Since the Applicable Date, no Acquired Corporation has received any notice regarding (i) any material violation or breach or default under any lease related to the Leased Real Property that has not since been cured; (ii) pending or threatened condemnation of any portion of the Leased Real Property; or (iii) building, fire or zoning code violations with respect to the Leased Real Property. None of the Leased Real Property is subleased.

2.8 Intellectual Property.

- (a) Section 2.8(a) of the Company Disclosure Schedule sets forth an accurate and complete list that identifies the following for each item of material Company IP that is Registered IP as applicable: (i) the name of the current owner, (ii) the jurisdiction of application or registration and (iii) the application or registration number, and all such Patents and Trademarks are subsisting and, to the knowledge of the Company, other than pending applications therefor, valid and enforceable. As of the date of this Agreement, no interference, opposition, reissue, reexamination proceeding, cancellation proceeding, or other Legal Proceeding (other than routine examination proceedings with respect to pending applications) is pending or, to the knowledge of the Company, threatened in writing against any Acquired Corporation, in which the scope, validity, enforceability, priority, inventorship or ownership of any Registered IP owned by or exclusively licensed to an Acquired Corporation is being contested or challenged. All application, registration, issuance, renewal and maintenance fees due for such Company IP owned or purported to be owned by the Company that is Registered IP having a final due date on or before the date hereof have been paid in full and are current.
- (b) The Acquired Corporations are the sole and exclusive owners of all right, title and interest in and to all Company IP owned or purported to be owned by the Acquired Corporations, free and clear of all Encumbrances, other than Permitted Encumbrances, and have the right, pursuant to enforceable agreements, to use all other Intellectual Property Rights used by the Acquired Corporations in their respective businesses.
- (c) No Company Associate owns or has any valid claim, right (whether or not currently exercisable) or interest to or in any material Company IP owned or purported to be owned by an Acquired Corporation and each Company Associate who is or was involved in the creation or development of any material Company IP has signed a valid and enforceable written agreement containing a present-tense assignment of Intellectual Property Rights in and to such Company IP to an Acquired Corporation and confidentiality provisions protecting such Company IP, and there is no material breach under any such agreement.
- (d) No funding, facilities or personnel of any Governmental Body or any university, college, research institute or other educational institution is being or has been used to create, in whole or in part, material Company IP owned or purported to be owned by the Acquired Corporations or, to the knowledge of the Company, material Company IP licensed to the Acquired Corporations, except for any such funding or use of facilities or personnel that does not result in such Governmental Body or institution obtaining ownership or other similar rights to such material Company IP.
- (e) The Acquired Corporations have taken reasonable steps, including reasonable security measures and other steps, to maintain the confidentiality of and otherwise protect and enforce their rights in all Trade Secrets held by an Acquired Corporation in all material respects.
- (f) Section 2.8(f) of the Company Disclosure Schedule sets forth each license agreement pursuant to which an Acquired Corporation has obtained a license to, or has been granted a covenant not to sue under, immunity from suit with respect to or other equivalent rights to, any material Intellectual Property Right (each an "In-bound License") or has granted a license or covenant not to sue, immunity from suit or other equivalent right to any material Intellectual Property Right owned or purported to be owned by, or licensed to, an Acquired Corporation (each an "Out-bound License"); provided, that (i) In-bound Licenses shall not include commercially available off-the-shelf software, clinical trial agreements, non-disclosure agreements or material transfer agreements entered into in the ordinary course of business consistent with past practice and which are not otherwise material, and (ii) Out-bound Licenses shall not include non-exclusive outbound licenses contained in clinical trial agreements, contract manufacturing agreements, non-disclosure agreements and material transfer agreements entered into in the ordinary course of business consistent with past practice and which are not otherwise material.

- (h) To the knowledge of the Company, no Person is infringing, misappropriating or otherwise violating, or has infringed, misappropriated or otherwise violated, any material Company IP. As of the date of this Agreement, no Legal Proceeding is pending or threatened in writing by an Acquired Corporation relating to any actual, alleged or suspected infringement, misappropriation or other violation of any Company IP.
- (i) None of the Company IP owned by an Acquired Corporation, or, to the knowledge of the Company licensed to an Acquired Corporation, is subject to any pending or outstanding injunction, directive, order, judgment, settlement, consent ruling or other disposition of dispute that adversely restricts the use, transfer or licensing of any such Company IP by an Acquired Corporation, or otherwise adversely affects the validity, scope, use, registrability, or enforceability of any such Company IP.
- (j) The consummation of the Transactions will not result in the loss or impairment of, violation of, alteration in, forfeiture of, termination of, or payment of any additional amounts with respect to, nor require the consent of any other Person in respect of, any material Company IP or any other material Intellectual Property Rights used by the Acquired Corporations in their respective businesses.

- (k) Except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, (i) the Company IT Assets operate in accordance with their specifications and related documentation and perform in a manner that permits the Acquired Corporations to conduct its business as currently conducted, and (ii) the Company has taken commercially reasonable actions, consistent with current industry standards, to protect the confidentiality, integrity and security of the Company IT Assets (and all data and other information and transactions stored or contained therein or processed or transmitted thereby) against any unauthorized use, access, interruption, modification or corruption, including the implementation of commercially reasonable data backup, disaster avoidance and recovery procedures and business continuity procedures. Since the Applicable Date, there has been no unauthorized use, access or security breaches, or interruption, modification, loss or corruption of any of Company IT Assets (or any data or other information or transactions stored or contained therein or processed or transmitted thereby).
- (I) The Acquired Corporations have established and implemented organizational, physical, administrative and technical measures regarding privacy, cybersecurity and data security that are commercially reasonable and consistent in all material respects with all (i) Data Privacy Laws, (ii) applicable industry standards including the Payment Card Industry Data Security Standard, (iii) internal and publicly available privacy policies and notices of the Company relating to Personal Information, (iv) Contract obligations of the Acquired Corporations or any of its Subsidiaries relating to the collection, Processing, disclosure and storage of Personal Information, and (v) consents and authorizations pursuant to which Personal Information was disclosed to the Acquired Corporations (the foregoing clauses (i) through (v), collectively, the "Privacy and Security Requirements"). The Acquired Corporations are, and since the Applicable Date have been, in all material respects, in compliance with and not in material default under, or in material violation of, any applicable Privacy and Security Requirements. Since the Applicable Date, no Acquired Corporation has been given notice of, or been charged with, any material violation of any applicable Privacy and Security Requirement. Since the Applicable Date, no Acquired Corporation has experienced any material security breach or cybersecurity event, including any theft, loss, or unauthorized access or acquisition of Personal Information.

2.9 Contracts.

- (a) Section 2.9(a) of the Company Disclosure Schedule identifies each Contract to which any Acquired Corporation is a party, or by which it is bound, that constitutes a Material Contract as of the date of this Agreement and identifies, with respect to each Material Contract, the clause of this Section 2.9(a) to which it applies. For purposes of this Agreement, each of the following (excluding any Employee Plan) to which any Acquired Corporation is a party or by which it is bound as of the date of this Agreement constitutes a "Material Contract":
- (i) any Contract that is a settlement, conciliation or similar agreement between any Acquired Corporation and any Governmental Body and pursuant to which (A) an Acquired Corporation will be required after the date of this Agreement to pay any monetary obligations or (B) that contains material obligations or limitations on such Acquired Corporation's conduct;

(ii) any Contract between any Acquired Corporation and any third Person (A) materially limiting the freedom or right of any
Acquired Corporation to engage in any line of business or to compete with any other Person in any location or line of business, (B) containing any "most
favored nations" terms and conditions (including with respect to pricing) granted by any Acquired Corporation, or (C) containing exclusivity obligations or
otherwise materially limiting the freedom or right of any Acquired Corporation to sell, distribute or manufacture any products or services for any other
Person;

- (iii) any Contract that requires by its terms or is reasonably expected to require the payment or delivery of cash or other consideration (A) to any Acquired Corporation in an amount having an expected value in excess of \$250,000 in the fiscal year ending December 31, 2023 or (B) by any Acquired Corporation in an amount having an expected value in excess of \$1,000,000 in the fiscal year ending December 31, 2023, and in each case (y) which cannot be cancelled by such Acquired Corporation without penalty or further payment without more than ninety (90) days' notice and (z) excluding commercially available off-the-shelf software licenses and Software-as-a-Service offerings, generally available patent license agreements, material transfer agreements, clinical trial agreements and non-exclusive outbound license agreements (in each case, entered into in the ordinary course of business);
- (iv) any Contract relating to Indebtedness of any Acquired Corporation in an aggregate principal amount in excess of \$100,000 (whether incurred, assumed, guaranteed or secured by any asset);
- (v) any Contract between an Acquired Corporation and a third Person (A) for the disposition of any material assets or business of the Acquired Corporations or (B) for the acquisition of a material portion of the assets or business of any third Person (whether by merger, sale of stock or assets or otherwise), in each case (A) or (B) that contains continuing indemnities or other material obligations or any continuing royalties or other amounts calculated based upon any revenues or income of the Company or any "earn out", "milestone" or other contingent payment obligations on the part of an Acquired Corporation;
- (vi) any Contract between any Acquired Corporation and any third Person constituting a material joint venture, collaboration, partnership or similar profit or revenue sharing arrangement;
- (vii) any Contract that by its express terms requires an Acquired Corporation, or any successor to, or acquirer of, an Acquired Corporation, to make any payment to another Person as a result of a change of control of such Acquired Corporation (a "Change of Control Payment") or gives another Person a right to receive or elect to receive a Change of Control Payment;
- (viii) any Contract that prohibits the declaration or payment of dividends or distributions in respect of the capital stock of an Acquired Corporation, the pledging of the capital stock or other equity interests of an Acquired Corporation or the issuance of any guaranty by an Acquired Corporation;

- (ix) any (A) In-bound License and (B) Out-bound License;
- (x) any Government Contract;
- (xi) any Contract that relates to the research, development, distribution, marketing, supply, license, collaboration, copromotion or manufacturing of the Product Candidates, which if terminated or not renewed, would reasonably be expected to have a material and adverse effect on the Product Candidates;
 - (xii) any Contract for the lease or sublease of any real property;
 - (xiii) any Contract that relates to any swap, forward, futures, or other similar derivative transactions;
- (xiv) any other Contract that is currently in effect and has been filed (or is required to be filed) by the Company as an exhibit pursuant to Item 601(b)(10) of Regulation S-K under the Securities Act or that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act; and
- (xv) any Contract (A) with any Affiliate (other than another Acquired Corporation), director, executive officer (as such term is defined in the Exchange Act), Person holding 5% or more of the Shares, or, to the knowledge of the Company, any Affiliate (other than another Acquired Corporation) or immediate family member of any of the foregoing or (B) in which any of the foregoing Persons has a direct or indirect material financial interest.
- (b) As of the date of this Agreement, the Company has either delivered or made available to Parent an accurate, unredacted and complete copy of each Material Contract or has publicly made available such Material Contract in the Electronic Data Gathering, Analysis and Retrieval (EDGAR) database of the SEC. No Acquired Corporation nor, to the knowledge of the Company, any other party thereto is in material breach of, or material default under, any Material Contract and no Acquired Corporation, or to the knowledge of the Company, any other party to a Material Contract has taken or failed to take any action that with or without notice, lapse of time or both would constitute a material breach of or material default under any Material Contract. Each Material Contract is, with respect to the Acquired Corporations and, to the knowledge of the Company, each other party thereto, a valid and binding agreement in full force and effect, enforceable in accordance with its terms, except as such enforcement may be subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar laws of general applicability relating to or affecting creditors' rights, and by general equitable principles. Since the Applicable Date, (i) the Acquired Corporations have not received or delivered any written notice regarding any material violation or breach or default under any Material Contract that has not since been cured, and (ii) no Acquired Corporation has waived in writing any material rights under any Material Contract.
- (c) As of the date of the Agreement, there are no transactions, arrangements or Contracts between any of the Acquired Corporations, on the one hand, and its Affiliates (other than its wholly owned Subsidiaries) or other Persons, on the other hand, that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act.

2.10 Liabilities. The Acquired Corporations do not have any liabilities (whether accrued, absolute, contingent or otherwise) which would be required to be reflected or reserved against on a consolidated balance sheet of the Company prepared in accordance with GAAP or the notes thereto, except for: (a) liabilities reflected or reserved against in the financial statements or notes thereto included in the Company SEC Documents filed prior to the date of this Agreement; (b) liabilities or obligations incurred pursuant to the terms of this Agreement; (c) liabilities for performance of obligations under Contracts binding upon the Acquired Corporations (other than resulting from any breach thereof) delivered or made available to Parent prior to the date of this Agreement; (d) liabilities incurred in the ordinary course of business consistent with past practice since January 1, 2023; and (e) liabilities that have not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

2.11 Compliance with Legal Requirements.

- (a) The Acquired Corporations are, and since the Applicable Date have been, in compliance with all applicable Legal Requirements, except as would not be, individually or in the aggregate, material to the Acquired Corporations taken as a whole.
- (b) Since the Applicable Date, (i) no Acquired Corporation has received any written notice from any Governmental Body that (A) alleges any violation or noncompliance (or reflects that an Acquired Corporation is under investigation or the subject of an inquiry by any such Governmental Body for such alleged noncompliance) with any applicable Legal Requirement, or (B) threatens any fine, assessment, enforcement action, or cease and desist order, or the suspension, revocation or limitation or restriction of any Governmental Authorization held by an Acquired Corporation, and (ii) no Acquired Corporation has entered into any agreement or settlement with any Governmental Body with respect to its alleged noncompliance with, or violation of, any applicable Legal Requirement.
- (c) No Acquired Corporation produces, designs, tests, manufactures, fabricates or develops a "critical technology," as that term is defined in 31 C.F.R. Section 800.215.
- (d) For each of the past five (5) years, each Acquired Corporation has been in compliance with all applicable Global Trade Laws. No Acquired Corporation nor any Acquired Corporation's Representatives is a Sanctioned Person or is acting on behalf of a Sanctioned Person. In the past five (5) years, no Acquired Corporation has done business, directly or indirectly, with a Sanctioned Person or any Sanctioned Territory. Since the Applicable Date, no Acquired Corporation has had any direct or indirect transactions with or engaged in an activity involving a benefit provided to or received from Russia, Belarus, or the Crimea and the so-called Donetsk People's Republic and Luhansk People's Republic regions of Ukraine, including from (i) the territory of, (ii) Persons located in, or (iii) Persons incorporated, headquartered, or whose principal place of business is in those jurisdictions or regions. In the past five (5) years, no Acquired Corporation has, in connection with or relating to the business of any Acquired Corporation, (A) received from any Governmental Body any notice or (B) been subject to an investigation or made any disclosure to a Governmental Body, in either case concerning any actual or potential violations concerning applicable Global Trade Laws.

2.12 Regulatory Matters.

- (a) Each Acquired Corporation is, and since the Applicable Date has been, in compliance in all material respects with all Health Care Laws, in each case as applicable to the operation of its business. Since the Applicable Date, no Acquired Corporation, nor, to the knowledge of the Company, any third parties that have performed or are performing any development, collaboration, manufacturing, testing, quality control, batch release, distribution or shipment activities on behalf of an Acquired Corporation or with respect to a Product Candidate, including any CMO, CRO, Clinical Trial site or investigator, while acting in such capacity (each, a "Collaboration Partner") (i) has been subject to any enforcement, regulatory or administrative proceedings alleging non-compliance with any Health Care Laws; (ii) has received written notice threatening any such enforcement, regulatory or administrative proceeding; or (iii) has been issued a FDA Form 483, warning letter, notice of violation of Health Care Laws, or similar correspondence from any Governmental Body, except where any such issuance or violation would not be reasonably expected, individually or in the aggregate, to be material to the Acquired Corporations taken as a whole.
- (b) Since the Applicable Date, the Acquired Corporations have held all Regulatory Permits required for their business as then-conducted, and each Regulatory Permit required for their business as currently conducted is valid and in in full force and effect. The Acquired Corporations are in compliance in all material respects with the terms and requirements of such Regulatory Permits. Since the Applicable Date, no deficiencies have been asserted in writing by any applicable Governmental Body with respect to any Regulatory Permits of the Acquired Corporations. None of the Acquired Corporations has received written notice that any material Regulatory Permit will not or is likely not to be issued.
- (c) To the knowledge of the Company, all preclinical and clinical investigations, studies, or trials ("Clinical Trials") sponsored or conducted by or on behalf of the Acquired Corporations have been and are being conducted in material compliance with applicable Health Care Laws and Data Privacy Laws. No Acquired Corporation, nor, to the knowledge of the Company, any of their Collaboration Partners or other Persons acting directly on their behalf, has received any written notice or other correspondence from the FDA or any other Governmental Body performing functions similar to those performed by the FDA with respect to any ongoing Clinical Trial requiring or recommending a clinical hold, or the termination, suspension or material modification of any such Clinical Trials.
- (d) The Acquired Corporations have filed with the FDA or any other Governmental Body performing functions similar to those performed by the FDA all required material filings, declarations, listings, registrations, reports or submissions, including but not limited to adverse event reports, and any required updates, corrections or modifications to each of the foregoing. All such filings, declarations, listings, registrations, reports or submissions were in material compliance with applicable Health Care Laws when filed, and no deficiencies have been asserted in writing by any applicable Governmental Body with respect thereto.

- (e) To the knowledge of the Company, no Acquired Corporation, nor, to the knowledge of the Company, any Collaboration Partner or other Person acting directly on its behalf has (i) made an untrue statement of a material fact or fraudulent statement to the FDA or any Governmental Body, (ii) failed to disclose a material fact required to be disclosed to the FDA or (iii) committed any other act, made any statement or failed to make any statement, that (in any such case) establishes a reasonable basis for the FDA to invoke its Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities Final Policy or for any Governmental Body to invoke a similar policy that may be applicable in another jurisdiction to an Acquired Corporation. No Acquired Corporation is the subject of any pending or, to the knowledge of the Company, threatened investigation by the FDA pursuant to its Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities Final Policy. No Acquired Corporation nor, to the knowledge of the Company, any Collaboration Partner, officer, employee, agent or other Person acting on behalf of any Acquired Corporation has been debarred or disqualified, excluded from participation in federal healthcare programs or convicted of any crime or engaged in any conduct that would reasonably be expected to result in (A) debarment under 21 U.S.C. § 335a or any similar Legal Requirement, (B) exclusion under 42 U.S.C. § 1320a-7 or any similar Legal Requirement, or (C) disqualification pursuant to 21 CFR Part 312.70 or any similar Legal Requirement.
- (f) Since the Applicable Date, the Acquired Corporations have not received any written notice from a Governmental Body that any of their products are misbranded as defined in 21 U.S.C. § 352 or adulterated as defined in 21 U.S.C. § 351, as amended, and the rules and regulations promulgated thereunder, or as defined in comparable Legal Requirements in any jurisdiction.
- (g) Except as has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, the Acquired Corporations and, to the knowledge of the Company, their Collaboration Partners, or other Persons acting directly on their behalf have prepared, submitted and implemented timely responses and, as applicable, any corrective action plans required to be prepared and submitted in response to all internal or third-party audits, inspections, investigations or examinations of the Product Candidates or the Acquired Corporation's business.
- 2.13 Certain Business Practices. No Acquired Corporation nor, any of its directors, officers, employees, or, to the knowledge of the Company, and other Representatives (in each case, acting in the capacity of a Representative of such Acquired Corporation) has (i) used any funds (whether of an Acquired Corporation or otherwise) for unlawful contributions, gifts, entertainment or other unlawful expenses, (ii) made or offered to make any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns or (iii) accepted any unlawful payments, or (iv) violated any provision of any Anti-Corruption Laws or any rules or regulations promulgated thereunder, anti-money laundering laws or any rules or regulations promulgated thereunder or any applicable Legal Requirement of similar effect. Since the Applicable Date, no Acquired Corporation has received any written communication from a Governmental Body that alleges or gives notice of any of the foregoing.

2.14 Governmental Authorizations. The Acquired Corporations hold all material Governmental Authorizations necessary to enable the Acquired Corporations to conduct their business in the manner in which such business is currently being conducted. The material Governmental Authorizations held by the Acquired Corporations are valid and in full force and effect. The Acquired Corporations are in compliance in all material respects with the terms and requirements of such Governmental Authorizations.

2.15 Tax Matters

- (a) (i) Each of the income (or franchise) and other material Tax Returns required to be filed by or on behalf of an Acquired Corporation with any Governmental Body have been filed on or before the applicable due date (including any extensions of such due date), have been prepared in accordance with all applicable Legal Requirements, and are accurate and complete, in each case in all material respects, and (ii) all income (or franchise) and other material Taxes due and payable by an Acquired Corporation (whether or not shown on any Tax Returns) have been paid, and all material Taxes required to be withheld (including with respect to payment or allocations made to any current or former employee, independent contractor, other service provider, creditor, shareholder, customer, vendor or other third party) by any Acquired Corporation have been withheld and paid, in each case, to the relevant Governmental Body.
- (b) To the knowledge of the Company, (i) there are no pending examinations or audits of any Tax Return in progress involving material Taxes and (ii) no written claim has been received by any Acquired Corporation from any Governmental Body in any jurisdiction where such Acquired Corporation does not file Tax Returns that such Acquired Corporation is or may be subject to Taxes in that jurisdiction. No Acquired Corporation is, or has been, subject to Tax in any jurisdiction other than its place of organization by virtue of having a permanent establishment (as defined by applicable Tax treaty) or other place of business or taxable presence in that jurisdiction. No extension or waiver of the statute of limitation period applicable to any income (or franchise) or other material Tax Returns of any Acquired Corporation has been granted and is currently in effect other than automatic extensions or automatic waivers obtained in the ordinary course of business.
- (c) Adequate reserves have been established on the Company's consolidated financial statements to provide for the payment of any Taxes which are not yet due and payable with respect to any Acquired Corporation for taxable periods or portions thereof ending on or before the date of the most recent financial statements of the Company. All Taxes of the Acquired Corporations incurred through the date of the most recent financial statements of the Company do not exceed Taxes accrued on such financial statements, and all Taxes of the Acquired Corporations accrued following the end of the most recent period covered by the most recent financial statements of the Company have been accrued in the ordinary course of business.
- (d) To the knowledge of the Company, no Legal Proceeding involving the IRS or any other Governmental Body is pending, ongoing, or has been threatened in writing against or with respect to any Acquired Corporation in respect of any income or other material Taxe, and no deficiency of income or other material Taxes has been asserted in writing as a result of any audit or examination by any Governmental Body that has not been paid.

(e)	No Acquired Corporation (i) has been a member of an affiliated group (within the meaning of Section 1504(a) of the Code) filing
consolidated federal inco	ome Tax Return (other than a group the common parent of which is or was the Company or another Acquired Corporation), or
(ii) has any material liab	ility for the Taxes of any other Person (other than another Acquired Corporation) under Section 1.1502-6 of the Treasury
Regulations (or any sim	ilar provision of state, local or non-U.S. law), or as a transferee or successor or otherwise (other than pursuant to agreements not
primarily related to Taxe	es and entered into in the ordinary course of business).

- (f) During the two (2)-year period ending on the date of this Agreement, none of the Acquired Corporations has been either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code.
- (g) No Acquired Corporation has participated in any "reportable transaction" within the meaning of Treasury Regulations Section 1.6011-4(b) or any comparable provision of state, local or non-U.S. law (including European Council Directive 2011/16 ("DAC 6") and domestic implementation of DAC 6). Each Acquired Corporation has disclosed on its U.S. federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of U.S. federal income Tax within the meaning of Section 6662 of the Code (or any corresponding or similar provision of state, local or non-U.S. law).
- (h) No Acquired Corporation will be required to include any material item of income in, or exclude any material item of deduction from, the computation of taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (i) change in method of accounting for a taxable period ending on or prior to the Closing Date as a result of transactions or events occurring, or accounting methods employed, prior to the Closing, (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. income Tax law) executed prior to the Closing, (iii) installment sale, intercompany transaction or open transaction disposition made prior to the Closing, or any "excess loss account" within the meaning of the regulations under Section 1502 of the Code (or any analogous or similar provision of state, local or non-U.S. Tax law) existing as of immediately prior to the Closing, or (iv) prepaid amount or any other income eligible for deferral under the Code or Treasury Regulations promulgated thereunder (including, without limitation, pursuant to Sections 451, 455 or 456 of the Code) received on or prior to the Closing Date
- (i) None of the Acquired Corporations has made an election under Section 965(h) of the Code or otherwise has any liability for Taxes pursuant to Section 965 of the Code that has not been fully paid to the appropriate Governmental Body.

(j)	No Acquired Corporation is party to, bound by or has any obligation under any Tax allocation or Tax sharing agreement or any
express or implied Tax	indemnity or other Tax allocation agreement or arrangement with any Person, other than any agreement not primarily related to
Taxes and entered into	in the ordinary course of business.

- (k) There are no material Encumbrances with respect to Taxes upon any of the assets or properties of any Acquired Corporation, other than Permitted Encumbrances.
- (l) Section 2.15(l) of the Company Disclosure Schedule discloses for each Acquired Corporation (i) all applied for or received loans or payments under the CARES Act and COVID Relief Programs, (ii) Tax credits claimed under the CARES Act and COVID Relief Programs and (iii) any other credits, deferrals (including any deferral of the employer portion of any payroll Taxes) or benefits available that any Acquired Corporation availed itself of under the CARES Act and COVID Relief Programs. None of the Acquired Corporations will have any liability for Taxes with respect to any relief claimed under the CARES Act and COVID Relief Programs for any Tax period (or portion thereof) prior to the Closing in a Tax period (or portion thereof) beginning after the Closing.
- (m) Each Acquired Corporation has timely and accurately filed all required foreign accounts on a Report of Foreign Bank Accounts on FinCEN Form 114.
- (n) Except as otherwise set forth in Section 2.15(n) of the Company Disclosure Schedule, none of the Acquired Corporations has been or is currently the beneficiary of a Tax holiday (or similar benefit) from any Governmental Body. The Acquired Corporations are, and have been, in full compliance with any Tax holiday (or similar benefit) from any Governmental Body disclosed in Section 2.15(n) of the Company Disclosure Schedule.
- (o) None of the Acquired Corporations owns, directly or indirectly, any equity interest in a "passive foreign investment company" within the meaning of Section 1297 of the Code ("PFIC") or has ever been classified as a PFIC.
- (p) All transactions entered into or conducted by any Acquired Corporation with any Person who is a member of the "controlled group" (within the meaning of Treasury Regulations Section 1.482-1 or any similar or comparable provision of any state, local or non-U.S. law) of which such Acquired Corporation is a member have been made in material compliance with Section 482 of the Code and the Treasury Regulations thereunder and no Governmental Body has imposed or, to the knowledge of the Company, has a basis to impose, any different terms on any such transactions. None of the Acquired Corporations is a party to any advance pricing agreement, or any similar contract or agreement, with any Governmental Body.
- (q) None of the Acquired Corporations has filed (or has had filed on its behalf) a request for (i) private letter ruling, (ii) technical advice, or (iii) change of any method of accounting, in each case, with any Governmental Body that relates to Taxes or Tax Returns of any of the Acquired Corporations.

- (r) None of the Acquired Corporations has (i) entered into any settlement or arrangement (including a closing agreement within the meaning of Section 7121 of the Code or a comparable or similar provision of state, local or non-U.S. law) with any Governmental Body that would be binding and result in a material Tax liability for any Tax period (or portion thereof) ending after the Closing Date or (ii) granted any power of attorney with respect to any Taxes that will be in force following the Closing.
- (s) None of the Acquired Corporations violated any antiboycott prohibitions, or failed to comply with the reporting requirements, of the Tax Reform Act of 1976 (26 U.S.C. § 999).

2.16 Employee Matters

- (a) Neither the Company nor any of its Subsidiaries is, nor at any time since the Applicable Date has been, a party to, bound by or negotiated, any collective bargaining agreement or other Contract with a labor organization or works council representing any of its employees, and no employees of the Company or its Subsidiaries are or have been represented by any labor union or other labor organization or works council with respect to their employment with the Company or its Subsidiaries. No Acquired Corporation has, or at any time since the Applicable Date had, any duty to bargain with any labor organization or works council. To the knowledge of the Company, there have never been any activities or proceedings of any labor union to organize any current or former employees of the Acquired Corporations. No labor dispute, strike or work stoppage against any Acquired Corporation is now pending, has occurred in the past, or, to the knowledge of the Company, is now threatened that would reasonably be expected to interfere with the business activities of the Acquired Corporations. No Acquired Corporation is, or has since the Applicable Date, engaged in any unfair labor practice.
- (b) The Acquired Corporations are, and at all times since the Applicable Date have been, in compliance in all material respects with all Legal Requirements governing the employment of labor, including Legal Requirements relating to employment practices, wages, compensation and benefits, hours, classification of employees and independent contractors, overtime and overtime payment, working during rest days, social benefits contributions, severance pay, affirmative action, collective bargaining, discrimination, harassment, retaliation, civil rights, disability rights or benefits, terms and conditions of employment, immigration, safety and health, hiring, promotions, plant closings, and termination of service, severance, gratuity, pay transparency and disclosures, pay equity, privacy, leaves of absence, paid sick leave, unemployment insurance, child labor, whistleblowing, pension insurance, medical insurance, work-related-injury insurance, maternity insurance, contributions to the public housing fund, the withholding and payment of social security and other Taxes, and COVID-19 ("Employment Practices"), such as the Age Discrimination in Employment Act, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, ERISA, the Fair Labor Standards Act, the Americans with Disabilities Act, the Occupational Safety and Health Act, the Family and Medical Leave Act, the National Labor Relations Act of 1935, Executive Order 11246 and any other executive orders or regulations governing affirmative action, EEO and VETS-100 reporting obligations, and the Immigration and Nationality Act and other similar Legal Requirements of the jurisdictions in which the Acquired Corporations are qualified or do business.

- (c) There are no, nor have there been at any time since the Applicable Date any, Legal Proceedings pending or, to the knowledge of the Company, threatened or reasonably anticipated involving any current or former employee or independent contractor of any Acquired Corporation (collectively, "Workers") or group of Workers or involving non-compliance with Employment Practices. No Acquired Corporation is, nor at any time since the Applicable Date has been, a party to a conciliation agreement, consent decree or other agreement or order with any Governmental Body with respect to Employment Practices. Since the Applicable Date, no current or former Worker has filed a complaint or claim with an Acquired Corporation with respect to Employment Practices.
- (d) No Acquired Corporation has at any time since the Applicable Date taken any action that (i) would constitute a "mass layoff" or "plant closing" within the meaning of the Worker Readjustment and Notification Act (29 U.S.C. § 2101) (the "WARN Act") or would otherwise trigger notice requirements or liability under any other Legal Requirement that is comparable to the WARN Act or (ii) resulted in the termination of employment of 50 or more Workers or more than 10% of the Workers of the Acquired Corporations during any 90-day period. No Acquired Corporation has any plans to undertake any action that would trigger the WARN Act or other similar notification requirements. The Acquired Corporations are, and have been at all times since the Applicable Date, in full compliance with the WARN Act and any applicable state laws or other Legal Requirements regarding redundancies, reductions in force, mass layoffs, and plant closings, including all obligations to promptly and correctly furnish all notices required to be given thereunder in connection with any redundancy, reduction in force, mass layoff, or plant closing to affected employees, representatives, any state dislocated worker unit and local government officials, or any other Governmental Body.
- (e) To the knowledge of the Company, no current or former Worker is or has been at any time since the Applicable Date in any material respect in violation of any employment contract, non-disclosure, confidentiality agreement, non-competition agreement, non-solicitation agreement, restrictive covenant, or consulting agreement with an Acquired Corporation. To the knowledge of the Company, no current or former Worker has been at any time since the Applicable Date in any material respect in violation of any employment contract, non-disclosure, confidentiality agreement, non-competition agreement, non-solicitation agreement, restrictive covenant or consulting agreement with a former employer or service recipient relating to the right of any such Worker to be employed by or provide services to the Acquired Corporations because of the nature of the business conducted or presently proposed to be conducted by it or to the use of trade secrets or proprietary information of others.
- (f) All employees of the Acquired Corporations are employed on an at will basis, except as provided by applicable Legal Requirements outside of the United States.

2.17 Benefit Plans

(a) Section 2.17(a) of the Company Disclosure Schedule sets forth an accurate and complete list of the material Employee Plans (other than (i) any employment agreements for non-officer employees of the Company or its Subsidiaries that do not provide for any severance rights, (ii) equity grant notices with respect to awards disclosed on Section 2.3(g) of the Company Disclosure Schedule and that do not materially deviate from the forms delivered or made available to Parent prior to the execution of this Agreement in accordance with Section 2.17(k), or (iii) agreements with individual consultants entered into in the ordinary course of business consistent with past practice). To the extent applicable, the Company has either delivered or made available to Parent prior to the execution of this Agreement with respect to each material Employee Plan accurate and complete copies of: (A) all plan documents and all amendments thereto, and all related trust or other funding documents, and in the case of unwritten material Employee Plans, written descriptions thereof, (B) the most recent annual actuarial valuation, if any, and the most recent annual report (Form Series 5500 and all schedules and financial statements attached thereto), (C) all material correspondence to or from the IRS, the United States Department of Labor or any other Governmental Body with respect to an Employee Plan, (D) all determination letters, rulings, opinion letters, information letters or advisory opinions issued by the IRS or the United States Department of Labor, and (E) the most recent summary plan descriptions and any material modifications thereto.

- (b) Neither the Company nor any ERISA Affiliate has ever maintained, contributed to, been required to contribute to, or otherwise incurred any liability with respect to, (i) a plan subject to Title IV or Section 302 of ERISA or Code Section 412 or 4971, including any "single employer" defined benefit plan or any "multiemployer plan," each as defined in Section 4001 of ERISA, or (ii) a plan that has two or more contributing sponsors at least two of whom are not under common control, within the meaning of Section 4063 of ERISA.
- (c) Each of the Employee Plans that is intended to be qualified under Section 401(a) of the Code has obtained a favorable determination letter (or opinion letter, if applicable) as to its qualified status under the Code, each such Employee Plan has timely adopted all currently effective amendments to the Code, and, to the knowledge of the Company, there are no existing circumstances or any events that have occurred that would reasonably be expected to affect the qualified status of any such Employee Plan.
- (d) Each of the Employee Plans is now and has been operated in compliance in all material respects with its terms and all applicable Legal Requirements, including but not limited to ERISA and the Code.
- (e) There are no pending or threatened claims (other than routine claims for benefits in the ordinary course of business consistent with past practice) or Legal Proceedings, and, to the knowledge of the Company, no set of circumstances exists that may reasonably give rise to a claim or Legal Proceeding, against the Employee Plans, any fiduciaries thereof or the assets of any related trusts. No Employee Plan is under audit or, to the Company's knowledge, the subject of an investigation by the IRS, the U.S. Department of Labor, the Pension Benefit Guaranty Corporation, the SEC or any other Governmental Body, nor is any such audit or investigation pending or, to the knowledge of the Company, threatened. The Company is not and could not reasonably be expected to be subject to either a material liability pursuant to Section 502 of ERISA or a material Tax imposed pursuant to Section 4975 or 4976 of the Code.

- (f) Each Employee Plan that is in any part a "nonqualified deferred compensation plan" subject to Section 409A of the Code complies both in form and operation with the requirements of Section 409A of the Code in all material respects.
- (g) All contributions required to be made to any Employee Plan by applicable Legal Requirements or otherwise, and all premiums due or payable with respect to insurance policies funding any Employee Plan, have been timely made or paid in full or, to the extent not required to be made or paid on or before the date hereof, have been fully reflected on the financial statements of the Company in accordance with GAAP.
- (h) Except to the extent required under Section 601 et seq. of ERISA or 4980B of the Code (or any other similar state or local Legal Requirement), no Acquired Corporation, ERISA Affiliate nor any Employee Plan has any present or future obligation to provide post-employment welfare benefits to or make any payment to, or with respect to, any present or former employee, officer, director or service provider of the Acquired Corporations pursuant to any retiree medical benefit plan or other retiree welfare plan.
- (i) Except as provided in Section 1.8, the consummation of the Transactions (including in combination with other events or circumstances) will not (i) entitle any current or former employee, director, officer, independent contractor or other service provider of an Acquired Corporation to severance pay, unemployment compensation or any other material payment, (ii) accelerate the time of payment or vesting, or increase the amount of, compensation or benefits due to any such employee, director, officer, independent contractor, (iii) directly or indirectly cause an Acquired Corporation to transfer or set aside any material assets to fund any payments or benefits under any Employee Plan, or (iv) result in any limitation on the right of an Acquired Corporation to amend, merge, terminate or receive a reversion of assets from any Employee Plan or related trust. Without limiting the generality of the foregoing, no amount paid or payable (whether in cash, in property, or in the form of benefits) by an Acquired Corporation in connection with the Transactions (whether alone or in combination with other events or circumstances) will be an "excess parachute payment" within the meaning of Section 280G of the Code.
 - (j) No Employee Plan provides for the gross-up or reimbursement of Taxes under Section 4999 or 409A of the Code, or otherwise.
- (k) The Company has made available to Parent copies of the Company Equity Plans and the forms of all agreements and instruments relating to or issued under the Company Equity Plans. Each outstanding Company Option has an exercise price equal to or above the fair market value on the date of grant (within the meaning of Section 409A of the Code) and is otherwise not subject to Section 409A of the Code.
- (l) No Employee Plan is subject to the Legal Requirements of a jurisdiction other than the United States (whether or not United States Legal Requirements also apply) or covers employees or other service providers of any Acquired Corporation working primarily outside the United States.

2.18 Environmental Matters.

- (a) The Acquired Corporations are and since the Applicable Date have been in compliance, in all material respects, with all applicable Environmental Laws, which compliance includes obtaining, maintaining or complying with all Governmental Authorizations required under Environmental Laws for the operation of their business.
- (b) As of the date of this Agreement, there is no Legal Proceeding relating to or arising under any Environmental Law that is pending or, to the knowledge of the Company, threatened against any Acquired Corporation or in respect of any Leased Real Property, except as would not be material to the Acquired Corporations taken as a whole.
- (c) Since the Applicable Date through the date of this Agreement, no Acquired Corporation has received any written notice, report or other information of or entered into any legally binding agreement, order, settlement, judgment, injunction or decree involving uncompleted, outstanding or unresolved material violations, liabilities or requirements on the part of any Acquired Corporation relating to or arising under Environmental Laws.
- (d) To the knowledge of the Company, there are, and since the Applicable Date have been, no Hazardous Materials present or Releases on, at, under or from any property or facility, including the Leased Real Property, in a manner and concentration that would reasonably be expected to result in any material claim against or material liability of an Acquired Corporation under any Environmental Law.
- (e) No Acquired Corporation has assumed, undertaken, or otherwise become subject to any material liability of another Person relating to Environmental Laws.
- 2.19 Insurance. The Company has delivered or made available to Parent accurate and complete copies of all material insurance policies and self-insurance programs relating to the business, assets and operations of the Acquired Corporations. The Acquired Corporations maintain insurance coverage in such amounts and covering such risks as are in accordance, in all material respects, with normal industry practice for companies of similar size and stage of development. All such insurance policies are in full force and effect, no notice of termination, cancellation, non-renewal or material modification has been received (other than a notice in connection with ordinary renewals), all premiums due and payable thereon have been paid in accordance with the terms of such policies and there is no existing material default or event which, with the giving of notice or lapse of time or both, would constitute a material default, by any insured thereunder. As of the date of this Agreement, there is no claim pending under any of the Company's insurance policies as to which coverage has been questioned, denied or disputed by the underwriters of such policies, except as would not be, or would not reasonably be expected to be, material to the Acquired Corporations taken as a whole.

2.20 Legal Proceedings; Orders.

(a) There are no Legal Proceedings pending and served (or, to the knowledge of the Company, pending and not served or threatened) against any Acquired Corporation or, to the knowledge of the Company, against any present or former officer, director or employee of an Acquired Corporation in such individual's capacity as such, except as, individually or taken together (i) would not be, or would not reasonably be expected to be, material to the Acquired Corporations taken as a whole and (ii) would not reasonably be expected to prevent, materially delay or materially impair the ability of the Company to perform its obligations under this Agreement or to consummate the Transactions in a timely manner.

- (b) There is no material order, writ, injunction or judgment to which an Acquired Corporation is subject.
- (c) To the knowledge of the Company, no material investigation or review by any Governmental Body with respect to an Acquired Corporation is pending or is being threatened.
- 2.21 Authority; Binding Nature of Agreement. The Company has the corporate power and authority to execute and deliver, and to perform its obligations under, this Agreement and to consummate the Transactions, including the Merger, subject to the adoption of this Agreement by holders of at least a majority of the outstanding shares of Company Common Stock entitled to vote thereon (the "Company Stockholder Approval"). The Company has taken all corporate action necessary, and no other corporate actions or proceedings on the part of the Company or vote of the Company's stockholders is necessary to authorize the consummation of the Transactions, other than, in the case of the Merger, the Company Stockholder Approval. The Board of Directors has unanimously (a) determined that this Agreement and the Transactions, including the Merger, are advisable and fair to, and in the best interest of, the Company and its stockholders, (b) declared it advisable to enter into this Agreement, (c) approved the execution, delivery and performance by the Company of this Agreement and the consummation of the Transactions, including the Merger, (d) resolved that the Merger shall be governed by Section 251(c) of the DGCL, upon the terms and subject to the conditions set forth in this Agreement, and (e) resolved to recommend that the stockholders of the Company adopt this Agreement at any meeting of the Company's stockholders held for such purpose and any adjournment or postponement thereof on the terms and subject to the conditions of this Agreement. The resolutions in the foregoing sentence, subject to the Board of Directors effecting a Company Adverse Change Recommendation in accordance with Section 5.1, have not been subsequently withdrawn or modified in a manner adverse to Parent. This Agreement has been duly executed and delivered by the Company, and assuming due authorization, execution and delivery by Guarantor, Parent and Merger Sub, this Agreement constitutes the legal, valid and binding obligation of the Company and is enforceable against the Company in accordance with its terms, except as such enforcement may be subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar laws of general applicability relating to or affecting creditors' rights, and by general equitable principles.
- 2.22 Takeover Laws. Assuming the accuracy of the representations and warranties of Parent and Merger Sub set forth in Section 3.8, the Board of Directors has taken and will take all actions so that the restrictions applicable to business combinations contained in Section 203 of the DGCL and any other Takeover Law are, and will be inapplicable to the execution, delivery and performance of this Agreement and to the consummation of the Merger and the other Transactions.

2.23 Non-Contravention; Consents.

- (a) Assuming compliance with the applicable provisions of the DGCL, the HSR Act, any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any), and the rules and regulations of the SEC and Nasdaq, the execution and delivery of this Agreement by the Company and the consummation of the Transactions will not: (i) cause a violation of any of the provisions of the certificate of incorporation or bylaws (or other organizational documents) of any Acquired Corporation; (ii) cause a violation by any Acquired Corporation of any Legal Requirement or order applicable to an Acquired Corporation, or to which an Acquired Corporation is subject; (iii) require any consent or notice under, conflict with, result in breach of, or constitute a default under (or an event that with notice or lapse of time or both would become a default), or give rise to any right of purchase, termination, amendment, cancellation, acceleration or other adverse change of any right or obligation or the loss of any benefit to which an Acquired Corporation is entitled under any provision of any Contract; or (iv) result in an Encumbrance (other than a Permitted Encumbrance) on any of the property or assets of any Acquired Corporation; except, in the case of clauses (iii) and (iv), as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- (b) Except for the filing of the certificate of merger with the Secretary of State of the State of Delaware or as may be required by the Exchange Act (including the filing with the SEC of the Merger Proxy Statement and such reports under the Exchange Act as may be required in connection with this Agreement and the Transactions), the DGCL, the HSR Act and any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any) and the applicable rules and regulations of the SEC and any national securities exchange, the Acquired Corporations are not required to give notice to, make any filing with, or obtain any Consent from any Governmental Body at any time prior to the Closing in connection with the execution and delivery of this Agreement by the Company, or the consummation by the Company of the Merger or the other Transactions, except those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **2.24** Opinion of Financial Advisors. The Board of Directors has received the oral opinion of each of BofA Securities, Inc. and Centerview Partners LLC (together, the "Company Financial Advisors"), to be subsequently confirmed by delivery of each Company Financial Advisor's written opinion to the Board of Directors to the effect that, as of the date of the opinion and subject to the factors and assumptions set forth in such written opinion, the Merger Consideration to be paid to the holders of Shares (other than excluded shares as provided in such opinions) in the Merger pursuant to this Agreement is fair, from a financial point of view, to such holders. The Company will make available to Parent solely for informational purposes and on a non-reliance basis, a signed copy of each such written opinion as soon as possible following the date of this Agreement.

- **2.25 Brokers and Other Advisors**. Except for the Company Financial Advisors, no broker, finder, investment banker, financial advisor or other Person is entitled to any brokerage, finder's, financial advisor's or other similar fee or commission, or the reimbursement of expenses in connection therewith, in connection with the Transactions based upon arrangements made by or on behalf of the Company. The Company has made available to Parent accurate copies of all agreements under which any such fee, commission or expense reimbursement is payable and all indemnification and other agreements related to the engagement of the persons to whom any such fee is payable.
- 2.26 Acknowledgment by Company. The Company is not relying and has not relied on any representations or warranties whatsoever regarding the Transactions or the subject matter of this Agreement, express or implied, except for the representations and warranties of Guarantor set forth in Section 8.10(b) and of Parent and Merger Sub set forth in Section 3. Such representations and warranties by Guarantor, Parent and Merger Sub constitute the sole and exclusive representations and warranties of Guarantor, Parent and Merger Sub in connection with the Transactions and the Company understands, acknowledges and agrees that all other representations and warranties of any kind or nature whether express, implied or statutory are specifically disclaimed by Guarantor, Parent and Merger Sub.

SECTION 3

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub represent and warrant to the Company as follows:

- 3.1 Due Organization. Each of Parent and Merger Sub is a corporation or other Entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all necessary power and authority: (a) to conduct its business in the manner in which its business is currently being conducted; and (b) to own and use its assets in the manner in which its assets are currently owned and used, except as would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- 3.2 Merger Sub. Merger Sub was formed solely for the purpose of engaging in the Transactions and activities incidental thereto and has not engaged, and prior to the Effective Time will not engage, in any business activities or conducted any operations other than as contemplated by this Agreement in connection with the Transactions and those incident to Merger Sub's formation and capitalization. Either Parent or a wholly owned Subsidiary of Parent owns beneficially and of record all of the outstanding capital stock of Merger Sub, free and clear of all Encumbrances and transfer restrictions, except for Encumbrances or transfer restrictions of general applicability as may be provided under the Securities Act or applicable securities laws.
- 3.3 Authority; Binding Nature of Agreement. Parent and Merger Sub each have the corporate power and authority to execute and deliver and perform their respective obligations under this Agreement and to consummate the Transactions. The board of directors of each of Parent and Merger Sub have approved the execution, delivery and performance by Parent and Merger Sub of this Agreement and the consummation of the Transactions, including the Merger. This Agreement has been duly executed and delivered by Parent and Merger Sub, and assuming due authorization, execution and delivery by the Company, this Agreement constitutes the legal, valid and binding obligation of Parent and Merger Sub and is enforceable against Parent and Merger Sub in accordance with its terms, except as such enforcement may be subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar laws of general applicability relating to or affecting creditors' rights, and by general equitable principles.

3.4 Non-Contravention; Consents.

- (a) Assuming compliance with the applicable provisions of the DGCL, the HSR Act and any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any), and the rules and regulations of the SEC and Nasdaq, the execution and delivery of this Agreement by Parent and Merger Sub, and the consummation of the Transactions, will not: (i) cause a violation of any of the provisions of the certificate of incorporation or bylaws (or other organizational documents) of Parent or Merger Sub; (ii) cause a violation by Parent or Merger Sub of any Legal Requirement or order applicable to Parent or Merger Sub, or to which Parent or Merger Sub are subject; or (iii) require any consent or notice under, conflict with, result in breach of, or constitute a default under (or an event that with notice or lapse of time or both would become a default), or give rise to any right of purchase, termination, amendment, cancellation, acceleration or other adverse change of any right or obligation or the loss of any benefit to which Parent or Merger Sub is entitled under any provision of any Contract, except in the case of clauses (ii) and (iii), as would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- (b) Except for the filing of the certificate of merger with the Secretary of State of the State of Delaware or as may be required by the Exchange Act, Takeover Laws, the DGCL, the HSR Act and any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any) and the applicable rules and regulations of the SEC and any national securities exchange, neither Parent nor Merger Sub, nor any of Parent's other Affiliates, is required to give notice to, make any filing with or obtain any Consent from any Governmental Body at any time prior to the Closing in connection with the execution and delivery of this Agreement by Parent or Merger Sub or, as provided by Section 8.10(b), Guarantor, or the consummation by Parent or Merger Sub of the Merger or the other Transactions, except those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect. No vote of Parent's, Merger Sub's or Guarantor's stockholders is necessary to approve this Agreement or any of the Transactions (except as has been obtained prior to the execution and delivery of this Agreement).
- 3.5 Disclosure. None of the information with respect to Parent or Merger Sub supplied or to be supplied by or on behalf of Parent or Merger Sub or any of their Subsidiaries, specifically for inclusion or incorporation by reference in the Merger Proxy Statement will, (a) at the time such document is filed with the SEC, (b) at any time such document is amended or supplemented or (c) at the time such document is first published, sent or given to the Company's stockholders, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. For clarity, the representations and warranties in this Section 3.5 will not apply to statements or omissions included or incorporated by reference in the Merger Proxy Statement based upon information supplied by the Company to Parent or Merger Sub or any of their respective Representatives on behalf of the Company specifically for inclusion therein.

- 3.6 Absence of Litigation. As of the date of this Agreement, there is no Legal Proceeding pending and served or, to the knowledge of Parent, pending and not served, against Parent or Merger Sub, except as would not, and would not reasonably be expected to, individually or in the aggregate, have a Parent Material Adverse Effect. To the knowledge of Parent or Merger Sub, as of the date of this Agreement, neither Parent nor Merger Sub is subject to any continuing order of, consent decree, settlement agreement or similar written agreement with, or continuing investigation by, any Governmental Body, or any order, writ, judgment, injunction, decree, determination or award of any Governmental Body, except as would not, and would not reasonably be expected to, individually or in the aggregate, have a Parent Material Adverse Effect.
- 3.7 Funds. Guarantor has, and as of the Effective Time will have, through cash, marketable investments and existing credit facilities (and will make available to Parent and Merger Sub in a timely manner) immediately available funds in cash in an amount sufficient to carry out all of Parent's and Merger Sub's obligations under this Agreement and to consummate the Transactions by payment in cash of the aggregate Merger Consideration payable following the Effective Time and the aggregate amounts payable to holders of Company Options, Company RSUs and Company PSUs following the Effective Time pursuant to Sections 1.8(a), 1.8(c) and 1.8(d).
- 3.8 Ownership of Shares. Except as contemplated by this Agreement, neither Parent nor any of Parent's Affiliates, directly or indirectly, owns, and at all times for the past three (3) years, neither Parent nor any of Parent's Affiliates has owned, beneficially or otherwise, any Shares or any securities, Contracts or obligations convertible into or exercisable or exchangeable for Shares. Neither Parent nor any of Parent's Affiliates is, nor for the past three (3) years has been, an "interested stockholder" of the Company under Section 203(c) of the DGCL.

3.9 Acknowledgement by Parent and Merger Sub.

- (a) Neither Parent nor Merger Sub is relying and neither Parent nor Merger Sub has relied on any representations or warranties whatsoever regarding the Transactions or the subject matter of this Agreement, express or implied, except for the representations and warranties of the Company set forth in Section 2, including the Company Disclosure Schedule. Such representations and warranties by the Company constitute the sole and exclusive representations and warranties of any of the Acquired Corporations in connection with the Transactions and each of Parent and Merger Sub understands, acknowledges and agrees that all other representations and warranties of any kind or nature whether express, implied or statutory are specifically disclaimed by the Acquired Corporations.
- (b) In connection with the due diligence investigation of the Acquired Corporations by Parent and Merger Sub and their respective Affiliates, stockholders or Representatives, Parent and Merger Sub and their respective Affiliates, stockholders and Representatives have received and may continue to receive after the date hereof from the Company, the other Acquired Corporations and their respective Affiliates and Representatives certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Acquired Corporations and their respective businesses and operations. Parent and Merger Sub hereby acknowledge that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, and agree that, except in the case of fraud, Parent and Merger Sub will have no claim against the Acquired Corporations, or any of their respective Affiliates, stockholders or Representatives, or any other Person with respect thereto unless any such information is expressly included in a representation or warranty contained in this Agreement. Accordingly, Parent and Merger Sub hereby acknowledge and agree that neither the Acquired Corporations nor any of their respective Affiliates, stockholders or Representatives, nor any other Person, has made or is making any express or implied representation or warranty with respect to such estimates, projections, forecasts, forward-looking statements or business plans unless any such information is expressly included in a representation or warranty contained in this Agreement.

3.10 Brokers and Other Advisors. Except for Persons, if any, whose fees and expenses shall be paid by Parent or Merger Sub, no broker, finder, investment banker, financial advisor or other Person is entitled to any brokerage, finder's, financial advisor's or other similar fee or commission, or the reimbursement of expenses in connection therewith, in connection with the Transactions based upon arrangements made by or on behalf of Parent, Merger Sub, or any of their respective Subsidiaries.

SECTION 4

CERTAIN COVENANTS OF THE COMPANY

4.1 Access and Investigation; Notice of Certain Events.

During the period from the execution and delivery of this Agreement until the earlier of the Effective Time and the valid termination of this Agreement pursuant to Section 7 (the "Pre-Closing Period"), upon reasonable advance notice to the Company, the Company shall, and shall cause the other Acquired Corporations and the respective Representatives of the Acquired Corporations to, provide Parent and Parent's Representatives with reasonable access to the Acquired Corporations' Representatives and to their properties and assets, and to all existing books, records, documents and information relating to the Acquired Corporations, and promptly provide Parent and Parent's Representatives with all reasonably requested information regarding the business of the Acquired Corporations and such additional financial, operating and other data and information regarding the Acquired Corporations, as Parent may reasonably request, in each case for any reasonable business purpose related to the Transactions and the integration of the Acquired Corporations into the operations of Parent; provided, however, that any such access to the Acquired Corporations' assets and personnel shall be conducted at Parent's expense, at a reasonable time, and under the supervision of appropriate personnel of the Acquired Corporations and all such information shall be requested in such a manner as not to unreasonably interfere with the normal operation of the business of the Acquired Corporations. Nothing herein shall require any of the Acquired Corporations to disclose any information to Parent if such disclosure would, in the Company's reasonable discretion (after consultation with outside counsel) and after notice to Parent (i) jeopardize any attorney-client or other legal privilege (so long as the Acquired Corporations have reasonably cooperated with Parent and used their reasonable best efforts to permit such inspection of or to disclose such information on a basis that does not waive such privilege with respect thereto), (ii) contravene any applicable Legal Requirement (so long as the Acquired Corporations have reasonably cooperated with Parent and used their reasonable best efforts to permit disclosure to the extent permitted by Legal Requirements, e.g., through the use of customary "clean room" arrangements pursuant to which certain Representatives of Parent could be provided access to such information) or (iii) contravene any Contract to which an Acquired Corporation is a party or by which an Acquired Corporation is bound as of the date of this Agreement (so long as the Acquired Corporations have reasonably cooperated with Parent and used their reasonable best efforts to permit disclosure on a basis that would not contravene such Contract). Notwithstanding the foregoing, nothing in this Section 4.1 shall require an Acquired Corporation to disclose any information to Parent or Parent's Representatives to the extent such information relates to the applicable portions of the minutes of the meetings of the Board of Directors or any committee thereof (including any presentations or other materials prepared by or for the Board of Directors or such committee thereof) in which the Board of Directors or committee thereof discussed (x) the Transactions, or any similar transaction involving an Acquired Corporation, (y) any Acquisition Proposal or (z) a Company Adverse Change Recommendation; provided that upon Parent's request in accordance with this Section 4.1, the Company shall disclose versions of such minutes, presentations or other materials to Parent and its Representatives in which such information is redacted. With respect to the information disclosed pursuant to this Section 4.1, Parent shall comply with, and shall instruct Parent's Representatives to comply with, all of its obligations under the Confidential Disclosure Agreement dated May 22, 2022, between the Company and Guarantor (as amended, the "Confidentiality Agreement").

(b) Subject to applicable Legal Requirements, during the Pre-Closing Period, each of the Company and Parent shall promptly notify
the other in writing of (i) any notice or other communication received by such Party from any Governmental Body in connection with this Agreement, the
Merger or the other Transactions, or from any Person alleging that the consent of such Person is or may be required in connection with the Merger or the
other Transactions; or (ii) any Legal Proceeding commenced or, to such Party's knowledge, threatened in writing, against the other Party or any of its
Subsidiaries or otherwise relating to, involving or affecting such Party or any of its Subsidiaries, in each case in connection with, arising from or otherwise
relating to the Merger or any other Transaction. In furtherance of the foregoing, during the Pre-Closing Period, the Company shall promptly furnish Parent
a copy of each report, schedule and other document filed or submitted by it pursuant to the requirements of federal or state securities Legal Requirements
and a copy of any communication received by the Company from the SEC concerning compliance with securities laws with respect to matters unrelated to
the Transactions.

(c) No investigation conducted or notice given pursuant to this <u>Section 4.1</u> shall affect or be deemed to qualify, modify or limit, or cure the breach of, any representation or warranty made by the Company in this Agreement.

- **4.2 Operation of the Acquired Corporations' Business**. During the Pre-Closing Period, except (x) as expressly required under this Agreement or as required by applicable Legal Requirements, (y) with the prior written consent of Parent (which consent shall not be unreasonably withheld, conditioned or delayed), or (z) as set forth in Section 4.2 of the Company Disclosure Schedule:
- (a) the Company shall, and shall cause each other Acquired Corporation to, use commercially reasonable efforts to (i) conduct its business in the ordinary course consistent with past practice in all material respects, (ii) preserve intact its material assets (including technology), properties, Contracts, licenses and business organization, (iii) keep available the services of its present executive officers and key employees and (iv) maintain satisfactory business relationships with licensors, licensees, suppliers, lessors, Governmental Bodies and others having material business dealings with the Acquired Corporations; and
 - (b) the Acquired Corporations shall not:
- (i) (A) establish a record date for, declare, set aside or pay any dividend or make any other distribution in respect of any shares of its capital stock (including the Shares) or enter into any Contract with respect to the voting of its securities, or (B) repurchase, redeem or otherwise reacquire any of the Shares, or any rights, warrants or options to acquire any of the Shares, other than: (1) repurchases of Shares outstanding as of the date hereof pursuant to the Company's right (under written commitments in effect as of the date hereof that have been made available to Parent) to purchase Shares held by a Company Associate only upon termination of such Person's employment or engagement by the Company; (2) repurchases or forfeitures of Company Options, Company PSUs (or, in each case, Shares issued upon the exercise thereof) outstanding on the date hereof (in cancellation thereof) pursuant to the terms of any such Company Option, Company RSU or Company PSU (as in effect as of the date hereof) between the Company and a Company Associate or member of the Board of Directors upon termination of such Person's employment or engagement by the Company; (3) in connection with withholding to satisfy the exercise price and/or Tax obligations with respect to Company Options, Company RSUs or Company PSUs pursuant to the terms thereof (as in effect as of the date hereof) or (4) between the Company and a wholly owned Acquired Corporation or between wholly owned Acquired Corporations;
 - (ii) split, combine, subdivide or reclassify any Shares or other equity interests;
- (iii) sell, issue, grant, deliver, pledge, transfer, encumber, dispose of, or otherwise divest or authorize the sale, issuance, grant, delivery, pledge, transfer, encumbrance disposal or divestiture of (A) any capital stock, equity interest or other security, (B) any option, call option, warrant, restricted securities or other right to acquire any capital stock, equity interest or other security, or (C) any instrument convertible into or exchangeable for any capital stock, equity interest or other security (except (x) that the Company may issue Shares as required to be issued upon the exercise or vesting (as the case may be) of Company Options, Company RSUs or Company PSUs outstanding as of the date of this Agreement pursuant to the terms thereof (as in effect as of the date hereof), or, subject to the Company's compliance with Section 5.4, issuable to participants in the Company ESPP in accordance with the terms thereof (as in effect on the date hereof), or (y) with respect to pledges or encumbrances (or authorizations with respect any of the foregoing) constituting (1) Permitted Encumbrances under clause (f) of the definition thereof that are in existence as of the date of this Agreement or (2) Encumbrances created or incurred in connection with any Indebtedness permitted to be established or incurred under Section 4.2(b)(ix));

- (iv) except as set forth in Section 1.8, as required under any Employee Plan as in effect on the date of this Agreement made available to Parent, or as set forth in Section 4.2(b)(iv) of the Company Disclosure Schedule, (A) establish, adopt, enter into, terminate or amend any Employee Plan (or any plan, program, arrangement or agreement that would be an Employee Plan if it were in existence on the date hereof), (B) amend or waive any of their rights under, or accelerate the payment or vesting under, any provision of any of the Employee Plans (or any plan, program, arrangement or agreement that would be an Employee Plan if it were in existence on the date hereof), including any Company Options, Company RSUs or Company PSUs; (C) grant, provide, promise, amend or increase for any employee or director any compensation, bonuses, severance, retention or other payments or benefits; (D) hire or retain any person for employment or to be a consultant with an Acquired Corporation at the level of officer or above, other than hiring an employee (x) to fill a position of an officer that becomes vacant after the date of this Agreement, and (y) with cash compensation and benefits (other than benefits upon severance or a change in control) consistent with past practice; or (E) forgive any loans or other amounts payable to the Company or any of its Subsidiaries by any director, officer, employee or individual consultant of the Company or any of its Subsidiaries;
- (v) amend or permit the adoption of any amendment to its certificate of incorporation or bylaws or other charter or organizational documents;
- (vi) form any Subsidiary, acquire (including by merger, consolidation, acquisition of stock or assets or otherwise), in one transaction or any series of related transactions, directly or indirectly, any equity interest in, or material assets, businesses (or any division thereof) of, any other Entity or enter into any joint venture, partnership or similar arrangement;
- (vii) make, commit to make, or authorize any capital expenditure (other than in accordance with the capital expenditure budget set forth on Section 4.2(b)(vii) of the Company Disclosure Schedule);
- (viii) acquire, lease, sublease, license, sublicense, pledge, encumber, sell or otherwise dispose of, divest or spin-off, abandon, surrender, cancel, waive, relinquish, covenant not to assert, or permit to lapse or expire (other than any Patent expiring at the end of its statutory term and not capable of being extended or abandonment of any application for registration of any Intellectual Property Right that is not material to the businesses of the Acquired Corporations in the ordinary course of business consistent with past practice), transfer, assign or subject to any Encumbrance (other than Permitted Encumbrances) any material right or other material asset or property or Governmental Authorizations, except, in each case (A) pursuant to any Contracts in effect as of the date of this Agreement which have been made available to Parent, (B) for entering into clinical trial agreements solely with respect to Clinical Trials that are ongoing as of the date of this Agreement (and have been made available to Parent), (C) for entering into material transfer agreements (MTAs) in the ordinary course of business consistent with past practice, (D) sale of inventory in the ordinary course of business, (E) purchases of raw materials and other supplies in the ordinary course of business, and in any event in amounts not exceeding \$1,000,000 individually, (F) pursuant to dispositions of obsolete, surplus or worn out assets that are no longer useful in the conduct of the business of the Acquired Corporations, (G) capital expenditures permitted by clause (vii) of this Section 4.2(b), (H) transactions between the Company and another Acquired Corporation or between Acquired Corporations or (I) with respect to pledges (x) constituting Permitted Encumbrances under clause (f) of the definition thereof or (y) constituting Encumbrances created or incurred in connection with any Indebtedness permitted to be established or incurred under Section 4.2(b)(ix);

- (ix) lend money or make capital contributions or advances to or make investments in, any Person, or incur, assume, guarantee, endorse or otherwise become responsible for (or amend or modify the terms of) any Indebtedness for borrowed money (except for (A) advances to directors, employees and consultants for travel and other business related expenses in the ordinary course of business consistent with past practice and in compliance with the Company's policies related thereto, (B) advances of expenses as required under the Company's certificate of incorporation or bylaws or any indemnification agreement with the Company's directors or officers made available to Parent and (C) surety bonds or letters of credit entered into for the benefit of Governmental Bodies in connection with the anticipated licensure or distribution of any Product Candidate) other than between or among the Company and the other Acquired Corporations;
- (x) except in the ordinary course of business consistent with past practice or as otherwise permitted by this Section 4.2(b), (A) enter into any Contract which would have been a Material Contract set forth in Sections 2.9(a)(i), 2.9(a)(iii), 2.9(a)(iv), 2.9(a)(v), 2.9(a)(v),
- (xi) (A) enter into any Contract which would have been a Material Contract under any subsection of $\underline{Section\ 2.9(a)}$ other than the subsections referred to in $\underline{Section\ 4.2(b)(x)}$, if such Contract was in effect as of the execution and delivery of this Agreement, or (B) amend, modify, restate or supplement in any material respect, or waive, assign or release any material rights or claims under, or voluntarily terminate, any Material Contract set forth in any subsection of $\underline{Section\ 2.9(a)}$ other than the subsections referred to in $\underline{Section\ 4.2(b)(x)}$ (or any Contract that would have been a Material Contract under any subsection of $\underline{Section\ 2.9(a)}$ other than the subsections referred to in $\underline{Section\ 4.2(b)(x)}$, if such Contract was in effect as of the execution and delivery of this Agreement);

(xii) enter into any agreement, arrangement or commitment that purports to bind or restrict Guarantor or any of its Affiliates, other than, following the Effective Time, the Surviving Corporation and its Subsidiaries;

- (xiii) (A) make any change to any accounting method or accounting period used for Tax purposes; (B) make, rescind or change any material Tax election; (C) file an amended income or other material amended Tax Return; (D) enter into a closing agreement with any Governmental Body regarding any material Tax liability or assessment; (E) settle, compromise or consent to any material Tax claim or assessment or surrender a right to a material Tax refund; (F) waive or extend the statute of limitations with respect to any income or other material Tax or income or other material Tax Return, other than automatic waivers or automatic extensions obtained in the ordinary course of business consistent with past practice; (G) request any private letter, technical advice or similar Tax ruling from any Governmental Body with respect to Taxes; or (H) enter into any Tax allocation, indemnity or sharing agreement (other than any such agreements entered into in the ordinary course of business consistent with past practice and not relating primarily to Taxes);
- (xiv) settle, release, waive or compromise any Legal Proceeding or other claim (or threatened Legal Proceeding or other claim) against any Acquired Corporation, other than any settlement, release, waiver or compromise that (A) results solely in monetary obligations involving only the payment of monies by the Acquired Corporations of not more than \$250,000 in the aggregate (excluding monetary obligations that are funded by an indemnity obligation to, or an insurance policy of, any Acquired Corporations), (B) does not involve the admission of wrongdoing by, does not impose any material restrictions or changes on the business or operations of, and does not involve any injunctive or equitable or other nonmonetary relief (other than immaterial and non-monetary relief incidental thereto) against, any Acquired Corporation, (C) does not provide for the grant to any third party of a license, cross license or other grant of rights to any material Intellectual Property Rights, and (D) provides for a complete release of the claims in dispute giving rise to such settlement, release, waiver or compromise of any Legal Proceeding or claim brought by the stockholders of the Company against the Company and/or its directors relating to the Transactions or a breach of this Agreement or any other agreements contemplated hereby shall be subject to Section 1.7 or Section 5.6, as applicable;
 - (xv) commence any material Legal Proceeding against any third Person;
- (xvi) fail to preserve and maintain any material Company IP (other than any Patent expiring at the end of its statutory term and not capable of being extended or abandonment of any application for registration of any Intellectual Property Right that is not material to the businesses of the Acquired Corporations in the ordinary course of business consistent with past practice), or disclose to any third Person (other than pursuant to written confidentiality agreements in the ordinary course of business consistent with past practice), or otherwise fail to use commercially reasonable efforts to preserve and maintain the confidentiality of any material Trade Secrets;

(xvii) terminate any Clinical Trials in respect of any Product Candidate that are ongoing as of the execution and delivery of this Agreement, other than pursuant to the terms thereof, or commence (alone or with any third party) any new Clinical Trial in respect of any Product Candidate;

- (xviii) (A) create, implement, operate, participate in, or offer any patient assistance or patient support program that offers, provides or intends to provide free drug product (including any Product Candidate) or any cost-sharing assistance, such as co-pay coupons or co-pay cards in relation to a drug product, to any patient, including any federal healthcare program beneficiaries (each, a "<u>Patient Assistance Program</u>"); or (B) offer, make or provide any grants, charitable contributions, donations, sponsorships or similar support (whether in cash or in kind) that relates to or otherwise supports any third-party Patient Assistance Program (including any co-pay assistance foundation);
- (xix) fail to maintain in full force and effect the existing insurance policies of the Acquired Corporations or to renew or replace such insurance policies with comparable insurance policies;
- (xx) enter into, adopt or amend any collective bargaining agreement or other agreement with any labor organization or works council (except to the extent required by applicable Legal Requirements);
 - (xxi) adopt or implement any stockholder rights plan or similar arrangement;
- (xxii) adopt a plan or agreement of complete or partial liquidation or dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of any of the Acquired Corporations; or
- (xxiii) authorize any of, or agree or commit to take, any of the actions described in the foregoing <u>clauses (i)</u> through (\underline{xxii}) of this <u>Section 4.2(b)</u>.

Notwithstanding the foregoing, nothing contained herein shall give to Parent or Merger Sub, directly or indirectly, rights to control or direct the operations of the Acquired Corporations prior to the Effective Time. Prior to the Effective Time, each of Parent and the Company shall exercise, consistent with the terms and conditions hereof, complete control and supervision of its and its, if applicable, Subsidiaries' respective operations.

4.3 No Solicitation.

(a) For the purposes of this Agreement, "<u>Acceptable Confidentiality Agreement</u>" means any customary confidentiality agreement that (i) contains provisions that are not less favorable to the Company than those contained in the Confidentiality Agreement and (ii) does not prohibit any Acquired Corporation from providing any information to Parent in accordance with this <u>Section 4.3</u> or otherwise prohibit the Acquired Corporations from complying with the provisions of this <u>Section 4.3</u>.

Except as expressly permitted by this Section 4.3, during the Pre-Closing Period the Acquired Corporations shall not, and shall direct and use their reasonable best efforts to cause their Representatives not to, directly or indirectly (i) continue any solicitation, knowing encouragement. discussions or negotiations with any Persons that may be ongoing with respect to an Acquisition Proposal; (ii) (A) solicit, initiate or knowingly facilitate or encourage (including by way of furnishing non-public information) any inquiries regarding, or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, an Acquisition Proposal, (B) engage in, continue or otherwise participate in any discussions or negotiations regarding, or furnish to any other Person any information in connection with, or for the purpose of soliciting or knowingly encouraging or facilitating, an Acquisition Proposal or any proposal or offer that could reasonably be expected to lead to an Acquisition Proposal, or (C) adopt, approve, endorse, recommend, declare advisable, or enter into any letter of intent, memorandum of understanding, agreement in principle, term sheet, or similar agreement whether binding or nonbinding, or any Contract, in each case, with respect to, or that could reasonably be expected to lead to, an Acquisition Proposal or that would reasonably be expected to cause the Company to abandon, terminate, delay or fail to consummate, or that would otherwise materially impede, interfere with or be inconsistent with the consummation of, the Transactions (other than an Acceptable Confidentiality Agreement); (iii) waive or release any Person from, forebear in the enforcement of, or amend any standstill agreement or any standstill provisions of any other Contract, or take any action to exempt any Person (other than Parent, Merger Sub or their Affiliates) from the restrictions on "business combinations" or any similar provision contained in applicable Takeover Laws or the organizational and other governing documents of an Acquired Corporation, unless in the case of this clause (iii), the Board of Directors determines in good faith, after consultation with the Company's outside legal counsel, that the failure to do so would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements and notifies Parent of any such waiver, release, forbearance or amendment; or (iv) resolve or publicly propose to take any of the actions set forth in the foregoing clauses (i) through (iii) of this Section 4.3(b). As promptly as reasonably practicable (and in any event within 24 hours) following the execution and delivery of this Agreement, the Company shall terminate access to any data room or similar facility established by the Company in connection with a potential Acquisition Proposal (including the process that culminated in the execution and delivery of this Agreement), and request (unless such a request was previously made by an Acquired Corporation before the execution and delivery of this Agreement) the prompt return or destruction of all non-public information previously furnished to any Person (other than Guarantor, Parent, Merger Sub and their respective Representatives) that has since October 1, 2019 made or indicated an intention to make an Acquisition Proposal or executed a confidentiality agreement in connection with its consideration of an Acquisition Proposal. The Company shall be fully responsible for any action taken by its or the other Acquired Corporations' Representatives that, had such action been taken by the Company, would constitute a breach of this Section 4.3 and any such action taken by any Representative of an Acquired Corporation shall constitute a breach of this Section 4.3 by the Company.

- (c) Notwithstanding anything in this Agreement to the contrary, if at any time after the execution and delivery of this Agreement and prior to the receipt of the Company Stockholder Approval (the "Cut-off Time") any Acquired Corporation receives a *bona fide* written Acquisition Proposal from any Person or group of Persons, which Acquisition Proposal was made or renewed after the execution and delivery of this Agreement and did not arise out of or result from a material breach of this Section 4.3 or Section 5.1, and the Board of Directors determines in good faith, after consultation with the Company's financial advisors and outside legal counsel, that (i) such Acquisition Proposal constitutes or would reasonably be expected to lead to a Superior Offer and (ii) the failure to take such action described in clauses (x) or (y) of this Section 4.3(c) would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements, then the Company and its Representatives may until the Cut-off Time, (x) furnish, pursuant to an Acceptable Confidentiality Agreement, information (including non-public information) with respect to the Acquired Corporations to the Person or group of Persons who has made such Acquisition Proposal and the Representatives of such Person or group of Persons; provided that the Company shall substantially concurrently provide to Parent and its Representatives, and (y) engage in or otherwise participate in discussions or negotiations with the Person or group of Persons making such Acquisition Proposal and the Representatives of such Person or group of Persons. If the Board of Directors makes any determination described in the foregoing clauses (i) or (ii) of this Section 4.3(c) or initially takes any action set forth in the foregoing clauses (x) or (y) of this Section 4.3(c), the Company shall notify Parent within 24 hours thereof.
- (d) During the Pre-Closing Period, the Company shall (i) promptly (and in any event within 24 hours after receipt thereof by an Acquired Corporation) notify Parent if any request, inquiry, proposal or offer with respect to, or that could reasonably be expected to lead to, an Acquisition Proposal is received by any Acquired Corporation or any Representative thereof and provide to Parent (w) copies of any written request, inquiry, proposal, offer or other materials, including proposed agreements (including any proposed term sheet, letter of intent, acquisition agreement, financing commitments, or similar agreements with respect thereto) received in connection therewith, (x) a summary of any material unwritten terms and conditions thereof, (y) a summary of the nature of any information requested, and (z) the name(s) of the Person or group of Persons making such request, inquiry, proposal or offer, (ii) keep Parent reasonably informed of any material developments, discussions or negotiations regarding any such request, inquiry, proposal or offer (including by furnishing copies of any further requests, inquires or proposals, or amendments thereto) on a prompt basis (and in any event within 24 hours of such material development, discussion or negotiation), and (iii) upon the request of Parent, reasonably inform Parent of the status of such Acquisition Proposal.
- (e) Nothing in this Section 4.3 or elsewhere in this Agreement shall prohibit the Company from (i) taking and disclosing to the stockholders of the Company a position contemplated by Rule 14e-2(a), Rule 14d-9 or Item 1012(a) of Regulation M-A promulgated under the Exchange Act, including any "stop, look and listen" communication pursuant to Rule 14d-9(f) promulgated under the Exchange Act, or (ii) making any disclosure to the stockholders of the Company that the Board of Directors determines, after consultation with its outside counsel, is required by applicable Legal Requirements; provided, however, that nothing in this Section 4.3(e) shall permit the Board of Directors to make a Company Adverse Change Recommendation and, unless the Board of Directors has made a Company Adverse Change Recommendation in accordance with the provisions of Section 5.1(b) that remains in effect and has not been withdrawn, such disclosure shall state that the Company Board Recommendation continues to be in effect.

4.4 Preparation of Proxy Statement; Stockholder Meeting.

(a) As promptly as reasonably practicable (and in any event within fifteen (15) business days) after the date of this Agreement, the Company shall prepare and file with the SEC a proxy statement in preliminary form related to the Company Stockholder Meeting (together with any amendments thereof or supplements thereto, the "Merger Proxy Statement"). The Company shall give Parent and its Representatives a reasonable opportunity to review and comment on the initial preliminary Merger Proxy Statement and all subsequent forms or versions of or amendments or supplements to the Merger Proxy Statement prior to the filing thereof with the SEC or dissemination to the holders of Shares and the Company shall give reasonable and good faith consideration to any timely comments thereon made by Parent or its Representatives. Parent and Merger Sub shall promptly supply to the Company in writing, for inclusion in the Merger Proxy Statement, all information concerning Guarantor, Parent and Merger Sub and their controlled Affiliates required under applicable Legal Requirements to be included in the Merger Proxy Statement as is reasonably requested by the Company. Parent, Merger Sub and the Company shall cooperate in good faith to determine the information regarding each of them that is necessary to include in the Merger Proxy Statement in order to satisfy applicable Legal Requirements. The Company covenants and agrees that the Merger Proxy Statement (i) at the time the Merger Proxy Statement (including the preliminary Merger Proxy Statement and all subsequent forms or versions of or amendments or supplements to the Merger Proxy Statement) is filed with the SEC, (ii) at the time the Merger Proxy Statement is first published, sent or disseminated to the holders of Shares, and (iii) at the time of the Company Stockholder Meeting, will (A) comply in all material respects with the Exchange Act and other applicable Legal Requirements and (B) not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading; provided, however, that no covenant is made by the Company with respect to information supplied by or on behalf of Guarantor, Parent or Merger Sub for inclusion or incorporation by reference in the Merger Proxy Statement. Each of Parent, Merger Sub and the Company agrees to respond promptly to any comments of the SEC or its staff and to promptly correct any information provided by it for use in the Merger Proxy Statement if and to the extent that such information shall have become false or misleading in any material respect, and the Company further agrees to take all steps necessary to cause the Merger Proxy Statement as so corrected to be promptly filed with the SEC and to be disseminated to holders of Shares, in each case as and to the extent required by applicable Legal Requirements. The Company shall (x) promptly notify Parent of the receipt of, and promptly provide Parent copies of, all comments (including oral comments) from, and all correspondence with, the SEC or its staff with respect to the Merger Proxy Statement and shall promptly notify Parent of any request by the SEC or its staff for any amendment or supplement thereto or for additional information, (y) provide Parent and its counsel with a reasonable opportunity to review and comment on any proposed correspondence between the Company or any of its Representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to the Merger Proxy Statement and shall give reasonable and good faith consideration to any timely comments thereon made by Parent or its Representatives and (z) promptly provide Parent with final copies of any correspondence sent by the Company or any of its Representatives to the SEC or its staff with respect to the Merger Proxy Statement, and of any amendments or supplements to the Merger Proxy Statement. The Merger Proxy Statement shall include the Company Board Recommendation, unless the Board of Directors has made a Company Adverse Change Recommendation in compliance with Section 5.1. The Merger Proxy Statement shall include (i) the fairness opinion of the Company's financial advisors referenced in Section 2.24, (ii) the notice of the Company Stockholder Meeting and (iii) the notice and other information required by Section 262(d) of the DGCL.

The Company shall establish a record date for duly call, convene and hold a meeting of its stockholders for the purpose of voting upon the adoption of this Agreement (together with any adjournments or postponements thereof, the "Company Stockholder Meeting") as reasonably promptly as practicable after the SEC Clearance Date (and shall initially schedule the Company Stockholder Meeting to be held within twenty-five (25) business days from the date of the mailing of the Merger Proxy Statement, or if the Company's nationally recognized proxy solicitor advises the Company that twenty-five (25) business days from the date of the mailing of the Merger Proxy Statement is insufficient time to obtain the Company Stockholder Approval, such later date to which Parent consents in writing (such consent not to be unreasonably withheld, conditioned or delayed)), and any adjournments or postponements of such meeting shall require the prior written consent of Parent (not to be unreasonably withheld, conditioned or delayed) other than to the extent necessary to allow reasonable additional time for the filing and/or mailing, and review by the Company's stockholders prior to the date of the Company Stockholder Meeting, of any supplemental or amended disclosure that the Board of Directors determines in good faith (after consultation with outside legal counsel) is required by applicable Legal Requirements or the rules and regulations of Nasdaq. Notwithstanding the foregoing, the Company may, and Parent may require the Company to, adjourn or postpone the Company Stockholder Meeting two (2) times (for a period of not more than ten (10) business days each but not past three (3) business days prior to the End Date), unless prior to such adjournment or postponement, the Company shall have received an aggregate number of proxies voting for the adoption of this Agreement, which have not been withdrawn, such that the condition in Section 6.1(a) would be satisfied at such meeting if it were to be held without such postponement or adjournment. In connection with the foregoing, the Company shall (i) file the definitive Merger Proxy Statement with the SEC and cause the definitive Merger Proxy Statement to be mailed to the Company's stockholders as of the record date established for the Company Stockholder Meeting as reasonably promptly as practicable (and in any event within four (4) business days) after the date on which the SEC confirms that it has no further comments on the Merger Proxy Statement (the "SEC Clearance Date"); provided, that if the SEC has failed to affirmatively notify the Company within ten (10) days after the initial filing of the Merger Proxy Statement with the SEC that it will or will not be reviewing the Merger Proxy Statement, then such date shall be the "SEC Clearance Date". Once the Company has established a record date for the Company Stockholder Meeting, the Company shall not change such record date or establish a different record date for the Company Stockholder Meeting without the prior written consent of Parent (such consent not to be unreasonably withheld, conditioned or delayed), unless required to do so by applicable Legal Requirements. Unless the Board of Directors shall have made a Company Adverse Change Recommendation in compliance with Section 5.1, the Company shall use reasonable best efforts to obtain the Company Stockholder Approval, including to solicit proxies in favor of the adoption of this Agreement. Unless this Agreement is validly terminated in accordance with Section 7.1, the Company shall submit this Agreement to its stockholders at the Company Stockholder Meeting even if the Board of Directors shall have made a Company Adverse Change Recommendation or proposed or announced any intention to do so. The Company shall, upon the reasonable request of Parent, advise Parent at least on a daily basis on each of the last seven (7) business days prior to the date of the Company Stockholder Meeting as to the aggregate tally of proxies received by the Company with respect to the Company Stockholder Approval. Without the prior written consent of Parent, the adoption of this Agreement and a nonbinding advisory vote on compensation matters shall be the only matters (other than related procedural matters) that the Company shall propose to be acted on by the stockholders of the Company at the Company Stockholder Meeting.

SECTION 5

ADDITIONAL COVENANTS OF THE PARTIES

5.1 Company Board Recommendation.

- (a) During the Pre-Closing Period, subject to the Board of Directors making a Company Adverse Change Recommendation in accordance with the provisions of Section 5.1(b), neither the Board of Directors nor any committee thereof shall (i) (A) withdraw (or modify or qualify in a manner adverse to Parent or Merger Sub), or publicly propose to withdraw (or modify or qualify in a manner adverse to Parent or Merger Sub), the Company Board Recommendation, or (B) approve, recommend or declare advisable, or publicly propose to approve, recommend or declare advisable, any Acquisition Proposal (any action described in the foregoing clauses (i)(A) and (B) of this Section 5.1(a) being referred to as a "Company Adverse Change Recommendation") or (ii) adopt, approve, recommend or declare advisable, or propose to adopt, approve, recommend or declare advisable, or allow the Company to execute or enter into any Contract with respect to any Acquisition Proposal, or Contract that would require, or would reasonably be expected to cause, the Company to abandon, terminate, delay or fail to consummate, or that would otherwise materially impede, interfere with or be inconsistent with, the Transactions (other than an Acceptable Confidentiality Agreement).
 - (b) Notwithstanding anything to the contrary contained in this Agreement, at any time prior to the Cut-off Time:
- (i) if, after the execution and delivery of this Agreement, any Acquired Corporation receives a bona fide written Acquisition Proposal that did not arise out of or result from a material breach of Section 4.3 or Section 5.1(a) from any Person, and such Acquisition Proposal has not been withdrawn, and after consultation with the Company's financial advisors and outside legal counsel, the Board of Directors shall have determined, in good faith, that such Acquisition Proposal is a Superior Offer, (x) the Board of Directors may make a Company Adverse Change Recommendation, or (y) the Company may terminate this Agreement pursuant to Section 7.1(e) to enter into a Specified Agreement with respect to such Superior Offer, but in each case under (x) or (y), only if: (A) the Board of Directors determines in good faith, after consultation with the Company's financial advisors and outside legal counsel, that the failure to do so would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements; (B) the Company has given Parent prior written notice of its intention to consider making a Company Adverse Change Recommendation or terminating this Agreement pursuant to Section 7.1(e) (a "Determination Notice") (which notice shall not constitute a Company Adverse Change Recommendation or termination), and, if desired by Parent, during the Negotiation Period, the Company has negotiated in good faith with respect to any revisions to the terms of this Agreement or another proposal, to the extent proposed by Parent, so that such Acquisition Proposal would cease to constitute a Superior Offer; and (C)(1) the Company has, prior to the commencement of the Negotiation Period, provided to Parent information required to have been provided pursuant to Section 4.3(c), and Section 4.3(d), (2) the Company has given Parent the full Negotiation Period to propose revisions to the terms of this Agreement or make another proposal so that such Acquisition Proposal would cease to constitute a Superior Offer, and (3) after giving effect to the proposals made by Parent during the Negotiation Period, if any, after consultation with the Company's financial advisors and outside legal counsel, the Board of Directors has determined, in good faith, that such Acquisition Proposal continues to be a Superior Offer and that the failure to make the Company Adverse Change Recommendation or terminate this Agreement pursuant to Section 7.1(e) would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements. Issuance of any "stop, look and listen" communication by or on behalf of the Company pursuant to Rule 14d-9(f) promulgated under the Exchange Act shall not be considered a Company Adverse Change Recommendation and shall not require the giving of a Determination Notice or compliance with the procedures set forth in this Section 5.1. The provisions of this Section 5.1(b)(i) shall also apply to any material amendment (which shall be deemed to include any change to the financial terms thereof) to any Acquisition Proposal and shall require a new Determination Notice; and

(ii) other than in connection with an Acquisition Proposal, the Board of Directors may make a Company Adverse Change Recommendation in response to an Intervening Event, but only if: (A) the Board of Directors determines in good faith, after consultation with the Company's outside legal counsel, that the failure to do so would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements; (B) the Company has given Parent a Determination Notice describing in reasonable detail the facts and circumstances resulting in such Intervening Event and that render a Company Adverse Change Recommendation necessary and, if desired by Parent, during the Negotiation Period, the Company has negotiated in good faith with respect to any revisions to the terms of this Agreement or another proposal, to the extent proposed by Parent, so that a Company Adverse Change Recommendation in response to such Intervening Event would no longer be necessary; and (C)(1) the Company has given Parent the full Negotiation Period to propose revisions to the terms of this Agreement or make another proposal so that a Company Adverse Change Recommendation would no longer be necessary, and (2) after giving effect to the proposals made by Parent during the Negotiation Period, if any, after consultation with the Company's outside legal counsel, the Board of Directors has determined, in good faith, that the failure to make the Company Adverse Change Recommendation would be inconsistent with the fiduciary duties of the Board of Directors to the Company's stockholders under applicable Legal Requirements.

5.2 Filings, Consents and Approvals.

- Subject to the terms and conditions of this Agreement, the Parties agree to use (and shall cause their respective Affiliates to use) their reasonable best efforts to take promptly any and all steps necessary to avoid or eliminate each and every impediment under Antitrust Laws that may be asserted by any Governmental Body so as to enable the Closing to occur as promptly as practicable, but in no case later than the End Date, including providing as promptly as reasonably practicable all information required by any Governmental Body pursuant to its evaluation of the Transactions under the HSR Act or other applicable Antitrust Laws. Subject to the terms of this Section 5.2, the Parties shall (and shall cause their respective Affiliates to) use their reasonable best efforts to obtain from any Governmental Body all consents, approvals, authorizations or orders required to be obtained under Antitrust Laws or to avoid the entry or enactment of any order, directive, judgment, decree or ruling relating to any Antitrust Law that would delay, restrain, prevent, enjoin or otherwise prohibit consummation of the Transactions. Notwithstanding the foregoing and any other provision of this Agreement to the contrary, in no event shall Parent, Merger Sub or any of their Affiliates (including Guarantor) be required to propose, negotiate, undertake, commit to or consent to any divestiture, sale, disposition, licensing, hold separate order or other structural or conduct relief, or other operational undertakings, in order to obtain clearance from any Governmental Body under Antitrust Laws. Nothing in this Section 5.2 shall require Parent, Merger Sub or the Company to take or agree to take any action unless the effectiveness of such action is conditioned upon Closing, and the Company shall not take or propose to undertake any divestiture, sale, disposition, hold separate order or other structural or conduct relief, or other operational undertaking without Parent's prior consent (which may be given or withheld in Parent's sole discretion). Notwithstanding the foregoing and any other provision of this Agreement to the contrary, if any Legal Proceeding under Antitrust Laws, whether judicial or administrative, is instituted by a Governmental Body challenging or seeking to restrain or prohibit the consummation of the Transactions, Parent and the Company shall (and shall cause their respective Affiliates to) contest and defend such Legal Proceeding through litigation on the merits in order to seek to avoid entry of, or to have vacated or terminated, any temporary restraining order, preliminary or permanent injunction or other order preventing or prohibiting the consummation of the Transactions prior to the End Date; provided, that in no event shall any Party be required to waive any right or condition set forth in this Agreement.
- (b) Subject to the terms and conditions of this Agreement, (i) each of the Parties shall (and shall cause their respective Affiliates, if applicable, to) promptly, but in no event later than fifteen (15) business days after the date hereof, make an appropriate filing of all notification and report forms as required by the HSR Act with respect to the Transactions, (ii) if Parent reasonably determines that a filing or notification is required to be made with, or any consent, approval, permit or authorization is required to be obtained from, a Specified Antitrust Authority pursuant to prong (ii) of the definition thereof under Antitrust Laws in connection with the Transactions under the circumstances set forth in Section 5.2(c) of the Company Disclosure Schedule, then Parent may make such filing or notification and seek such consent, approval, permit or authorization, and Parent and the Company shall, and shall cause their respective Affiliates to, cooperate with each other in connection therewith, and (iii) the Parties shall (and shall cause their respective Affiliates, if applicable, to) cooperate with each other in promptly determining (such determination to be evidenced by mutual written consent of the Parties, which consent shall not be unreasonably withheld, conditioned or delayed) whether any other filings or notifications are required to be made with, or any other consents, approvals, permits or authorizations are required to be obtained from, any other Governmental Bodies (including any Specified Antitrust Authorities pursuant to prong (iii) of the definition thereof) under Antitrust Laws in connection with the Transactions, and if so, to promptly prepare and make any such filings or notifications and to seek any such other consents, approvals, permits or authorizations (the foregoing clauses (i) through (iii) collectively, "Regulatory Filings").

- (c) For purposes of this Agreement, "Specified Antitrust Authority" shall mean (i) any U.S. federal or state Governmental Body, (ii) any non-U.S. or supranational Governmental Body listed on Section 5.2(c) of the Company Disclosure Schedule to the extent such Governmental Body's waiting period, clearance, consent or approval is required under Antitrust Laws in connection with the Transactions and (iii) any other Governmental Body in respect of which a Regulatory Filing is made pursuant to Section 5.2(b)(iii).
- (d) Without limiting the generality of anything contained in this Section 5.2, during the Pre-Closing Period, each Party shall give the other Parties prompt notice of (i) the making or commencement of any request, inquiry, investigation, action or Legal Proceeding brought or threatened in writing to be brought by a Governmental Body or brought or threatened in writing to be brought by a third party before any Governmental Body, in each case, with respect to the Transactions (other than any request, inquiry, investigation, action or Legal Proceeding brought or threatened in writing to be brought by the stockholders of the Company against the Company and/or its directors relating to the Transactions or a breach of this Agreement or any other agreements contemplated hereby, which are addressed in Sections 1.7 or 5.6 as applicable), and (ii) its receipt of any notice or other communication from any third party alleging that the consent of a third party is or may be required in connection with the Transactions (the foregoing clauses (i) and (ii), the "Identified Communications"). Additionally, each Party shall (A) keep the other Parties reasonably informed as to the status of any Regulatory Filings and Identified Communications or response thereto, (B) promptly inform the other Parties of, and (wherever practicable) give the other Parties reasonable advance notice of, and the opportunity to participate in, any communication to or from the FTC, DOJ or any other Governmental Body or third party in connection with any such Regulatory Filings and Identified Communication, (C) promptly furnish to the other Parties, copies of documents provided to or received from any Governmental Body in connection with any Regulatory Filings or Identified Communication (other than "4(c) documents" as that term is used in the rules and regulations under the HSR Act, that contain valuation information (which can be redacted)), (D) to the extent reasonably practicable, consult and cooperate with the other Parties and consider in good faith the views of the other Parties in connection with any analysis, appearance, presentation, memorandum, brief, argument, opinion or proposal made or submitted in connection with any Regulatory Filings or Identified Communication, and (E) except as may be prohibited by any Legal Requirement, in connection with any such Regulatory Filings and Identified Communications, give the other Parties reasonable advance notice of, and permit authorized Representatives of the other Party to be present at each meeting or conference relating to such Regulatory Filing or Identified Communication and to have access to and be consulted in connection with any argument, opinion or proposal made or submitted to any third party in connection with such Regulatory Filing or Identified Communication. Notwithstanding anything to the contrary in this Section 5.2, the Parties may (y) as they deem reasonably advisable and necessary based on the advice of outside counsel to prevent a violation of Antitrust Law, designate competitively sensitive materials and information provided to the other under this Section 5.2 as "outside counsel only" and such materials and the information contained therein shall be given only to outside counsel of the recipient and will not be disclosed by such outside counsel to employees, officers, or directors of the recipient without the advance written consent of the Party providing such materials, and (z) redact information provided to one another pursuant to this Section 5.2 to the extent providing such information would (1) contravene any applicable Legal Requirement (so long as the disclosing Party has reasonably cooperated with recipient and used its reasonable best efforts to permit disclosure to the extent permitted by Legal Requirements), (2) jeopardize any attorney-client or other legal privilege (so long as the disclosing Party has reasonably cooperated with recipient and used its reasonable best efforts to permit disclosure on a basis that does not waive such privilege with respect thereto), or (3) contravene any Contract to which the disclosing Party is a party or by which the disclosing Party is bound as of the date of this Agreement (so long as the disclosing Party has reasonably cooperated with the recipient and used its reasonable best efforts to permit disclosure on a basis that would not contravene such Contract).

- (e) Subject to the terms of this Section 5.2, and subject to Parent consulting with and considering in good faith the views and comments of the Company, Parent shall have the right to (i) direct, devise and implement the strategy for obtaining any necessary approval of, and for responding to any request from, or inquiry or investigation by (including directing the nature and substance of all such responses), and lead all meetings and communications (including any negotiations) with, any Governmental Body in connection with Regulatory Filings and (ii) control the defense and settlement of any investigation or Legal Proceeding relating to the Transactions that is brought by or before any Governmental Body in connection with the Regulatory Filings. Notwithstanding the foregoing or anything to the contrary in this Agreement, no Party shall (or shall permit any of its Affiliates to) commit to or agree with any Governmental Body to stay, toll, or extend, any applicable waiting period or enter into any similar timing agreement, without the prior written consent of the other Parties (not to be unreasonably withheld, conditioned or delayed); provided that Parent and Merger Sub shall be permitted without the Company's consent (it being understood that Parent and Merger Sub may exercise this right only one time) to withdraw their filing under the HSR Act in connection with the Transactions and promptly refile the notification and report forms as required by the HSR Act with respect to the Transactions.
- (f) The Company shall give prompt notice to Parent of any event, development, occurrence, circumstance, change or effect that has had or would reasonably be expected to have a Material Adverse Effect, or would reasonably be expected to make the satisfaction of any of the conditions set forth in Section 6.1 or Section 6.2 impossible or unlikely, and Parent shall give prompt notice to the Company of any event, development, occurrence, circumstance, change or effect that has had or would reasonably be expected to have a Parent Material Adverse Effect, or would reasonably be expected to make the satisfaction of any of the conditions set forth in Section 6.1 or Section 6.3 impossible or unlikely.

- under applicable Legal Requirements, (i) offer Parent the opportunity to consult with the Acquired Corporations prior to any proposed material meeting or other material communication with the FDA, EMA, the Centers for Medicare & Medicaid Services ("CMS") or any other Specified Governmental Body relating to any Product Candidate or material Governmental Authorization under a Health Care Law, (ii) promptly inform Parent of, and provide Parent with a reasonable opportunity to review, in advance, (A) any material filing proposed to be made by or on behalf of an Acquired Corporation, and (B) any material correspondence or other material communication, in each case (A) or (B) proposed to be submitted or otherwise transmitted to the FDA, EMA, CMS or any other Specified Governmental Body by or on behalf of an Acquired Corporation relating to any Product Candidate or material Governmental Authorization under a Health Care Law, (iii) keep Parent reasonably informed of any material communication (written or oral) with or from the FDA, EMA, CMS or any other Specified Governmental Body or relating to any Product Candidate or Governmental Authorization and (iv) promptly inform Parent and provide Parent with a reasonable opportunity to comment, in each case, prior to making any material change to any study protocol, making any material change to a manufacturing plan or process, making any material change to a development timeline or initiating, or making any material change to, commercialization and reimbursement activities or materials (including promotional and marketing activities and materials) relating to any Product Candidate. The Company shall promptly notify Parent of any material data relating to any Product Candidate, including information related to any serious or unexpected adverse events (as such concepts are contemplated under applicable Health Care Laws) with respect to any Product Candidate, in each case which it discovers after the execution and delivery of this Agreement.
- (h) The delivery of any notice pursuant to this <u>Section 5.2</u> shall not cure any breach of any representation or warranty requiring disclosure of such matter prior to the date of this Agreement or otherwise limit or affect the remedies available hereunder to any Party.

5.3 Employee Benefits.

(a) For a period of one (1) year following the Effective Time, Parent shall provide, or cause to be provided, to each employee of an Acquired Corporation who is employed by an Acquired Corporation as of immediately prior to the Effective Time and who continues to be actively employed by the Surviving Corporation (or any Affiliate thereof, including, for the avoidance of doubt, Parent and Parent's Affiliates) during such one (1)-year period (each, a "Continuing Employee") with (i) a base salary or wage rate that is no less than that provided to such Continuing Employee by any Acquired Corporation immediately prior to the Effective Time, (ii) other target cash incentive compensation opportunities that are substantially comparable, in the aggregate, to either (x) those provided to such Continuing Employee by any Acquired Corporation immediately prior to the Effective Time, or (y) those provided by Parent or its Affiliates to similarly situated employees, and (iii) other compensation and employee benefits (excluding equity-based compensation, nonqualified deferred compensation and defined benefit plans) that are substantially comparable, in the aggregate, to either (x) those provided to such Continuing Employee by any Acquired Corporation immediately prior to the Effective Time or (y) those provided by Parent or its Affiliates to similarly situated employees.

- (b) Parent shall cause all Continuing Employees to be eligible to continue to participate in the Surviving Corporation's health and welfare benefit plans (to the same extent such Continuing Employees were eligible to participate under the Company's health and welfare benefit plans immediately prior to the Effective Time). To the extent that service is relevant for eligibility or vesting under any benefit plan of Parent and/or the Surviving Corporation (other than equity-based compensation, nonqualified deferred compensation and defined benefit plans), then Parent shall ensure that such benefit plan shall, for purposes of eligibility and vesting, credit Continuing Employees for service prior to the Effective Time with the Company and its Affiliates or their respective predecessors to the same extent that such service was recognized prior to the Effective Time under the corresponding benefit plan of the Company, but no credit for any service will be required that would result in a duplication of benefits. In addition, Parent and/or the Surviving Corporation shall credit each Continuing Employee with paid time off equal to the accrued paid time off such Continuing Employee had accrued with the Company that was unused as of the Effective Time.
- (c) Following the Effective Time, Parent or an Affiliate of Parent shall use commercially reasonable efforts to: (i) waive any preexisting condition limitations otherwise applicable to Continuing Employees and their eligible dependents under any plan of Parent or an Affiliate that provides health benefits in which Continuing Employees are eligible to participate following the Effective Time, other than any limitations that were in effect with respect to such employees immediately prior to the Effective Time under the corresponding Employee Plan, (ii) honor any deductible, copayment and out-of-pocket maximums incurred by the Continuing Employees and their eligible dependents under the health plans in which they participated immediately prior to transitioning into a plan of Parent or an Affiliate during the portion of the plan year prior to such transition in satisfying any deductibles, co-payments or out-of-pocket maximums under health plans of Parent or an Affiliate, in each case to the extent recognized for such purpose under an analogous Employee Plan prior to the Effective Time; and (iii) waive any waiting period limitation or evidence of insurability requirement that would otherwise be applicable to a Continuing Employee and his or her eligible dependents on or after the Effective Time, in each case to the extent such Continuing Employee or eligible dependent had satisfied any similar limitation or requirement under an analogous Employee Plan prior to the Effective Time.
- (d) The provisions of this <u>Section 5.3</u> are solely for the benefit of the Parties, and no provision of this <u>Section 5.3</u> is intended to, or shall, (i) constitute the establishment or adoption of or an amendment to any employee benefit plan for purposes of ERISA or otherwise, or (ii) prohibit or limit the ability of Parent or any of its Affiliates (including, following the Effective Time, the Company) to amend, modify or terminate any employee benefit or compensation plan, program or arrangement. No current or former employee or any other individual associated therewith shall be regarded for any purpose as a third party beneficiary of the Agreement or have the right to enforce the provisions hereof. Nothing in this <u>Section 5.3</u> or elsewhere in this Agreement shall be construed to create a right in any Person to employment with Parent, the Surviving Corporation or any other Affiliate of the Surviving Corporation or to any compensation or benefits and the employment of each Continuing Employee shall be "at will" employment.

- (e) At the written request of Parent provided no later than five (5) days prior to the Effective Time, the Company shall, at least one (1) business day prior to the Effective Time, cease contributions to, and adopt written resolutions (or take other necessary and appropriate action) to terminate, the Company 401(k) Savings Plan (the "Company 401(k) Plan"), effective as of the day prior to the Closing Date and to one hundred percent (100%) vest all participants under the Company 401(k) Plan, such termination and vesting to be effective no later than the business day preceding the Effective Time; provided, however, that such Company 401(k) Plan cessation of contributions, vesting and termination may be made contingent upon the Closing. The resolutions proposed to be adopted in connection with the foregoing termination shall be provided to Parent prior to adoption and shall be subject to review and approval by Parent (which approval shall not be unreasonably withheld, conditioned or delayed). If the Company 401(k) Plan is terminated in accordance with the provisions of this Section 5.3(e), effective as of, or as soon as reasonably practicable following, the Closing Date, Parent shall designate a tax-qualified defined contribution retirement plan of Parent (or an Affiliate of Parent) with a qualified cash or deferred arrangement under Section 401(k) of the Code (the "Parent 401(k) Plan") that will cover Continuing Employees on and after the Closing Date. Parent shall cause the Parent 401(k) Plan to accept direct rollovers to such Parent 401(k) Plan of the account balances of each Continuing Employee (including the in-kind rollover of promissory notes evidencing all outstanding loans) from the Company 401(k) Plan, if such direct rollover is elected by such Continuing Employee in accordance with the terms of the Company 401(k) Plan (as it may be amended to provide for such rollover) and the Code. Parent shall take any and all commercially reasonable actions needed to permit each Continuing Employee with an outstanding loan balance under the Company 401(k) Plan as of the date such plan is terminated to continue to make scheduled loan payments to the Company 401(k) Plan after the Closing Date, pending the distribution and in-kind rollover of the promissory notes evidencing such loans from the Company 401(k) Plan to the Parent 401(k) Plan, as provided in the preceding sentence, such as to prevent, to the extent reasonably possible, a deemed distribution or loan offset with respect to such outstanding loans.
- (f) During the Pre-Closing Period, any written or broad-based notices or communication materials (including website postings) from the Company to its employees with respect to the Transactions or employment, compensation or benefits matters addressed in this Agreement or related, directly or indirectly, to the Transactions or employment with Parent, the Surviving Corporation or any of their respective Subsidiaries thereafter, shall be subject to the prior review, comment and approval of Parent, which such review, comment or approval shall not be unreasonably withheld, conditioned or delayed.
- 5.4 ESPP. As soon as reasonably practicable after the date of this Agreement, the Company shall take all actions necessary pursuant to the terms of the Company ESPP or otherwise to provide that (a) the "Offering" (as defined in the Company ESPP) in effect as of the date of this Agreement shall be the final Offering (such period, the "Final Offering") and no new Offering will be commenced following the date hereof under the Company ESPP, and (b) each individual participating in the Final Offering on the date of this Agreement shall not be permitted to (i) increase his or her payroll contribution rate pursuant to the Company ESPP from the rate in effect when the Final Offering commenced or (ii) make separate non-payroll contributions to the Company ESPP on or following the date of this Agreement, except as may be required by applicable Legal Requirements. Prior to the Effective Time, the Company shall take all action that may be necessary to, effective upon the Effective Time, (A) cause the Final Offering, to the extent that it would otherwise be ongoing at the Effective Time, to be terminated no later than five (5) business days prior to the date on which the Effective Time occurs; (B) make any *pro rata* adjustments that may be necessary to reflect the Final Offering, but otherwise treat the Final Offering as a fully effective and completed Offering for all purposes pursuant to the Company ESPP; and (C) cause the exercise (as of no later than five (5) business days prior to the date on which the Effective Time occurs) of each outstanding purchase right pursuant to the Company ESPP. On the final exercise date under the Company ESPP, the Company shall apply the funds credited as of such date pursuant to the Company ESPP within each participant's payroll withholding account to the purchase of Shares in accordance with the terms of the Company ESPP, and each Share shall be an outstanding Share as of the Effective Time. The Company shall adopt such resolutions as are necessary to terminate the Company ESPP effective

5.5 Indemnification of Officers and Directors.

For a period of six (6) years from the Effective Time, Parent agrees that all rights to indemnification, advancement of expenses and exculpation from liabilities for acts or omissions occurring at or prior to the Effective Time (whether asserted or claimed prior to, at or after the Effective Time) existing in favor of the current or former directors or officers of any Acquired Corporation pursuant to the organizational documents of the Company and any indemnification or other similar agreements of any Acquired Corporation set forth in Section 5.5(a) of the Company Disclosure Schedule, in each case as in effect on the date of this Agreement, shall continue in full force and effect in accordance with their terms, and Parent shall cause the Acquired Corporations to perform their obligations thereunder. Without limiting the foregoing, during the period commencing at the Effective Time and ending on the sixth anniversary of the Effective Time, Parent shall, and shall cause the Acquired Corporations to, indemnify and hold harmless each individual who is as of the date of this Agreement, or who becomes prior to the Effective Time, a director or officer of any Acquired Corporation or who is as of the date of this Agreement, or who thereafter commences prior to the Effective Time, serving at the request of any Acquired Corporation as a director or officer of another Person (the "Indemnified Persons"), against all claims, losses, liabilities, damages, judgments, inquiries, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements, incurred in connection with any claim, action, suit or proceeding, whether civil, criminal, administrative or investigative (including with respect to matters existing or occurring at or prior to the Effective Time, including this Agreement and the Transactions), arising out of or pertaining to the fact that the Indemnified Person is or was a director or officer of any Acquired Corporation or is or was serving at the request of any Acquired Corporation as a director or officer of another Person, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent permitted under applicable Legal Requirements. In the event of any such claim, action, suit or proceeding, Parent shall, and shall cause the Acquired Corporations to, (x) pay, in advance of the final disposition of such claim, action, suit or proceeding, any expenses incurred in defense thereof by the Indemnified Person upon receipt of an undertaking by such Indemnified Person to repay such amount if it shall ultimately be determined that such Indemnified Person is not entitled to be indemnified pursuant to the preceding sentence and (y) reasonably cooperate in the defense of any such matter.

- For a period of six (6) years from and after the Effective Time, Parent shall cause the Surviving Corporation to either maintain in effect its current policies of directors' and officers' and fiduciary liability insurance maintained by or for the benefit of the Acquired Corporations or provide a substitute policy for the Acquired Corporations and their current and former directors and officers who are currently covered by the directors' and officers' and fiduciary liability insurance coverage currently maintained by or for the benefit of the Acquired Corporations as of the date of this Agreement, in either case, of not less than the existing coverage as of the date of this Agreement and having other terms that are in aggregate not less favorable to the insured persons than the directors' and officers' and fiduciary liability insurance coverage currently maintained by or for the benefit of the Acquired Corporations with respect to claims arising from facts or events that occurred at or before the Effective Time (with insurance carriers having at least an "A" rating by A.M. Best with respect to directors' and officers' and fiduciary liability insurance), except that in no event shall Parent or the Surviving Corporation be required to pay with respect to such insurance policies more than 300% of the aggregate annual premium most recently paid by the Acquired Corporations prior to the date of this Agreement (the "Maximum Amount"), and if the Surviving Corporation is unable to obtain the insurance required by this Section 5.5(b) it shall obtain as much comparable insurance as possible for the years within such six (6)-year period for a premium not exceeding the Maximum Amount. In lieu of such insurance, prior to the Closing Date the Company may, at its option, purchase a "tail" directors' and officers' and fiduciary liability insurance policy for the Acquired Corporations and their current and former directors and officers who are currently covered by the directors' and officers' and fiduciary liability insurance coverage currently maintained by or for the benefit of the Acquired Corporations, such tail to provide coverage in an amount not less than the existing coverage and to have other terms not less favorable to the insured persons than the directors' and officers' and fiduciary liability insurance coverage currently maintained by or for the benefit of the Acquired Corporations with respect to claims arising from facts or events that occurred at or before the Effective Time; provided that in no event shall the cost of any such tail policy exceed the Maximum Amount, and if such cost would exceed the Maximum Amount, then the Company may obtain a tail policy with the greatest coverage available for a cost not exceeding the Maximum Amount. Parent and the Surviving Corporation shall maintain such policies in full force and effect for a period of six (6) years after the Effective Time, and continue to honor the obligations thereunder.
- (c) In the event that any Acquired Corporation or any of its successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or Entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, Parent shall cause the applicable Acquired Corporation to make proper provision so that the successors and assigns of such Acquired Corporation assume the obligations set forth in this Section 5.5.

- (d) The provisions of this Section 5.5 (i) shall survive the Effective Time and (ii) are intended to be for the benefit of, and will be enforceable by, each indemnified or insured party (including the Indemnified Persons), his or her heirs, successors, assigns and representatives, and (iii) are in addition to, and not in substitution for, any other rights to indemnification, advancement of expenses, exculpation or contribution that any such Person may have by contract or otherwise. Unless required by applicable Legal Requirement, this Section 5.5 may not be amended, altered or repealed after the Effective Time in such a manner as to adversely affect the rights of any Indemnified Person or any of their successors, assigns or heirs, without the prior written consent of the affected Indemnified Person.
- 5.6 Stockholder Litigation. In the event that any claims or Legal Proceedings related to this Agreement, the Merger or the other Transactions are brought by any stockholder or other holder of Company securities (whether directly or on behalf of the Company or otherwise) against any Acquired Corporation or any of the Acquired Corporations' directors or officers, the Company shall promptly notify Parent in writing and shall keep Parent reasonably and promptly informed with respect to the status thereof. The Company shall give Parent (a) the right to participate in the defense of, and review and comment on all material filings or responses to be made by the Company or other Acquired Corporation in connection with, any such claims or Legal Proceedings (and shall give due consideration to Parent's comments and other advice with respect to such litigation), and (b) the right to consult on any settlement with respect to such claims or Legal Proceedings, and no such settlement shall be agreed to without Parent's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed).
- 5.7 Additional Agreements. Subject to the terms and conditions of this Agreement, including Section 5.2(a), Parent and the Company shall (and shall cause their respective Affiliates to) use reasonable best efforts to take, or cause to be taken, all actions necessary to consummate the Merger and make effective the other Transactions. Without limiting the generality of the foregoing, subject to the terms and conditions of this Agreement, each Party shall (and shall cause its respective Affiliates to) use reasonable best efforts to (a) make all filings (if any) and give all notices (if any) required to be made and given by such Party pursuant to any Material Contract in connection with the Merger and the other Transactions, (b) seek each Consent (if any) required to be obtained pursuant to any Material Contract by such Party in connection with the Transactions to the extent requested in writing by Parent; provided that (i) without the prior written consent of Parent, the Company shall not, and shall cause each of the other Acquired Corporations not to, pay or commit to pay to such Person that is not a Governmental Body, whose approval or consent is being solicited any cash or other consideration, make any commitment or incur any liability or other obligation due to such Person, and (i) neither Parent nor any of its Affiliates shall be required to pay or commit to pay to any Person that is not a Governmental Body whose approval or consent is being solicited any cash or other consideration, make any commitment or incur any liability or other obligation to any such Person, and (c) seek to lift any restraint, injunction or other legal bar to the Merger brought by any third Person, other than a Governmental Body pursuant to an Antitrust Law (which matter is addressed in the final sentence of Section 5.2(a)), against such Party.

- **Disclosure.** The initial press release relating to this Agreement shall be a joint press release in a form previously agreed to by Parent and the Company and issued by the Company and Guarantor promptly after the execution and delivery of this Agreement; provided that Guarantor may issue. substantially concurrent with the issuance of such joint press release, a Japanese press release in a form agreed by the Parties prior to issuance thereof. Following the issuance of the press releases described in the immediately preceding sentence, Parent and the Company shall consult with each other before issuing, or causing the publication of, any further press release(s) or otherwise making any public statement or any announcement to employees (to the extent not previously issued or made in accordance with this Agreement) with respect to the Merger, this Agreement or any of the other Transactions and shall not issue any such press release or make any such public statement or announcement to employees without the other Party's prior consent (which consent shall not be unreasonably withheld, conditioned or delayed). Notwithstanding the foregoing: (a) each Party may, without such consultation or consent, make any public statement in response to questions from the press, analysts, investors or those attending industry conferences, make internal announcements to employees and make disclosures in Company SEC Documents, so long as such statements are consistent in all material respects with information previously disclosed in previous press releases, public disclosures or public statements made jointly by the Parties in compliance with this Section 5.8 (or individually, if approved by the other Party) and do not contain material nonpublic information regarding this Agreement or the Transactions; (b) a Party may, without the prior consent of the other Party but subject to giving advance notice to the other Party and, to the extent reasonably practical, allow such other Party to comment on such release, announcement or statement, issue any such press release or make any such public announcement or statement as may be required by Legal Requirement or applicable stock exchange rule; and (c) neither Party shall be required to consult with the other or obtain the other's approval in connection with such portion of any press release, public statement or filing to be issued or made pursuant to Section 4.3(e)(i) or with respect to any Acquisition Proposal or Company Adverse Change Recommendation (it being understood that nothing in this Section 5.8 shall limit the Company's obligations under Section 4.3 and Section 5.1).
- **5.9 Takeover Laws**. If any Takeover Law may become, or may purport to be, applicable to the Transactions, each of Parent and the Company and the members of their respective boards of directors shall use their respective reasonable best efforts to grant such approvals and take such actions as are necessary so that the Transactions may be consummated as promptly as practicable on the terms and conditions contemplated hereby and otherwise act to lawfully eliminate the effect of any Takeover Law on any of the Transactions.
- **5.10** Section 16 Matters. The Company, and the Board of Directors, shall, to the extent necessary, take appropriate action, prior to or as of the Effective Time, to approve, for purposes of Section 16(b) of the Exchange Act, the disposition and cancellation or deemed disposition and cancellation of Shares, Company Options, Company RSUs and Company PSUs in the Merger by applicable individuals and to cause such dispositions and/or cancellations to be exempt under Rule 16b-3 promulgated under the Exchange Act.

5.11 Credit Agreement. The Company shall use reasonable best efforts to deliver (or cause to be delivered) to Parent (preceded by a draft of the same at least five (5) business days earlier) prior to the Closing Date a customary payoff letter (or similar document) with respect to the Credit Agreement. from the lenders (or an agent on behalf of all lenders) under the Credit Agreement (the "Payoff Letter"), which Payoff Letter shall specify the aggregate payoff amount required to be paid to fully satisfy all Indebtedness outstanding under the Credit Agreement (including any "Prepayment Charge," as such term is defined in the Credit Agreement, but other than (x) any inchoate indemnity obligations thereunder, (y) any obligations under certain bank product services agreements that are cash collateralized in accordance with the applicable terms thereof and (z) any other obligations which, by their terms, are to survive the termination of the Credit Agreement) and provide that, upon receipt of such specified amount, all Encumbrances (and any guarantees) granted in connection therewith relating to the assets, rights and properties of the Company and its Subsidiaries securing such Indebtedness shall be released and terminated. The Company shall use its reasonable best efforts to deliver to the agent under the Credit Agreement, at least five (5) "Business Days," as such term is defined in the Credit Agreement, prior to the anticipated Closing Date, a written notice of prepayment of all outstanding Indebtedness under the Credit Agreement, which notice shall provide that such prepayment is subject to and conditioned upon the consummation of the Closing (or obtain a waiver thereof in the Payoff Letter). The Company shall, and shall cause the other applicable Acquired Corporations to, use their reasonable best efforts to deliver (or cause the agent or lenders under the Credit Agreement to deliver) to Parent on or prior to the Closing (including by way of attachments to the Payoff Letter), all customary related release documents, filings and notices required to effect the release of all Encumbrances securing the obligations under the Credit Agreement (other than (x) any inchoate indemnity obligations thereunder, (y) any obligations under certain bank product services agreements that are cash collateralized in accordance with the applicable terms thereof and (z) any other obligations which, by their terms, are to survive the termination of the Credit Agreement), including (and in each case to the extent applicable) the filing of UCC termination statements, terminations of control agreements, terminations of intellectual property security agreements and arrange for delivery of possessory collateral (to the extent reasonably practicable on the Closing Date), which shall in each case be subject to and conditioned upon the occurrence of the Closing and the repayment in full of all obligations then outstanding under the Credit Agreement. On the Closing Date, Parent shall effect or cause to be effected the repayment in full of all Indebtedness outstanding under the Credit Agreement, in accordance with the Credit Agreement and the Payoff Letter. The Parties shall reasonably cooperate with each other in connection with their respective obligations set forth in this Section 5.11.

5.12 Stock Exchange Delisting; Deregistration. Prior to the Closing Date, the Company shall cooperate with Parent and use its reasonable best efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under applicable Legal Requirements and rules and policies of Nasdaq to enable the delisting by the Surviving Corporation of the Shares from Nasdaq and the deregistration of the Shares under the Exchange Act as promptly as practicable after the Effective Time.

SECTION 6

CONDITIONS PRECEDENT TO THE MERGER

- **6.1** Conditions to Each Party's Obligation to Effect the Merger. The respective obligations of each Party to effect the Merger are subject to the satisfaction (or waiver by the Company, Parent and Merger Sub, to the extent permitted by applicable Legal Requirements) on or prior to the Closing Date of the following conditions:
 - (a) Company Stockholder Approval. The Company Stockholder Approval shall have been obtained.
- (b) <u>Approvals Under Antitrust Laws</u>. (i) Any waiting period (or any extension thereof) under the HSR Act applicable to the Merger and any commitment by the Parties not to consummate the Transactions before a certain date under a timing agreement shall have expired or been terminated, and (ii) any waiting period, clearance or affirmative approval of any Specified Antitrust Authority (other than under the HSR Act) with respect to the Transactions shall have been obtained and any mandatory waiting period related thereto shall have expired or been terminated.
- (c) No Restraints. There shall not have been issued by any Governmental Body of competent jurisdiction in a jurisdiction where either Parent and its Affiliates or the Acquired Corporations operate their respective businesses or own any material assets (a "Specified Governmental Body") and remain in effect any temporary restraining order, preliminary or permanent injunction or other order, directive, judgment, decree or ruling preventing the consummation of the Merger, nor shall any Legal Requirement have been promulgated, entered, enforced, enacted, issued or deemed applicable to the Merger by any Specified Governmental Body which, directly or indirectly, prohibits or makes illegal the consummation of the Merger.
- **6.2** Conditions to Obligations of Parent and Merger Sub to Effect the Merger. The obligations of Parent and Merger Sub to effect the Merger are further subject to the satisfaction (or waiver by Parent, on its own behalf and on behalf of Merger Sub, to the extent permitted by applicable Legal Requirements) on or prior to the Closing Date of the following conditions:
- Representations and Warranties. (i) The representations and warranties of the Company set forth in (A) the first two sentences of Section 2.1(a) and the first two sentences of Section 2.1(b) (Due Organization; Subsidiaries, Etc.), Section 2.2 (Certificate of Incorporation and Bylaws), clauses (b), (c), (e) and (f) of Section 2.3 (Capitalization, Etc.), Section 2.21 (Authority; Binding Nature of Agreement), Section 2.22 (Takeover Laws), Section 2.24 (Opinion of Financial Advisors), and Section 2.25 (Brokers and Other Advisors) shall be accurate in all material respects (without taking into account any "Material Adverse Effect" and "materiality" qualifications contained in such representations and warranties) as of the date of this Agreement and at and as of Closing Date as if made on and as of the Closing Date (except to the extent any such representation or warranty expressly relates to an earlier date or period, in which case as of such date or period), (B) clauses (a) and (d) of Section 2.3 (Capitalization, Etc.) shall be accurate in all respects except for any de minimis inaccuracies as of the date of this Agreement and at and as of the Closing Date as if made on and as of the Closing Date (except to the extent any such representation or warranty expressly relates to an earlier date or period, in which case as of such date or period), and (C) Section 2.5(b) (No Material Adverse Effect) shall be accurate in all respects as of the date of this Agreement and at and as of the Closing Date as if made on and as of the Closing Date with respect to the earlier period set forth in Section 2.5(b); and (ii) the representations and warranties of the Company set forth in this Agreement, other than those referred to in clauses (A), through (C) above, shall be accurate (without taking into account any "Material Adverse Effect" and "materiality" qualifications contained in such representations and warranties) as of the date of this Agreement and at and as of the Closing Date as if made on and as of the Closing Date (except to the extent any such representation or warranty expressly relates to an earlier date or period, in which case as of such date or period), except where the failure of such representations and warranties to be so accurate has not had, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

- (b) <u>Performance of Obligations of the Company</u>. The Company shall have complied with or performed in all material respects the covenants and agreements it is required to comply with or perform at or prior to the Closing Date.
- (c) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Material Adverse Effect which is continuing.
- (d) <u>Delivery of Officer Certificates</u>. Parent and Merger Sub shall have received a certificate, dated the Closing Date, executed on behalf of the Company by the Company's Chief Executive Officer or Chief Financial Officer certifying to the effect that the conditions set forth in <u>Sections 6.2(a)</u>, <u>6.2(b)</u> and <u>6.2(c)</u> have been satisfied.
- **6.3** Conditions to Obligations of the Company to Effect the Merger. The obligations of the Company to effect the Merger are subject to the satisfaction (or waiver by the Company, to the extent permitted by applicable Legal Requirements) on or prior to the Closing Date of the following conditions:
- (a) Representations and Warranties. (i) The representations and warranties of Parent and Merger Sub set forth in (A) Sections 3.1 (Due Organization), 3.2 (Merger Sub) and 3.3 (Authority; Binding Nature of Agreement) shall be accurate in all material respects (without taking into account any "Parent Material Adverse Effect" and "materiality" qualifications contained in such representations and warranties) as of the date of this Agreement and at and as of Closing Date as if made on and as of the Closing Date (except to the extent any such representation or warranty expressly relates to an earlier date or period, in which case as of such date or period), and (B) Section 3.10 (Brokers and Other Advisors) shall be accurate in all respects as of the date of this Agreement and at and as of the Closing Date as if made on and as of the Closing Date; and (ii) the representations and warranties of Parent and Merger Sub set forth in this Agreement, other than those referred to in clause (i) above, shall be accurate (without taking into account any "Parent Material Adverse Effect" and "materiality" qualifications contained in such representations and warranties) as of the date of this Agreement and at and as of the Closing Date as if made on and as of the Closing Date (except to the extent any such representation or warranty expressly relates to an earlier date or period, in which case as of such date or period), except where the failure of such representations and warranties to be so true and correct has not had, and would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

- (b) <u>Performance of Obligations of Parent and Merger Sub</u>. Parent and Merger Sub shall have complied with or performed in all material respects the covenants and agreements they are required to comply with or perform at or prior to the Closing Date.
- (c) <u>Delivery of Officer Certificate</u>. Parent shall have delivered to the Company a certificate, dated the Closing Date and signed on its behalf by its Chief Executive Officer or another senior executive officer, certifying to the effect that the conditions set forth in <u>Sections 6.3(a)</u> and <u>6.3(b)</u> have been satisfied.

SECTION 7

TERMINATION

- **7.1 Termination**. This Agreement may be terminated prior to the Effective Time:
 - (a) by mutual written consent of Parent and the Company at any time prior to the Closing;
- (b) by either Parent or the Company, at any time prior to the Closing, if the Closing shall not have occurred on or prior to the later of (such time, as applicable, the "End Date"): (i) midnight Eastern Time, on October 27, 2023 (the "Initial End Date"), (ii) if on the Initial End Date all of the conditions set forth in Section 6, other than any condition set forth in Sections 6.1(b) or 6.1(c) (solely in respect of Antitrust Laws), shall have been satisfied or waived, to the extent waivable (other than conditions that by their nature are to be satisfied on the Closing Date, each of which is then capable of being satisfied), midnight Eastern Time, on the date that is ninety (90) days after the Initial End Date (the "Extended End Date"), and (iii) if on the Extended End Date all of the conditions set forth in Section 6, other than any condition set forth in Sections 6.1(b) or 6.1(c) (solely in respect of Antitrust Laws), shall have been satisfied or waived, to the extent waivable (other than conditions that by their nature are to be satisfied on the Closing Date, each of which is then capable of being satisfied), midnight Eastern Time, on the date that is one hundred and eighty (180) days after the Initial End Date; provided, that the right to terminate this Agreement pursuant to this Section 7.1(b) shall not be available to any Party whose material breach of this Agreement has proximately caused the Merger not being consummated by such date;
- (c) by either Parent or the Company if a Specified Governmental Body of competent jurisdiction shall have issued any permanent injunction or other order, directive, judgment, decree or ruling, or shall have taken any other action, having the effect of permanently restraining, enjoining or otherwise prohibiting the acceptance for payment of Shares pursuant to the Merger or making the consummation of the Merger illegal, which order, directive, judgment, decree, ruling or other action shall be final and nonappealable; *provided*, *however*, that the right to terminate this Agreement pursuant to this Section 7.1(c) shall not be available to any Party whose material breach of this Agreement has proximately caused the issuance of such final and nonappealable order, directive, judgment, decree, ruling or other action;

- by Parent at any time prior to the Cut-off Time, if: (i) the Board of Directors shall have failed to include the Company Board Recommendation in the Merger Proxy Statement when filed with the SEC or mailed, or shall have effected a Company Adverse Change Recommendation: (ii) in the case of an Acquisition Proposal structured as a tender offer or exchange offer subject to Regulation 14D under the Exchange Act, the Board of Directors (A) states that it recommends such tender or exchange offer or expresses no opinion or is unable to take a position (other than a "stop, look and listen" communication pursuant to Rule 14d-9(f) promulgated under the Exchange Act) with respect to such tender or exchange offer or (B) fails to recommend, in a Solicitation/Recommendation Statement on Schedule 14D-9, rejection of such tender offer or exchange offer and reaffirm the Company Board Recommendation within ten (10) business days of the commencement of such tender offer or exchange offer within the meaning of Rule 14d-2 under the Exchange Act (or, if earlier, by the close of business on the business day immediately preceding the then scheduled Cut-off Time); (iii) after any public announcement of an Acquisition Proposal (other than a tender offer or exchange offer), the Board of Directors fails to publicly affirm the Company Board Recommendation within three (3) business days after receipt of a written request by Parent to do so (or, if earlier, by the close of business on the business day immediately preceding the then scheduled Cut-off Time, provided that the Company has received such request prior to the second (2nd) business day before the then scheduled date of the Cut-off Time); provided, that Parent may only make such request twice with respect to each Acquisition Proposal or material modification thereof: or (iv) the Board of Directors fails to publicly reaffirm the Company Board Recommendation within three (3) business days after receipt of a written request by Parent to do so (it being understood that the Company will have no obligation to make such reaffirmation pursuant to this prong (iv) on more than three (3) occasions); or (v) the Board of Directors or the Company intentionally breaches its obligations under Section 4.3 or Section 5.1 in any material respects.
- (e) by the Company, at any time prior to the Cut-off Time, in order to substantially concurrently with such termination enter into a binding written definitive acquisition agreement providing for the consummation of the Superior Offer approved by the Board of Directors pursuant to, and in accordance with the provisions of, Section 5.1(b) (a "Specified Agreement"); provided that such termination shall be effective only if the Company shall have paid the Termination Fee immediately prior to or substantially concurrently with such termination;
- (f) by either Parent or the Company if the Company Stockholder Approval shall not have been obtained at the Company Stockholder Meeting duly convened and held and at which a vote on the matter was taken;
- (g) by Parent at any time prior to the Closing, if a breach of any representation or warranty contained in this Agreement or failure to perform any covenant or obligation in this Agreement on the part of the Company shall have occurred such that a condition set forth in clauses (a), (b) or (c) of Section 6.2 would not be satisfied and cannot be cured by the Company by the End Date, or if capable of being cured in such time period, shall not have been cured within thirty (30) days of the date Parent gives the Company written notice of such breach or failure to perform; provided, however, that Parent shall not have the right to terminate this Agreement pursuant to this Section 7.1(g) if either Parent or Merger Sub is then in breach of any representation, warranty, covenant or obligation hereunder which breach would permit the Company to terminate this Agreement pursuant to Section 7.1(h); or

- (h) by the Company at any time prior to the Closing, if a breach of any representation or warranty contained in this Agreement or failure to perform any covenant or obligation in this Agreement on the part of Parent or Merger Sub shall have occurred, (i) which breach or failure would reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect or (ii) such that a condition set forth in Section 6.3(a) or Section 6.3(b) would not be satisfied, and such breach or failure cannot be cured by Parent or Merger Sub, as applicable, by the End Date, or, if capable of being cured in such time period, shall not have been cured within thirty (30) days of the date the Company gives Parent written notice of such breach or failure to perform; provided, however, that the Company shall not have the right to terminate this Agreement pursuant to this Section 7.1(h) if the Company is then in breach of any representation, warranty, covenant or obligation hereunder which breach would permit Parent to terminate this Agreement pursuant to Section 7.1(g).
- 7.2 Effect of Termination. In the event of the termination of this Agreement as provided in Section 7.1, written notice thereof shall be given to the other Party or Parties, specifying the provision hereof pursuant to which such termination is made, and this Agreement shall be of no further force or effect and there shall be no liability on the part of Parent, Merger Sub or the Company or any of their respective former, current or future officers, directors, partners, stockholders, managers, members or Affiliates following any such termination; provided, however, that (a) the final sentence of Section 4.1(a), this Section 7.2, Section 7.3 and Section 8 (other than Section 8.5(b) and Section 8.10(b)) shall survive the termination of this Agreement and shall remain in full force and effect, (b) the Confidentiality Agreement shall survive the termination of this Agreement and shall remain in full force and effect in accordance with its terms and (c) the termination of this Agreement shall not relieve any Party from any liability for fraud or willful and material breach of this Agreement prior to termination.

7.3 Expenses; Termination Fees.

- (a) Except as set forth in this <u>Section 7.3</u>, all fees and expenses incurred in connection with this Agreement and the Transactions shall be paid by the Party incurring such expenses, whether or not the Merger is consummated.
 - (b) In the event that:
 - (i) this Agreement is terminated by the Company pursuant to Section 7.1(e);
 - (ii) this Agreement is terminated by Parent pursuant to Section 7.1(d); or

(iii) (A) this Agreement is terminated by Parent or the Company pursuant to Section 7.1(b) (but in the case of a termination by the Company, only if at such time Parent would not be prohibited from terminating this Agreement pursuant to proviso (z) to Section 7.1(b)), or Section 7.1(f), or terminated by Parent pursuant to Section 7.1(g), (B) any Person shall have publicly disclosed a bona fide Acquisition Proposal or an Acquisition Proposal has otherwise been communicated to the Board of Directors after the execution and delivery of this Agreement and prior to such termination and such Acquisition Proposal has not been unconditionally and, in the case of a publicly disclosed Acquisition Proposal, publicly withdrawn prior to such termination and (C) within twelve (12) months of such termination the Company shall have entered into a definitive agreement with respect to, or consummated, an Acquisition Proposal; provided that for purposes of this clause (C) the references to "15%" in the definition of "Acquisition Proposal" shall be deemed to be references to "50%";

then, in any such event under <u>clause (i), (ii)</u> or <u>(iii)</u> of this <u>Section 7.3(b)</u>, the Company shall pay to Parent or its designee the Termination Fee by wire transfer of same day funds (x) in the case of <u>Section 7.3(b)(ii)</u>, prior to or substantially concurrently with (and as a condition to the effectiveness of) such termination, (y) in the case of <u>Section 7.3(b)(ii)</u>, within one (1) business day after such termination or (z) in the case of <u>Section 7.3(b)(iii)</u>, prior to or concurrently with the earlier of entering into the definitive agreement with respect to, or consummating, the Acquisition Proposal referred to in <u>clause (C)</u> of <u>Section 7.3(b)(iii)</u>; it being understood that in no event shall the Company be required to pay the Termination Fee on more than one occasion. As used herein, "<u>Termination Fee</u>" shall mean a cash amount equal to \$222,370,000. Payment of the Termination Fee pursuant to this <u>Section 7.3(b)</u>, together with any amounts that become due pursuant to <u>Section 7.3(d)</u>, shall be deemed to be liquidated damages for any and all losses or damages suffered or incurred by Parent, Merger Sub, any of their respective Affiliates or any other Person in connection with this Agreement (and the termination hereof), the Transactions (and the abandonment thereof) or any matter forming the basis for such termination, and none of Parent, Merger Sub or any of their respective Affiliates shall be entitled to bring or maintain any claim, action or proceeding against the Company or any of its Affiliates arising out of or in connection with this Agreement, any of the Transactions or any matters forming the basis for such termination; *provided*, that the foregoing shall not relieve any Company Related Party from any liability for fraud or willful breach of this Agreement prior to such termination. Notwithstanding anything in this Agreement to the contrary, in the event this Agreement is terminated by the Company for any reason at a time when Parent would have had the right to terminated this Agreement a

(c) In the event of any termination described in Section 7.3(b), (i) payment from the Company to Parent of the Termination Fee pursuant to Section 7.3(b), together with any amounts that become due under Section 7.3(d), shall be the sole and exclusive remedy of Parent, Merger Sub or any of their respective Affiliates against the Company Related Parties for any loss suffered as a result of the failure of the Merger to be consummated or for a breach or failure to perform hereunder or otherwise, and (ii) upon payment of such amount(s), none of the Company Related Parties shall have any further liability or obligation relating to or arising out of this Agreement or the Transactions; provided, that the foregoing shall not relieve any Company Related Party from any liability for fraud or willful breach of this Agreement prior to such termination; provided, further, that Parent may seek specific performance to cause the Company to consummate the Transactions in accordance with Section 8.5(b), but in no event shall Parent be entitled to both specific performance and the payment of the Termination Fee.

(d) The Parties acknowledge that the agreements contained in this Section 7.3 are an integral part of the Transactions, that the Termination Fee is not a penalty, but a reasonable amount that will compensate Parent and Merger Sub in the circumstances in which such payment is payable for the efforts and resources expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the Transactions and that, without these agreements, the Parties would not enter into this Agreement; accordingly, if the Company fails to timely pay any amount due pursuant to Section 7.3(b), and, in order to obtain the payment, Parent commences a Legal Proceeding which results in a judgment against the Company, the Company shall pay Parent its reasonable and documented costs and expenses (including reasonable and documented attorneys' fees) in connection with such suit, together with interest on such amount at the prime rate as published in the Wall Street Journal in effect on the date such payment was required to be made through the date such payment was actually received.

SECTION 8

MISCELLANEOUS PROVISIONS

- **8.1** Amendment. Prior to the Effective Time, this Agreement may be amended with the approval of the respective Boards of Directors of the Company, Parent and Merger Sub at any time. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the Parties.
- **8.2** Waiver. No failure on the part of any Party to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any Party in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy such that rights and remedies provided herein shall be cumulative and not exclusive of any rights or remedies provided by any Legal Requirement. At any time prior to the Effective Time, Parent and Merger Sub, on the one hand, and the Company, on the other hand, may, to the extent permissible under applicable Legal Requirements, (a) extend the time for the performance of any of the obligations or other acts of the other, (b) waive any breach of the representations and warranties of the other contained herein or in any document delivered pursuant hereto or (c) waive compliance by the other with any of the agreements or covenants contained herein; *provided*, *however*, that after receipt of the Company Stockholder Approval, no waiver shall be made which by applicable Legal Requirement requires further approval by the holders of Shares without obtaining such further approval. Any such extension or waiver shall be valid only if is expressly set forth in a written instrument duly executed and delivered on behalf of the Party or Parties to be bound thereby, but such extension or waiver or failure to insist on strict compliance with an obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

- **8.3** No Survival of Representations and Warranties. None of the representations and warranties contained in this Agreement, the Company Disclosure Schedule or in any certificate or schedule or other document delivered by any Person pursuant to this Agreement shall survive the Effective Time.
- **8.4 Entire Agreement; Counterparts.** This Agreement (including its Exhibits, Annexes and the Company Disclosure Schedule) and the Confidentiality Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the Parties and their respective Affiliates, with respect to the subject matter hereof and thereof. This Agreement may be executed in one or more counterparts, including by facsimile or by email with .pdf attachments, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Parties.

8.5 Applicable Legal Requirements; Jurisdiction; Specific Performance; Remedies.

- (a) This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof. In any action or proceeding arising out of or relating to this Agreement or any of the Transactions: (i) each of the Parties irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Chancery Court of the State of Delaware and any state appellate court therefrom or, if (but only if) such court lacks subject matter jurisdiction, the United States District Court sitting in New Castle County in the State of Delaware and any appellate court therefrom (collectively, the "Delaware Courts"); and (ii) each of the Parties irrevocably consents to service of process by first class certified mail, return receipt requested, postage prepaid, to the address at which such Party is to receive notice in accordance with Section 8.8. Each of the Parties irrevocably and unconditionally (1) agrees not to commence any such action or proceeding except in the Delaware Courts, (2) agrees that any claim in respect of any such action or proceeding may be heard and determined in the Delaware Courts, (3) waives, to the fullest extent it may legally and effectively do so, any objection that it may now or hereafter have to the jurisdiction or laying of venue of any such action or proceeding in the Delaware Courts and (4) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in the Delaware Courts. The Parties agree that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Legal Requirements; provided, however, that nothing in the foregoing shall restrict any Party's rights to seek any post-judgment relief regarding, or any appeal from, such final trial court judgment.
- (b) The Parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the Parties do not perform their obligations under the provisions of this Agreement in accordance with its specified terms or otherwise breach such provisions. Subject to the following sentence, the Parties acknowledge and agree that (i) the Parties shall be entitled to an injunction or injunctions, specific performance, or other equitable relief, to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in Section 8.5(a) without proof of damages or otherwise, this being in addition to any other remedy to which they are entitled under this Agreement, and (ii) the right of specific performance is an integral part of the Transactions and without that right, neither the Company nor Parent would have entered into this Agreement. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance or other equitable relief on the basis that the other Parties have an adequate remedy at law or that an award of specific performance is not an appropriate remedy for any reason at law or equity. The Parties acknowledge and agree that any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 8.5(b) shall not be required to provide any bond or other security in connection with any such order or injunction.

- (c) EACH OF THE PARTIES IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING BETWEEN THE PARTIES (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE), INCLUDING ANY COUNTERCLAIM, ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THE ACTIONS OF ANY PARTY HERETO IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT THEREOF. EACH PARTY (I) MAKES THIS WAIVER VOLUNTARILY AND (II) ACKNOWLEDGES THAT SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS CONTAINED IN THIS SECTION 8.5(c).
- **8.6** Assignability. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the Parties and their respective successors and permitted assigns; *provided*, *however*, that neither this Agreement nor any of the rights hereunder may be assigned by a Party without the prior written consent of the other Parties, and any attempted assignment of this Agreement or any of such rights without such consent shall be void and of no effect, except that Parent may designate, by written notice to the Company, another wholly-owned direct or indirect Subsidiary of Parent (provided that it is a Delaware corporation) to act as Merger Sub hereunder, in which event all references to "Merger Sub" in this Agreement (and in Exhibit A) shall be deemed references to such other Subsidiary.
- **8.7 No Third Party Beneficiaries**. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the Parties) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement; except for: (a) if the Effective Time occurs, (i) the right of the Company's stockholders to receive the Merger Consideration pursuant to Section 1 following the Effective Time in accordance with the terms of this Agreement, and (ii) the right of the holders of Company Options, Company RSUs and Company PSUs to receive the Merger Consideration pursuant to Section 1.8 following the Effective Time in accordance with the terms of this Agreement; (b) the provisions set forth in Section 5.5; and (c) the limitations on liability of the Company Related Parties set forth in Section 7.3(c).

8.8 Notices. Any notice or other communication required or permitted to be delivered to any Party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) upon receipt when delivered by hand, (b) upon receipt of proof of delivery on a business day before 6:00 p.m. in the time zone of the receiving Party, otherwise upon the following business day after receipt of proof of delivery if sent by registered mail or by courier or express delivery service, (c) if sent by email prior to 6:00 p.m. recipient's local time, upon transmission when receipt is confirmed, or (d) if sent by email after 6:00 p.m. recipient's local time and receipt is confirmed, the business day following the date of transmission; provided that in each case the notice or other communication is sent to the physical address or email address set forth beneath the name of such Party below (or to such other physical address or email address as such Party shall have specified in a written notice given to the other Parties):

if to Parent, Merger Sub or Guarantor (or following the Effective Time, the Surviving Corporation):

c/o Astellas Pharma US, Inc. 2375 Waterview Drive Northbrook, IL 60062-6111 Attn: General Counsel Email: catherine.levitt@astellas.com

with a copy (which shall not constitute notice) to:

Jones Day 3161 Michelson Drive, Suite 800 Irvine CA 92612 Attn: Jonn Beeson; Ben Chouka

Email: jbeeson@jonesday.com; bchouka@jonesday.com

if to the Company (prior to the Effective Time):

IVERIC bio, Inc. 8 Sylvan Way Parsippany, NJ 07054 Attn: General Counsel

Email: todd.anderman@ivericbio.com

with a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP 500 Boylston Street Boston, MA 02116 Attn: Graham Robinson; Laura Knoll

Email: graham.robinson@skadden.com; laura.knoll@skadden.com

8.9 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the Parties agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the Parties agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

8.10 Obligation of Parent; Guaranty.

- (a) Parent shall cause Merger Sub (and, following the Effective Time, the Surviving Corporation) to duly perform, satisfy and discharge on a timely basis each of the covenants, obligations and liabilities applicable to Merger Sub or the Surviving Corporation, as applicable, under this Agreement, as though each of such covenants, obligations and liabilities was binding upon Parent.
- (b) Guarantor hereby unconditionally and irrevocably guarantees to the Company, the Surviving Corporation and their successors and assigns the payment and performance of each of the covenants, obligations and liabilities applicable to Parent, Merger Sub or the Surviving Corporation, as applicable, and their respective successors and assigns under this Agreement (the "Guaranteed Obligations"). This guaranty is an absolute, unconditional and continuing guaranty of the full and punctual discharge and performance of the Guaranteed Obligations. This guaranty is a guaranty of payment and performance and not of collection. Guarantor expressly waives any requirement that any Person exhaust any right, remedy or power or proceed against Parent or Merger Sub (or any of their successors or assigns) under this Agreement or against any other Person under any other guaranty of, or security for, any of the Guaranteed Obligations. Should Merger Sub or Parent (or any of their successors or assigns) default in the timely discharge or performance of all or any portion of the Guaranteed Obligations, Guarantor shall immediately fully and punctually discharge and perform such Guaranteed Obligations. So long as this Section 8.10(b) is in effect, Guarantor shall not exercise any right or remedy arising by reason of its performance of its guaranty, whether by subrogation, reimbursement, indemnification, contribution or otherwise, against the Company, the Surviving Corporation or their successors and assigns or any express intended third party beneficiary of any Guaranteed Obligations described in Section 8.7, or any other guarantor of the Guaranteed Obligations or any security therefor. Guarantor represents and warrants to the Company as of the date of this Agreement and as of the Closing Date as if made as of the Closing Date (except to the extent any representation or warranty expressly relates to an earlier date or period, in which case as of such date or period) as follows:

- (i) Guarantor is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all necessary power and authority: (A) to conduct its business in the manner in which its business is currently being conducted; and (B) to own and use its assets in the manner in which its assets are currently owned and used, except as would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect;
- (ii) Guarantor has the corporate power and authority to execute and deliver and perform its obligations under this Section 8.10(b). The board of directors of Guarantor has approved the execution, delivery and performance of this Section 8.10(b) by Guarantor. This Section 8.10(b) has been duly executed and delivered by Guarantor, and assuming due authorization, execution and delivery of this Agreement by the Company, this Section 8.10(b) constitutes the legal, valid and binding obligation of Guarantor and is enforceable against Guarantor in accordance with its terms, except as such enforcement may be subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar laws of general applicability relating to or affecting creditors' rights, and by general equitable principles;
- (iii) Assuming compliance with the applicable provisions of the DGCL, the HSR Act and any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any), and the rules and regulations of the SEC and Nasdaq, the execution and delivery of this Section 8.10(b) by Guarantor, and the consummation of the transactions contemplated by this Section 8.10(b), will not: (A) cause a violation of any of the provisions of the organizational documents of Guarantor; (B) cause a violation by Guarantor of any Legal Requirement or order applicable to Guarantor, or to which Guarantor is subject; or (C) require any consent or notice under, conflict with, result in breach of, or constitute a default under (or an event that with notice or lapse of time or both would become a default), or give rise to any right of purchase, termination, amendment, cancellation, acceleration or other adverse change of any right or obligation or the loss of any benefit to which Guarantor is entitled under any provision of any Contract, except in the case of the foregoing clauses (B) and (C), as would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect:

- (iv) Except for the filing of the certificate of merger with the Secretary of State of the State of Delaware or as may be required by the Exchange Act, Takeover Laws, the DGCL, the HSR Act and any applicable filing, notification or approval in any foreign jurisdiction required by Antitrust Laws (if any) and the applicable rules and regulations of the SEC and any national securities exchange, Guarantor is not required to give notice to, make any filing with or obtain any Consent from any Governmental Body at any time prior to the Closing in connection with the execution and delivery of this Agreement by Parent or Merger Sub or, as provided by Section 8.10(b), Guarantor, or the consummation by Parent or Merger Sub of the Merger or the other Transactions, except those that the failure to make or obtain would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect. No vote of Guarantor's stockholders is necessary to approve this Agreement or any of the Transactions;
- (v) As of the date of this Agreement, there is no Legal Proceeding pending and served or, to the knowledge of Guarantor, pending and not served, against Guarantor, except as would not, and would not reasonably be expected to, individually or in the aggregate, have a Parent Material Adverse Effect. To the knowledge of Guarantor, as of the date of this Agreement, Guarantor is not subject to any continuing order of, consent decree, settlement agreement or similar written agreement with, or continuing investigation by, any Governmental Body, or any order, writ, judgment, injunction, decree, determination or award of any Governmental Body, except as would not, and would not reasonably be expected to, individually or in the aggregate, have a Parent Material Adverse Effect; and
 - (vi) Guarantor acknowledges and agrees to all matters set forth in Section 3.9.

Guarantor acknowledges and agrees that all provisions of Section 8 (other than Section 8.10(a) and this Section 8.10(b)) binding upon Parent with respect to the Agreement shall apply to the Guarantor *mutatis mutandis* with respect to this Section 8.10(b). Guarantor's obligations under this Section 8.10(b) are expressly limited to the Guaranteed Obligations and shall automatically expire upon the full discharge and performance of all Guaranteed Obligations and thereafter, Guarantor shall no longer have any duties or obligations under this Agreement.

8.11 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include the masculine and feminine genders.
- (b) The Parties agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.

- (c) As used in this Agreement, the word "including" and words of similar import shall mean including without limiting the generality of any description preceding or following such term, unless otherwise specified.
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" or "Annexes" are intended to refer to Sections of this Agreement and Exhibits or Annexes to this Agreement.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.
 - (f) The term "dollars" and character "\$" shall mean United States dollars.
- (g) The phrases "made available" and "delivered," when used in reference to any documents or information made available to Parent, Merger Sub or any of their respective Representatives prior to the execution of this Agreement, shall be deemed to mean (i) uploaded to, and accessible to Parent, Merger Sub or any of their respective Representatives in, the online data rooms hosted on behalf of the Company by Firmex under the name "Project Terroir Corporate Data Room" and "IVERIC bio ACP (Astellas)" in complete and unredacted form at least 48 hours prior to the execution and delivery of this Agreement or (ii) provided via email by the Company or its Representatives to Parent, Merger Sub or their respective Representatives in complete and unredacted form at least 48 hours prior to the execution and delivery of this Agreement.

[Signature pages follow]

IVERIC BIO, INC.

By: /s/ Glenn P. Sblendorio

Name: Glenn P. Sblendorio
Title: Chief Executive Officer

ASTELLAS US HOLDING, INC.

By: /s/ Mark Reisenauer

Name: Mark Reisenauer Title: President

BERRY MERGER SUB, INC.

By: /s/ Mark Reisenauer

Name: Mark Reisenauer Title: President

ASTELLAS PHARMA INC.

By: /s/ Naoki Okamura

Name: Naoki Okamura

Title: President & Chief Executive Officer

EXHIBIT A

CERTAIN DEFINITIONS

For purposes of the Agreement (including this **Exhibit A**):

Acceptable Confidentiality Agreement. "Acceptable Confidentiality Agreement" is defined in Section 4.3(a) of the Agreement.

Acquired Corporations. "Acquired Corporations" is defined in Section 2.1(a) of the Agreement.

Acquisition Proposal. "Acquisition Proposal" shall mean any proposal or offer from any Person (other than Parent and its Affiliates) or "group", within the meaning of Section 13(d) of the Exchange Act, relating to, in a single transaction or series of related transactions, any (a) acquisition, transfer, disposition or license of assets of any Acquired Corporation equal to 15% or more of the Company's consolidated assets (based on the fair market value thereof), (b) issuance or acquisition of (i) 15% or more of the outstanding Company Common Stock or other voting or equity securities of the Company, (ii) securities and indebtedness that would, in the aggregate, represent 15% or more of the outstanding voting power of any class of the Company's securities, or (iii) any options, rights or warrants to purchase or securities convertible into or exchangeable for equity or debt interests described in the foregoing clauses (i) or (ii) (the forgoing securities described in this clause (b), "Designated Securities"), (c) recapitalization, tender offer or exchange offer that if consummated would result in any Person or group beneficially owning 15% or more of the Designated Securities, or (d) merger, consolidation, amalgamation, share exchange, business combination, recapitalization, liquidation, dissolution or similar transaction involving an Acquired Corporation that if consummated would result in any Person or "group" beneficially owning 15% or more of the Designated Securities, in each case of clauses (a) through (d), other than the Transactions.

Affiliate. "Affiliate" shall mean, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. For this purpose, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by Contract or otherwise.

Agreement. "Agreement" is defined in the preamble to the Agreement.

Anti-Corruption Laws. "Anti-Corruption Laws" shall mean the Foreign Corrupt Practices Act of 1977, as amended, the Anti-Kickback Act of 1986, as amended, the UK Bribery Act of 2010, and the Anti-Bribery Laws of the People's Republic of China or any applicable Legal Requirements of similar effect, and the related regulations and published interpretations thereunder.

Antitrust Laws. "Antitrust Laws" shall mean the Sherman Act, as amended, the Clayton Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, all applicable foreign anti-trust laws and all other applicable Legal Requirements issued by a Governmental Body that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition.

Applicable Date. "Applicable Date" is defined in Section 2.4(a) of the Agreement.

Board of Directors. "Board of Directors" is defined in the Introduction to the Agreement.

Book-Entry Shares. "Book-Entry Shares" shall mean non-certificated Shares represented by book-entry.

business day. "business day" shall mean a day except a Saturday, a Sunday or other day on which banks in the City of New York or in Tokyo, Japan are authorized or required by Legal Requirements to be closed.

Capitalization Date. "Capitalization Date" is defined in Section 2.3(a) of the Agreement.

CARES Act. "CARES Act" is defined in the definition of CARES Act and COVID Relief Programs.

CARES Act and COVID Relief Programs. "CARES Act and COVID Relief Programs" mean, collectively, the Coronavirus Aid, Relief, and Economic Security Act (H.R. 748, Pub Law 116-136) (the "CARES Act"), the Families First Coronavirus Response Act (H.R.6201, Pub Law 116-127), Paycheck Protection Program Flexibility Act of 2020 (H.R. 7010, Pub Law 116-142), American Rescue Plan Act (H.R. 1319, Pub Law 117-7) and all FAQs or interim final rules (to the extent having the force of a Legal Requirement) issued by any Governmental Body related thereto, including any programs or facilities established by the Board of Governors of the Federal Reserve System to which the U.S. Treasury Department has provided financing as contemplated by Title IV of the CARES Act, and any analogous U.S. state, local, or non-US law.

Certificated Shares. "Certificated Shares" is defined in Section 1.6(b) of the Agreement.

Certificates. "Certificates" is defined in Section 1.6(b) of the Agreement.

Change of Control Payment. "Change of Control Payment" is defined in Section 2.9(a)(vii) of the Agreement.

Clinical Trials. "Clinical Trials" is defined in Section 2.12(c) of the Agreement.

Closing. "Closing" is defined in Section 1.3(a) of the Agreement.

Closing Date. "Closing Date" is defined in Section 1.3(a) of the Agreement.

CMO. "CMO" shall mean a contract manufacturing organization.

CMS. "CMS" is defined in Section 5.2(g) of the Agreement.

Code. "Code" shall mean the U.S. Internal Revenue Code of 1986, as amended.

Collaboration Partner. "Collaboration Partner" is defined in Section 2.12(a) of the Agreement.

Company. "Company" is defined in the preamble to the Agreement.

Company 401(k) Plan. "Company 401(k) Plan" is defined in Section 5.3(e) of the Agreement.

Company Adverse Change Recommendation. "Company Adverse Change Recommendation" is defined in Section 5.1(a) of the Agreement.

Company Associate. "Company Associate" shall mean each current or former officer or other employee, or individual who is or was at any time an independent contractor, consultant or director, of or to the Company or its Subsidiaries.

Company Board Recommendation. "Company Board Recommendation" is defined in the Introduction to the Agreement.

Company Common Stock. "Company Common Stock" shall mean the common stock, \$0.001 par value per share, of the Company.

Company Disclosure Schedule. "Company Disclosure Schedule" shall mean the disclosure schedule that has been prepared by the Company in accordance with the requirements of the Agreement and that has been delivered by the Company to Parent on the date of the Agreement.

Company Equity Plans. "Company Equity Plans" shall mean the Company's Amended and Restated 2007 Stock Incentive Plan, 2013 Stock Incentive Plan, and 2019 Inducement Stock Incentive Plan, each as amended.

Company ESPP. "Company ESPP" shall mean the Company's 2016 Employee Stock Purchase Plan, as amended.

Company Financial Advisors. "Company Financial Advisors" is defined in Section 2.24 of the Agreement.

Company IP. "Company IP" shall mean any and all (a) Intellectual Property Rights that are owned or purported to be owned by any Acquired Corporation and (b) third party Intellectual Property Rights exclusively licensed to any of the Acquired Corporations.

Company IT Assets. "Company IT Assets" shall mean computers, computer software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines and all other information technology equipment, and all associated documentation owned by any Acquired Corporation or licensed or leased by any Acquired Corporation (excluding any public networks).

Company Options. "Company Options" shall mean all options to purchase Shares (whether granted pursuant to the Company Equity Plans or otherwise issued or granted).

Company Preferred Stock. "Company Preferred Stock" shall mean the preferred stock, \$0.001 par value per share, of the Company.

Company PSUs. "Company PSUs" shall mean performance vesting restricted stock units with respect to Shares (whether granted pursuant to the Company Equity Plans or otherwise issued or granted).

Company Related Parties. "Company Related Parties" shall mean the Acquired Corporations and any of their respective former, current or future officers, directors, partners, stockholders, managers, members or Affiliates.

Company RSUs. "Company RSUs" shall mean restricted stock units with respect to Shares (whether granted pursuant to the Company Equity Plans or otherwise issued or granted).

Company SEC Documents. "Company SEC Documents" is defined in Section 2.4(a) of the Agreement.

Company Stockholder Approval. "Company Stockholder Approval" is defined in Section 2.21 of the Agreement.

Company Stockholder Meeting. "Company Stockholder Meeting" is defined in Section 4.4(b) of the Agreement.

Confidentiality Agreement. "Confidentiality Agreement" is defined in Section 4.1(a) of the Agreement.

Consent. "Consent" shall mean any approval, consent, ratification, permission, waiver or authorization.

Continuing Employee. "Continuing Employee" is defined in Section 5.3(a) of the Agreement.

Contract. "Contract" shall mean any contract, subcontract, lease, understanding, instrument, loan, credit agreement, bond, debenture, note, option, warrant, license, sublicense, commitment, undertaking or other legally binding agreement.

Converted PSU. "Converted PSU" is defined in Section 1.8(d) of the Agreement.

Copyrights. "Copyrights" is defined in the definition of Intellectual Property Rights.

Credit Agreement. "Credit Agreement" shall mean the Loan and Security Agreement, dated as of July 26, 2022, by and among the Company and each of the Company's Subsidiaries from time to time party thereto, as borrowers, the lenders from time to time party thereto and Hercules Capital, Inc., as administrative agent and collateral agent, as amended from time to time.

CRO. "CRO" shall mean a Person (including a commercial, academic, or other organization) contracted by a sponsor to perform one or more of a sponsor's Clinical Trial-related duties and functions, including those defined in (a) 21 C.F.R. Part 312.3(b); (b) ICH GCP E6; and (c) foreign equivalents of the foregoing, each as may be amended from time to time.

Cut-off Time. "Cut-off Time" is defined in Section 4.3(c) of the Agreement.

DAC 6. "DAC 6" is defined in Section 2.15(g) of the Agreement

Data Privacy Laws. "Data Privacy Laws" shall mean all applicable Legal Requirements governing the Processing of Personal Information.

Delaware Courts. "Delaware Courts" is defined in Section 8.5(a) of the Agreement.

Designated Securities. "Designated Securities" is defined in the definition of Acquisition Proposal.

Determination Notice. "Determination Notice" is defined in Section 5.1(b)(i) of the Agreement.

DGCL. "DGCL" shall mean the Delaware General Corporation Law, as amended.

Dissenting Shares. "<u>Dissenting Shares</u>" is defined in <u>Section 1.7</u> of the Agreement.

DOJ. "DOJ" shall mean the U.S. Department of Justice.

Domain Names. "Domain Names" is defined in the definition of Intellectual Property Rights.

Effective Time. "Effective Time" is defined in Section 1.3(b) of the Agreement.

EMA. "EMA" shall mean the European Medicines Agency.

Employee Plan. "Employee Plan" shall mean any (a) employee benefit plan (as defined in Section 3(3) of ERISA, whether or not subject to ERISA, (b) bonus, deferred compensation, incentive compensation, stock purchase, stock option, other equity-based plan, severance pay, termination pay, death and disability benefits, hospitalization, medical, life or other insurance or welfare benefits, flexible benefits, supplemental unemployment benefits, profit-sharing, pension or retirement plan, policy, program, agreement or arrangement, and (c) employment, consulting, severance or similar agreement, and each other employee benefit plan, program, agreement, or arrangement, in each case that is (i) sponsored, maintained, contributed to or required to be contributed to by any Acquired Corporation or any of their ERISA Affiliates for the benefit of any current or former employee or other individual service provider of any Acquired Corporation, (ii) with respect to which any of the Acquired Corporations has any current or contingent liability or (iii) to which any Acquired Corporation is a party.

Employment Practices. "Employment Practices" is defined in Section 2.16(b) of the Agreement.

Encumbrance. "Encumbrance" shall mean any lien, pledge, hypothecation, charge, mortgage, security interest, encumbrance, claim, infringement, option, right of first refusal or first offer, preemptive right, title retention, community property interest, or similar restriction of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the use of, or receipt of any income derived from, any asset, and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

End Date. "End Date" is defined in Section 7.1(b) of the Agreement.

Entity. "Entity" shall mean any corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any company limited by shares, limited liability company or joint stock company), firm, society or other enterprise, association, organization or entity.

Environmental Law. "Environmental Law" shall mean any federal, state, local or foreign Legal Requirement relating to occupational health and safety, pollution or protection of human health, worker health or the environment (including ambient air, surface water, ground water, sediment, land surface or subsurface strata), including any law or regulation relating to emissions, discharges, Releases or threatened Releases of Hazardous Materials, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials.

ERISA. "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

ERISA Affiliate. "ERISA Affiliate" shall means any Person that, together with any of the Acquired Corporations, at any relevant time would be treated as a single employer under Section 414 of the Code.

Exchange Act. "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

Excluded Shares. "Excluded Shares" is defined in Section 1.5(a)(iii) of the Agreement.

Extended End Date. "Extended End Date" is defined in Section 7.1(b)(ii).

FDA. "FDA" shall mean the United States Food and Drug Administration.

FDCA. "FDCA" shall mean the Federal Food, Drug and Cosmetic Act, as amended, and all related rules, regulations and guidelines.

Final Offering. "Final Offering" is defined in Section 5.4 of the Agreement.

FTC. "FTC" shall mean the U.S. Federal Trade Commission.

GAAP. "GAAP" is defined in Section 2.4(b) of the Agreement.

Global Trade Laws. "Global Trade Laws" shall mean all applicable Legal Requirements governing the export, import and provision of goods (including technical data and technology) and services in the jurisdictions in which the Acquired Corporations operate, including (a) the applicable Legal Requirements of the United States governing embargoes, sanctions, and boycotts, under the International Emergency Economic Powers Act (50 U.S.C. § 1701 et seq.), Trading with the Enemy Act (50 U.S.C. App. §§ 1—44), and all rules, regulations and executive orders relating to any of the foregoing, including regulations promulgated by the Office of Foreign Assets Control of the United States Department of the Treasury at 15 C.F.R. Parts 500-599 and by the U.S. Department of State; (b) all Legal Requirements governing the export, re-export, or transfer of goods, software, technology, or technical data, including the Export Administration Act of 1979 (50 U.S.C. App. §§ 2401-2420), the Export Control Reform Act of 2018 (Pub. L. 115-232), the Export Administration Regulations (15 C.F.R. Parts 730 -774), the Arms Export Control Act (22 U.S.C. § 2778), and the International Traffic in Arms Regulations (22 C.F.R. § 120.1 et seq.); (c) the Foreign Trade Regulations (15 C.F.R. Part 30) administered by the Census Bureau; (d) all applicable Legal Requirements governing the importation of products, technology, technical data, and services, including those administered by United States Customs and Border Protection (19 C.F.R. Parts 1-199); (e) the antiboycott laws set forth in section 999 of the Internal Revenue Code, the Department of Treasury Guidelines concerning international boycotts promulgated thereunder, and Part 760 of the EAR; and (f) any other applicable Legal Requirements relating to the export and import activities of the Acquired Corporations.

Good Clinical Practices. "Good Clinical Practices" shall mean, as applicable, those current good clinical practices, standards, and procedures set forth in Legal Requirements, including (a) the FDCA and its applicable implementing regulations at 21 C.F.R. Parts 50, 54, 56 and 312; (b) the International Conference on Harmonization (ICH) guidance titled "Guidance for Industry E6 Good Clinical Practice: Consolidated Guidance" and including related requirements imposed by Regulation No (EU) 536/2014 and Regulation No (EU) 2017/556, Directive 2005/28/EC and, to the extent still applicable to any ongoing Clinical Trials, Directive 2001/20/EC; and (c) other foreign equivalents of the foregoing, in each case, as same may be amended from time to time.

Good Laboratory Practices. "Good Laboratory Practices" shall mean, as applicable, the current good laboratory practices set forth in Legal Requirements, including (a) the FDCA and its applicable implementing regulations at 21 C.F.R. Part 58; (b) Directive 2004/10/EC of the European Parliament and of the Council of 11 February 2004 on the harmonization of laws, regulations and administrative provisions relating to the application of the principles of good laboratory practice and the verification of their applications for tests on chemical substances (codified version) and Directive 2004/9/EC of the European Parliament and of the Council of 11 February 2004 on the inspection and verification of good laboratory practice (GLP) (codified version); and (c) other foreign equivalents of the foregoing, in each case, as same may be amended from time to time.

Good Manufacturing Practices. "Good Manufacturing Practices" shall mean, as applicable, those current good manufacturing practices related to the manufacture of pharmaceutical products and any precursors thereto set forth in Legal Requirements, including (a) the FDCA and 21 C.F.R. Parts 210-211, (b) guidelines and regulations of standard compilations in Directive (EU) 2017/1572 of 15 September 2017 supplementing Directive 2001/83/EC of the European Parliament and of the Council as regards the principles and guidelines of good manufacturing practice for medicinal products for human use; and (c) other foreign equivalents of the foregoing, in each case, as same may be amended from time to time.

Government Contract. "Government Contract" shall mean any Contract, including any subsequent modification of such Contract, between an Acquired Corporation and (a) a Governmental Body, (b) any prime contractor to a Governmental Body where the Contract is in furtherance of a prime contract between that contractor and a Governmental Body or where an Acquired Corporation otherwise has knowledge that the prime contractor is acting in its capacity as such, or (c) any subcontractor (of any tier) in connection with or with respect to any Contract described in clause (a) or (b) where the Contract is in furtherance of a higher tier contract that is in furtherance of a Contract with a Governmental Body (either as a prime contract or subcontract) or where an Acquired Corporation otherwise has knowledge that the subcontractor is in furtherance of a higher tier contract that is in furtherance of a Contract with a Governmental Body (either as a prime contract or subcontract).

Governmental Authorization. "Governmental Authorization" shall mean any: permit, license, certificate, franchise, grant, permission, clearance, registration, consent, approval, qualification or authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Body or pursuant to any Legal Requirement, including all Regulatory Permits, each as amended or supplemented from time to time.

Governmental Body. "Governmental Body" shall mean any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) supranational, international, multinational, EU, federal, state, local, municipal, foreign or other government; or (c) governmental or quasi-governmental authority of any nature including any governmental board, branch, bureau, division, department, institution, agency, commission, instrumentality, official, ministry, fund, foundation, center, organization, unit or body and any court, arbitrator or other tribunal. "Governmental Body" shall also include any officials, agents, employees or representatives acting on behalf of any of the foregoing.

Guaranteed Obligations. "Guaranteed Obligations" is defined in Section 8.10(b) of the Agreement.

Guarantor. "Guarantor" is defined in the preamble to the Agreement.

GxP. "GxP" shall mean, collectively, Good Clinical Practices, Good Laboratory Practices, Good Manufacturing Practices and other applicable, generally accepted industry best practice standards for the pharmaceutical or biotech industry.

Hazardous Materials. "<u>Hazardous Materials</u>" shall mean any waste, material, or substance that is listed, regulated or defined under any Environmental Law and includes any pollutant, chemical substance, hazardous substance, hazardous waste, special waste, solid waste, asbestos, mold, radioactive material, polychlorinated biphenyls, petroleum or petroleum-derived substance or waste.

Health Care Laws, "Health Care Laws" shall mean all Legal Requirements and guidances that regulate pharmaceuticals, biologics and other medical products, including those related to the development, manufacturing, sale, distribution and promotional activities, product quality, the conduct of Clinical Trials, reporting of safety events, conflicts of interest, debarment, exclusion or disqualification, and interactions with and licensure and accreditation of health care professionals, including the following: (a) the FDCA; (b) the PHSA; (c) the Clinical Laboratory Improvement Amendments (42 U.S.C. § 263a) and the regulations set forth in 42 C.F.R. Part 493; (d) Medicare (Title XVIII of the Social Security Act) and Medicaid (Title XIX of the Social Security Act); (e) the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), the Stark Anti-Self-Referral Law (42 U.S.C. § 1395nn), the Civil Monetary Penalties Law (42 U.S.C. § 1320a-7a and 1320a-7b), the civil False Claims Act (31 U.S.C. §§ 3729 et seq.), the administrative False Claims Law (42 U.S.C. § 1320a-7b(a)), the exclusion laws (42 U.S.C. § 1320a-7), the federal Physician Payments Sunshine Act (42 U.S.C. § 1320a-7h), the regulations promulgated under any of the foregoing, and any similar or equivalent state Legal Requirements; (f) the Patient Protection and Affordable Care Act (Public Law No. 111-148), as amended by the Health Care and Education Reconciliation Act of 2010 (Public Law No. 111-152); (g) 45 C.F.R. Parts 46 and 21 C.F.R. Parts 312, 812, 50, 54 and 56 and state research regulations; (h) the FDA software validation principles; (i) the regulations set forth at 21 C.F.R. Part 11; (j) the Federal Controlled Substances Act, 21 U.S.C. § 801 et seq.; (k) the Health Insurance Portability and Accountability Act of 1996, the Health Information and Technology for Economic and Clinical Health Act, and the regulations promulgated thereunder; (1) EU Directive 2001/83/EC (the Community code relating to medicinal products for human use), Regulation (EC) No 726/2004, Regulation (EC) No 141/2000, Regulation (EC) No 1901/2006, Regulation (EC) No 1394/2007, each as amended, Directive 2005/28/EC, Regulation No (EU) 536/2014 and Regulation No (EU) 2017/556 and EU Directive 2001/20/EC to the extent still applicable to any ongoing Clinical Trials; (m) GxP; and (n) similar or equivalent Legal Requirements of all applicable jurisdictions.

HSR Act. "HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

Identified Communications. "Identified Communications" is defined in Section 5.2(d) of the Agreement.

In-bound License. "In-bound License" is defined in Section 2.8(f) of the Agreement.

Indebtedness. "Indebtedness" shall mean, without duplication, (i) any indebtedness for borrowed money (including the issuance of any debt security) to any Person, including that portion of obligations with respect to any capital leases that is classified as a liability on a balance sheet in conformity with GAAP, (ii) any obligations evidenced by notes, bonds, debentures or similar Contracts for indebtedness for borrowed money owing to any Person (other than an Acquired Corporation), (iii) any reimbursement or other payment obligations in respect of letters of credit and bankers' acceptances used as security for leases), bank guarantees, surety bonds and similar instruments, regardless of whether drawn upon, including the principal, interest and fees owing thereon, (iv) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to property acquired, (v) any net obligations under swaps, options, derivatives and other hedging agreements or arrangements that would be payable upon termination thereof (assuming they were terminated on the date of determination), (vi) all obligations representing the deferred and unpaid purchase price of property or services, and (vii) any guaranty (or any other arrangement having the economic effect of a guaranty) of any such obligations described in clauses (i) through (vi) of any Person other than an Acquired Corporation (other than, in any case, accounts payable to trade creditors and accrued expenses, in each case, arising in the ordinary course of business consistent with past practice that are not past due).

Indemnified Persons. "Indemnified Persons" is defined in Section 5.5(a) of the Agreement.

Initial End Date. "Initial End Date" is defined in Section 7.1(b)(i).

Intellectual Property Rights. "Intellectual Property Rights" shall mean any and all intellectual property and industrial property rights of every kind and description throughout the world, including all U.S. and foreign (a) patents, patent applications, invention disclosures, and all related continuations, continuations-in-part, divisionals, reissues, re-examinations, substitutions, and extensions thereof ("Patents"), (b) trademarks, service marks, names, corporate names, trade names, Internet domain names, logos, slogans, trade dress, design rights, and other similar designations of source or origin (whether registered, common law, statutory or otherwise), together with the goodwill symbolized by any of the foregoing and any applications and registrations for the foregoing ("Copyrights"), (d) trade secrets and confidential ideas, know-how, inventions, proprietary processes, formulae, models, data (including pharmacological, toxicological, non-clinical, pre-clinical and clinical data, analytical and quality control data, manufacturing data and descriptions, market data, financial data or descriptions), databases, data collections, data sets, curated data content, data layers, devices, assays, specifications, physical, chemical and biological materials and compounds, compound libraries, methodologies, and the like, in written, electronic, oral or other tangible or intangible form, whether or not patentable ("Trade Secrets"), (e) Internet domain names and social media accounts ("Domain Names"), and (f) all causes of action and rights to sue or seek other remedies arising from or relating to the foregoing.

In the Money Option. "In the Money Option" is defined in Section 1.8(a) of the Agreement.

Intervening Event. "Intervening Event" shall mean any event, development, occurrence, circumstance, change or effect occurring after the date of this Agreement that has a material positive effect on the financial condition or results of operations of the Acquired Corporations (taken as a whole): (a) that was not known to the Board of Directors prior to the date of this Agreement, and the material consequences of which (based on facts known to members of the Board of Directors as of the date of this Agreement) were not reasonably foreseeable as of the date of this Agreement but become known to the Board of Directors prior to the Cut-off Time, and (b) that does not relate to any Acquisition Proposal; provided, however, that any event, development, occurrence, circumstance, change or effect related to any of the following shall not be deemed to constitute or be taken into account in determining whether there is, or would reasonably be expected to be, an Intervening Event: (i) any breach of this Agreement by the Company; (ii) any event, development, occurrence, circumstance, change or effect directly resulting from the announcement (whether or not authorized by the Parties, including any pre signing reports in the press or otherwise, reporting on a potential transaction among the Parties or otherwise relating to the acquisition of the Company) or pendency of this Agreement or the Transactions, including the identity of, or events, developments, occurrences, circumstances, changes or effects relating to, Parent or any of its Affiliates or any communication by Parent or any of its Affiliates regarding plans, proposals or projections with respect to the Acquired Corporations or their employees (including any impact on the relationship of an Acquired Corporation contractual or otherwise, with its customers, suppliers, distributors, vendors, licensors, licensees, lenders, employees or partners); (iii) any change in the trading price or trading volume of the Shares or any change in the Company's credit rating (although for purposes of clarity, any underlying facts, events, changes, developments or set of circumstances, with respect to this <u>clause (iii)</u> relating to or causing such change may be considered, along with the effects or consequences thereof); (iv) the fact that the Company has exceeded or met any projections, forecasts, revenue or earnings predictions or expectations of the Company or any securities analysts for any period ending (or for which revenues or earnings are released) on or after the date hereof (although for purposes of clarity, any underlying facts, events, changes, developments or set of circumstances relating to or causing such material improvement or improvements may be considered, along with the effects or consequences thereof); (v) any event, development, occurrence, circumstance, change or effect arising from any change in, or any compliance with or action taken for the purpose of complying with any change in, any Legal Requirement or GAAP (or interpretations of any Legal Requirement or GAAP) after the date of the Agreement; or (vi) any changes in general economic or political conditions, or in the financial, credit or securities markets in general (including changes in interest rates, exchange rates, stock, bond and/or debt prices).

IRB. "IRB" shall mean any independent body (a review board or a committee, institutional, regional, national, or supranational), constituted of medical professionals and non-medical members, whose responsibility it is to ensure the protection of the rights, safety and well-being of human subjects involved in a trial and to provide public assurance of that protection, by, among other things, reviewing and approving / providing favorable opinion on, a trial protocol, the suitability of investigator(s), facilities, and the methods and material to be used in obtaining and documenting informed consent of the trial subjects and includes any such entity as described in 21 C.F.R. Part 56 or foreign equivalent of the foregoing.

IRS. "IRS" shall mean the U.S. Internal Revenue Service.

knowledge. "knowledge" with respect to an Entity shall mean with respect to any matter in question the actual knowledge, after reasonable inquiry, of such Entity's executive officers.

Leased Real Property. "Leased Real Property" is defined in Section 2.7(b) of the Agreement.

Legal Proceeding. "Legal Proceeding" shall mean any action, suit, charge, complaint, litigation, arbitration, mediation, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination, citation, summons, claim, demand, qui tam action, subpoena, or investigation commenced, brought, conducted or heard (whether formally or informally, whether publicly or privately) by or before, or otherwise involving, any Governmental Body or any arbitrator or arbitration panel.

Legal Requirement. "<u>Legal Requirement</u>" shall mean any national, supranational, EU, federal, state, local, municipal, foreign or other law, statute, constitution, treaty, resolution, ordinance, common law, code, order, edict, decree, rule, regulation, accreditation standard, certification standard, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Body (or under the authority of Nasdaq or another stock exchange).

Material Adverse Effect. "Material Adverse Effect" shall mean any event, development, occurrence, circumstance, change or effect which, individually or when taken together with all other events, developments, occurrences, circumstances, changes or effects which have occurred in the applicable determination period for a Material Adverse Effect, has had or would reasonably be expected to have a material adverse effect on (a) the ability of the Company to (x) perform any of its material obligations under this Agreement required to consummate the Transactions on or before the End Date, or (y) to consummate the Transactions on or before the End Date, or (b) the business, assets, financial condition or results of operations of the Acquired Corporations, taken as a whole; provided, however, that, for purposes of clause (b) above only, none of the following shall be deemed to constitute or be taken into account in determining whether there is, or would reasonably be expected to be, a Material Adverse Effect: (i) any change in the market price or trading volume of the Company's stock or change in the Company's credit ratings; provided that the underlying causes of any such change may be considered in determining whether a Material Adverse Effect has occurred to the extent not otherwise excluded by another exception herein; (ii) any event, development, occurrence, circumstance, change or effect directly resulting from the announcement, pendency or performance of the Transactions; provided that this clause (ii) shall not apply to any representation or warranty (or condition to the consummation of the Merger relating to such representation or warranty) to the extent the purpose of such representation and warranty is to address the consequences resulting from the execution and delivery of this Agreement or the pendency, performance or consummation of the Transactions (including the Merger), including any representations or warranties contained in Section 2.8(j), Section 2.17(i), Section 2.21 or Section 2.23 and the condition set forth in Section 6.2(a) solely as such condition relates to Section 2.8(j), Section 2.17(i), Section 2.21 or Section 2.23; (iii) any event, development, occurrence, circumstance, change or effect generally affecting the industries in which the Acquired Corporations operate or in the economy generally or other general business, financial or market conditions; (iv) any event, development, occurrence, circumstance, change or effect arising from fluctuations in the value of any currency or interest rates; provided that the underlying causes of such event, development, occurrence, circumstance, change or effect may be considered in determining whether a Material Adverse Effect has occurred to the extent not otherwise excluded by another exception herein; (v) any event, development, occurrence, circumstance, change or effect arising from any act of terrorism, war, national or international calamity, natural disaster, acts of god, epidemic, pandemic or any other similar event; (vi) the failure of the Company to meet internal or analysts' expectations or projections; provided that the underlying causes of such failure may be considered in determining whether a Material Adverse Effect has occurred to the extent not otherwise excluded by another exception herein; (vii) any event, development, occurrence, circumstance, change or effect resulting or arising from the identity of, or any facts or circumstances relating to, Parent, Merger Sub or any of their respective Affiliates; or (viii) any event, development, occurrence, circumstance, change or effect arising from any change in, or any compliance with or action taken solely for the purpose of complying with any change in, any Legal Requirement or GAAP (or interpretations of any Legal Requirement or GAAP) after the date of the Agreement; provided that any event, development, occurrence, circumstance, change or effect referred to in the foregoing clauses (iii), (iv), (v) and (viii) may be taken into account in determining whether there is, or would be reasonably expected to be, a Material Adverse Effect to the extent such event, development, occurrence, circumstance, change or effect disproportionately affects the Acquired Corporations relative to other participants in the industries in which the Acquired Corporations operate.

Material Contract. "Material Contract" is defined in Section 2.9(a) of the Agreement.

Maximum Amount. "Maximum Amount" is defined in Section 5.5(b) of the Agreement.

Merger. "Merger" is defined in the Introduction to the Agreement.

Merger Consideration. "Merger Consideration" is defined in the Introduction to the Agreement.

Merger Proxy Statement. "Merger Proxy Statement" is defined in Section 4.4(a) of the Agreement.

Merger Sub. "Merger Sub" is defined in the preamble to the Agreement.

Nasdaq. "Nasdaq" shall mean The Nasdaq Global Select Market.

Negotiation Period. "Negotiation Period" shall mean the period (a) beginning on the date that Merger Sub receives a Determination Notice with respect to a Superior Offer in accordance with Section 5.1(b)(i) or an Intervening Event in accordance with Section 5.1(b)(ii) and (b) ending at 11:59 p.m. Eastern Time on the fifth (5th) business day after such date; provided that with respect to any Determination Notice given in respect of a material amendment to a Superior Offer for which a Determination Notice was previously given, the Negotiation Period shall end on the later of (x) the date and time on which the Negotiation Period was initially scheduled to end in respect of the initial Determination Notice with respect to such Superior Offer and (y) 11:59 p.m. Eastern Time on the third (3rd) business day after the date that Merger Sub received the Determination Notice with respect to the material amendment to such Superior Offer.

Out-bound License. "Out-bound License" is defined in Section 2.8(f) of the Agreement.

Parent. "Parent" is defined in the preamble to the Agreement.

Parent 401(k) Plan. "Parent 401(k) Plan" is defined in Section 5.3(e) of the Agreement.

Parent Material Adverse Effect. "Parent Material Adverse Effect" shall mean any effect, change, event or occurrence that would or would reasonably be expected to, individually or in the aggregate, have a material adverse effect on (a) the ability of Guarantor to perform any of its material obligations under Section 8.10(b) required to consummate the Transactions on or before the End Date, or (b) the ability of Parent or Merger Sub to (x) perform any of its material obligations under this Agreement required to consummate the Transactions on or before the End Date, or (y) to consummate the Transactions on or before the End Date.

Parties. "Parties" shall mean Parent, Merger Sub, and the Company.

Patient Assistance Program. "Patient Assistance Program" is defined in Section 4.2(b)(xviii) of the Agreement.

Paying Agent. "Paying Agent" is defined in Section 1.6(a) of the Agreement.

Payment Fund. "Payment Fund" is defined in Section 1.6(a) of the Agreement.

Payoff Letter. "Payoff Letter" is defined in Section 5.11 of the Agreement.

Permitted Encumbrance: "Permitted Encumbrance" shall mean (a) any statutory liens for Taxes (i) that are not due and payable or (ii) the validity of which is being contested in good faith by appropriate proceedings and, in each case, for which adequate reserves have been established in the applicable financial statements in accordance with GAAP, (b) mechanics', materialmen's, carriers', workmen's, warehouseman's, repairmen's, landlords' and similar liens granted, or which arise, in the ordinary course of business consistent with past practice as to which there is no default, (c) with respect to leased property, any interest in title of the lessor under the applicable lease (other than a capital lease) entered into by the Company or its Subsidiaries in the ordinary course of business and under which there is no default by the Company or its Subsidiaries, (d) non-exclusive licenses of Intellectual Property Rights entered in the ordinary course of business consistent with past practice, (e) in the case of real property, Encumbrances that are easements, rights-of-way, encroachments, restrictions, conditions and other similar Encumbrances incurred or suffered in the ordinary course of business consistent with past practice and which, individually or in the aggregate, do not and would not materially impair the use (or contemplated use), utility or value of the applicable real property or otherwise materially impair the present or contemplated business operations at such location, or zoning, entitlement, building and other land use regulations imposed by Governmental Bodies having jurisdiction over such real property or that are otherwise set forth on a title report and (f) Encumbrances securing the obligations under the Credit Agreement.

Person. "Person" shall mean any individual, Entity or Governmental Body.

Personal Information. "Personal Information" shall mean any information that (a) identifies or could reasonably be used, alone or in combination with other information held by the Company or any of its Subsidiaries, to identify an individual, or (b) is subject to any Legal Requirement relating to privacy, or (c) "personally identifiable information," "personal data," or similarly defined personal information under Data Privacy Laws, or (d) as defined in any legally binding internal or publicly available privacy policies or notices of the Company or any other Acquired Corporation.

PFIC. "PFIC" is defined in Section 2.15(o) of the Agreement.

PHSA. "PHSA" shall mean the United States Public Health Services Act (42 U.S.C.§ 262 et seq.) and the regulations promulgated thereunder, including 21 C.F.R. Part 600-680.

Pre-Closing Period. "Pre-Closing Period" is defined in Section 4.1 of the Agreement.

Privacy and Security Requirements. "Privacy and Security Requirements" is defined in Section 2.8(k).

Processing. "Processing" shall mean with respect to Personal Information, the use, collection, receipt, processing, aggregation, storage, adaption, alteration, transfer (including cross-border transfers), retrieval, disclosure, dissemination, combination, erasure, disposal, destruction, or anonymization of such Personal Information, or any other operation or set of operations that is performed on data or on sets of data, in each case, whether or not by automated means, and any other form of processing, including as defined by or under any applicable Privacy and Security Requirements.

Product Candidate. "Product Candidate" shall mean each biological, drug, combination product, compound, device or product candidate being developed, tested, labeled, manufactured, stored or marketed by an Acquired Corporation, or regarding which an Acquired Corporation has rights.

Registered IP. "Registered IP" shall mean all Patents, Trademarks and Copyrights that are registered or issued under the authority of any Governmental Body, and all applications for any of the foregoing, and all Domain Name registrations.

Regulatory Filings. "Regulatory Filings" is defined in Section 5.2(b) of the Agreement.

Regulatory Permit. "Regulatory Permit" shall mean any Governmental Authorization required for the development, manufacturing or marketing of a Product Candidate under applicable Health Care Laws, including all investigational new drug applications and other authorizations to conduct Clinical Trials, IRB approvals, new drug applications, supplemental new drug applications, abbreviated new drug applications, biologic license applications, as defined in 21 C.F.R. § 601.2, establishment registrations, as defined in 21 C.F.R. § 207, and product listings, as defined in 21 C.F.R. § 207, all supplements or amendments thereto, and all comparable Governmental Authorizations.

Release. "Release" shall mean any presence, emission, spill, seepage, leak, escape, leaching, discharge, injection, pumping, pouring, emptying, dumping, disposal, migration, or release of Hazardous Materials from any source into or upon the environment, including the air, soil, improvements, surface water, groundwater, the sewer, septic system, storm drain, publicly owned treatment works, or waste treatment, storage, or disposal systems.

Representatives. "Representatives" shall mean officers, directors, employees, attorneys, accountants, investment bankers, consultants, agents, financial advisors, and other advisors and representatives.

Sanctioned Person. "Sanctioned Person" shall mean any Person, aircraft, or vessel that is the subject or target of sanctions or restrictions under the Global Trade Laws, including, any Person: (a) listed on any list of sanctioned persons maintained by the United States, United Nations Security Council, including, but not limited to, (i) the "Specially Designated Nationals and Blocked Persons" list maintained by the U.S. Office of Foreign Assets Control or (ii) the Entity List or Military End User List maintained by the U.S. Department of Commerce's Bureau of Industry and Security; (b) located in, resident in, or incorporated in, a Sanctioned Country; and (c) any Person which is 50% or more owned, directly or indirectly, individually or in the aggregate, or otherwise controlled, by any such Person or Persons described in subclauses (a)(i) or (b) of this definition.

Sanctioned Territory. "Sanctioned Territory" shall mean any country or territory which is itself the subject or target of any country-wide or territory-wide comprehensive economic sanctions imposed by the United States at any point over the previous five years, including, but not limited to, Cuba, Iran, North Korea, Syria, or the Crimea and so-called Donetsk People's Republic and Luhansk People's Republic regions of Ukraine.

Sarbanes-Oxley Act. "Sarbanes-Oxley Act" is defined in Section 2.4(a) of the Agreement.

SEC. "SEC" shall mean the United States Securities and Exchange Commission.

SEC Clearance Date. "SEC Clearance Date" is defined in Section 4.4(b) of the Agreement.

Securities Act. "Securities Act" shall mean the Securities Act of 1933, as amended.

Shares. "Shares" is defined in the Introduction to the Agreement.

Specified Agreement. "Specified Agreement" is defined in Section 7.1(e) of the Agreement.

Specified Antitrust Authority. "Specified Antitrust Authority" is defined in Section 5.2(c) of the Agreement.

Specified Governmental Body. "Specified Governmental Body" is defined in Section 6.1(c) of the Agreement.

Subsidiary. An Entity shall be deemed to be a "<u>Subsidiary</u>" of another Person if such Person directly or indirectly owns, beneficially or of record, (a) an amount of voting securities or other interests in such Entity that is sufficient to enable such Person to elect at least a majority of the members of such Entity's board of directors or other governing body, or (b) at least 50% of the outstanding equity or financial interests of such Entity.

Superior Offer. "Superior Offer" shall mean a *bona fide* written Acquisition Proposal that the Board of Directors determines, in its good faith judgment, after consultation with the Company's outside legal counsel and financial advisors, is reasonably likely to be consummated in accordance with its terms, taking into account all legal, regulatory and financing aspects (including certainty of closing) of the proposal and the Person making the proposal and other aspects of the Acquisition Proposal that the Board of Directors deems relevant, and if consummated, would result in a transaction more favorable to the Company's stockholders (solely in their capacity as such) from a financial point of view than the Transactions (including after giving effect to proposals, if any, made by Parent pursuant to Section 5.1(b)(i)); provided that for purposes of the definition of "Superior Offer," the references to "15%" in the definition Proposal shall be deemed to be references to "75%."

Surviving Corporation. "Surviving Corporation" is defined in the Introduction to the Agreement.

Takeover Laws. "Takeover Laws" shall mean any "moratorium," "control share acquisition," "fair price," "supermajority," "affiliate transactions," or "business combination statute or regulation" or other similar state anti-takeover laws and regulations (including, for the avoidance of doubt, Section 203 of the DGCL).

Tax. "Tax" shall mean (a) any federal, state, local, or non-U.S. or other tax (including any net income tax, gross income tax, franchise tax, capital gains tax, gross receipts tax, gross profits tax, branch profits tax, value-added tax, surtax, estimated tax, employment tax, unemployment tax, national health insurance tax, excise tax, estimated tax, alternative or minimum tax, ad valorem tax, transfer tax, stamp tax, sales tax, use tax, service tax, property tax, escheat or unclaimed property tax, business tax, withholding tax or payroll tax), levy, assessment, tariff, impost, impost, imposition, duty (including any customs duty) or other tax or charge of any kind whatsoever, imposed, assessed or collected by or under the authority of any Governmental Body, together with any interest, penalties, inflationary adjustments, additions to tax, fines or other additional amounts imposed thereon, with respect thereto, or related thereto and (b) any liability in respect of any items described in clause (a) above payable by reason of being or having been a member of any group of corporations that files, will file, or has filed Tax Returns on a combined, consolidated, unitary or similar basis or by reason of any obligation under any contract, assumption, transferee or successor liability, operation of law, Treasury Regulations Section 1.1502-6 (or any predecessor or successor thereof or any analogous or similar provision of state, local or non-U.S. law) or otherwise.

Tax Return. "Tax Return" shall mean any return (including any information return), report, statement, declaration, estimate, schedule, form, election, certificate or other document or information filed or required to be filed with any Governmental Body in connection with the determination, assessment, collection or payment of any Tax and any attachments thereto or amendments thereof.

Termination Fee. "Termination Fee" is defined in Section 7.3(b) of the Agreement.

Trade Secrets. "Trade Secrets" is defined in the definition of Intellectual Property Rights.

Trademarks. "Trademarks" is defined in the definition of Intellectual Property Rights.

Transactions. "Transactions" shall mean (a) the execution and delivery of the Agreement and (b) all of the transactions contemplated by the Agreement, including the Merger.

WARN Act. "WARN Act" is defined in Section 2.16(d) of the Agreement

Workers. "Workers" is defined in Section 2.16(c) of the Agreement.

ANNEX I

FORM OF CERTIFICATE OF INCORPORATION OF THE SURVIVING CORPORATION

CERTIFICATE OF INCORPORATION OF IVERIC BIO, INC.

FIRST

The name of the corporation is IVERIC bio, Inc. (the "Company").

SECOND

The Company's registered office in the State of Delaware is located at 251 Little Falls Drive, in the City of Wilmington, County of New Castle, 19808. Its registered agent at such address is Corporation Service Company.

THIRD

The purpose of the Company is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware, as the same exists or may hereafter be amended ("DGCL") or any successor statute.

FOURTH

The total number of shares of stock that the Company shall have authority to issue is 100 shares of capital stock, all of which shall be designated Common Stock, having a par value of \$0.001.

FIFTH

The following provisions are inserted for the management of the business and the conduct of the affairs of the Company, and for further creation, definition, limitation and regulation of the powers of the Company and its directors, officers and stockholders:

- (1) The business and affairs of the Company shall be managed by or under the direction of the Board of Directors of the Company (the "Board of Directors").
- (2) The Board of Directors is expressly authorized to make, adopt, alter, amend, change, add to or repeal the bylaws of the Company (the "<u>Bylaws</u>"), without any action on the part of the stockholders, but the stockholders may make additional bylaws and may alter, amend or repeal any bylaw whether adopted by them or otherwise. The Company may in its bylaws confer powers upon the Board of Directors in addition to the foregoing and in addition to the powers and authorities expressly conferred upon the Board of Directors by applicable law.

- (3) The number of directors of the Company shall be as from time to time fixed by, or in the manner provided in, the Bylaws. Election of directors need not be by written ballot unless the Bylaws shall so provide.
- (4) No director or officer of the Company shall be personally liable to the Company or any of its stockholders for monetary damages for breach of fiduciary duty as a director or officer, except for liability (i) for any breach of the director or officer's duty of loyalty to the Company or its stockholders, (ii) for a director's or officer's acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) of a director pursuant to Section 174 of the DGCL, (iv) for any transaction from which the director or officer derived an improper personal benefit, (v) of an officer in any action by or in the right of the corporation or (vi) for any act or omission occurring before the effective date of this Article FIFTH. Any amendment, repeal or modification of this Article FIFTH shall not adversely affect any right or protection of such director or officer of the Company existing at the time of such amendment, repeal or modification with respect to acts or omissions occurring prior to such amendment, repeal or modification.
- (5) In addition to the powers and authority hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Company, subject, nevertheless, to the provisions of the DGCL, this Certificate of Incorporation, and any bylaws adopted by the stockholders; provided, however, that no bylaws hereafter adopted by the stockholders shall invalidate any prior act of the directors which would have been valid if such bylaws had not been adopted.

SIXTH

The Company shall provide indemnification as follows:

1. Actions, Suits and Proceedings Other than by or in the Right of the Company. The Company shall indemnify each person who was or is a party or threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Company) by reason of the fact that he or she is or was, or has agreed to become, a director or officer of the Company, or is or was serving, or has agreed to serve, at the request of the Company, as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) (all such persons being referred to hereafter as an "Indemnitee"), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974), and amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee in connection with such action, suit or proceeding and any appeal therefrom, if Indemnitee acted in good faith and in a manner which Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Company, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Company, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his or her conduct was unlawful.

- 2. Actions or Suits by or in the Right of the Company. The Company shall indemnify any Indemnitee who was or is a party to or threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Company to procure a judgment in its favor by reason of the fact that Indemnitee is or was, or has agreed to become, a director or officer of the Company, or is or was serving, or has agreed to serve, at the request of the Company, as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees) and, to the extent permitted by law, amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee in connection with such action, suit or proceeding and any appeal therefrom, if Indemnitee acted in good faith and in a manner which Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Company, except that no indemnification shall be made under this Section 2 in respect of any claim, issue or matter as to which Indemnitee shall have been adjudged to be liable to the Company, unless, and only to the extent, that the Court of Chancery of Delaware or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of such liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnity for such expenses (including attorneys' fees) which the Court of Chancery of Delaware or such other court shall deem proper.
- 3. <u>Indemnification for Expenses of Successful Party.</u> Notwithstanding any other provisions of this Article SIXTH, to the extent that an Indemnitee has been successful, on the merits or otherwise, in defense of any action, suit or proceeding referred to in Sections 1 and 2 of this Article SIXTH, or in defense of any claim, issue or matter therein, or on appeal from any such action, suit or proceeding, Indemnitee shall be indemnified against all expenses (including attorneys' fees) actually and reasonably incurred by or on behalf of Indemnitee in connection therewith. Without limiting the foregoing, if any action, suit or proceeding is disposed of, on the merits or otherwise (including a disposition without prejudice), without (i) the disposition being adverse to Indemnitee, (ii) an adjudication that Indemnitee was liable to the Company, (iii) a plea of guilty or nolo contendere by Indemnitee, (iv) an adjudication that Indemnitee did not act in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company, and (v) with respect to any criminal proceeding, an adjudication that Indemnitee had reasonable cause to believe his or her conduct was unlawful, Indemnitee shall be considered for the purposes hereof to have been wholly successful with respect thereto.
- Notification and Defense of Claim. As a condition precedent to an Indemnitee's right to be indemnified, such Indemnitee must notify the Company in writing as soon as practicable of any action, suit, proceeding or investigation involving such Indemnitee for which indemnity will or could be sought. With respect to any action, suit, proceeding or investigation of which the Company is so notified, the Company will be entitled to participate therein at its own expense and/or to assume the defense thereof at its own expense, with legal counsel reasonably acceptable to Indemnitee. After notice from the Company to Indemnitee of its election so to assume such defense, the Company shall not be liable to Indemnitee for any legal or other expenses subsequently incurred by Indemnitee in connection with such action, suit, proceeding or investigation, other than as provided below in this Section 4. Indemnitee shall have the right to employ his or her own counsel in connection with such action, suit, proceeding or investigation, but the fees and expenses of such counsel incurred after notice from the Company of its assumption of the defense thereof shall be at the expense of Indemnitee unless (i) the employment of counsel by Indemnitee has been authorized by the Company, (ii) counsel to Indemnitee shall have reasonably concluded that there may be a conflict of interest or position on any significant issue between the Company and Indemnitee in the conduct of the defense of such action, suit, proceeding or investigation or (iii) the Company shall not in fact have employed counsel to assume the defense of such action, suit, proceeding or investigation, in each of which cases the fees and expenses of counsel for Indemnitee shall be at the expense of the Company, except as otherwise expressly provided by this Article SIXTH. The Company shall not be entitled, without the consent of Indemnitee, to assume the defense of any claim brought by or in the right of the Company or as to which counsel for Indemnitee shall have reasonably made the conclusion provided for in clause (ii) above. The Company shall not be required to indemnify Indemnitee under this Article SIXTH for any amounts paid in settlement of any action, suit, proceeding or investigation effected without its written consent. The Company shall not settle any action, suit, proceeding or investigation in any manner which would impose any penalty or limitation on Indemnitee without Indemnitee's written consent. Neither the Company nor Indemnitee will unreasonably withhold or delay its consent to any proposed settlement.

- 5. Advancement of Expenses. Subject to the provisions of Section 6 of this Article SIXTH, in the event of any threatened or pending action, suit, proceeding or investigation of which the Company receives notice under this Article SIXTH, any expenses (including attorneys' fees) incurred by or on behalf of Indemnitee in defending an action, suit, proceeding or investigation or any appeal therefrom shall be paid by the Company in advance of the final disposition of such matter; provided, however, that the payment of such expenses incurred by or on behalf of Indemnitee in advance of the final disposition of such matter shall be made only upon receipt of an undertaking by or on behalf of Indemnitee to repay all amounts so advanced in the event that it shall ultimately be determined by final judicial decision from which there is no further right to appeal that Indemnitee is not entitled to be indemnified by the Company as authorized in this Article SIXTH; and provided further that no such advancement of expenses shall be made under this Article SIXTH if it is determined (in the manner described in Section 6) that (i) Indemnitee did not act in good faith and in a manner he or she reasonably believed to be in, or not opposed to, the best interests of the Company, or (ii) with respect to any criminal action or proceeding, Indemnitee had reasonable cause to believe his or her conduct was unlawful. Such undertaking shall be accepted without reference to the financial ability of Indemnitee to make such repayment.
- 6. Procedure for Indemnification and Advancement of Expenses. In order to obtain indemnification or advancement of expenses pursuant to Section 1, 2, 3 or 5 of this Article SIXTH, an Indemnitee shall submit to the Company a written request. Any such advancement of expenses shall be made promptly, and in any event within 60 days after receipt by the Company of the written request of Indemnitee, unless (i) the Company has assumed the defense pursuant to Section 4 of this Article SIXTH (and none of the circumstances described in Section 4 of this Article SIXTH that would nonetheless entitle the Indemnitee to indemnification for the fees and expenses of separate counsel have occurred) or (ii) the Company determines within such 60-day period that Indemnitee did not meet the applicable standard of conduct set forth in Section 1, 2 or 5 of this Article SIXTH, as the case may be. Any such indemnification, unless ordered by a court, shall be made with respect to requests under Section 1 or 2 only as authorized in the specific case upon a determination by the Company that the indemnification of Indemnitee is proper because Indemnitee has met the applicable standard of conduct set forth in Section 1 or 2, as the case may be. Such determination shall be made in each instance (a) by a majority vote of the directors of the Company consisting of persons who are not at that time parties to the action, suit or proceeding in question ("disinterested directors"), whether or not a quorum, (b) by a committee of disinterested directors designated by majority vote of disinterested directors, whether or not a quorum, (c) if there are no disinterested directors, or if the disinterested directors so direct, by independent legal counsel (who may, to the extent permitted by law, be regular legal counsel to the Company) in a written opinion, or (d) by the stockholders of the Company.
- 7. Remedies. The right to indemnification or advancement of expenses as granted by this Article SIXTH shall be enforceable by Indemnitee in any court of competent jurisdiction. Neither the failure of the Company to have made a determination prior to the commencement of such action that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company pursuant to Section 6 of this Article SIXTH that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct. In any suit brought by Indemnitee to enforce a right to indemnification, or brought by the Company to recover an advancement of expenses pursuant to the terms of an undertaking, the Company shall have the burden of proving that Indemnitee is not entitled to be indemnified, or to such advancement of expenses, under this Article SIXTH. Indemnitee's expenses (including attorneys' fees) reasonably incurred in connection with successfully establishing Indemnitee's right to indemnification, in whole or in part, in any such proceeding shall also be indemnified by the Company. Notwithstanding the foregoing, in any suit brought by Indemnitee to enforce a right to indemnification hereunder it shall be a defense that the Indemnitee has not met any applicable standard for indemnification set forth in the DGCL.

- 8. <u>Limitations</u>. Notwithstanding anything to the contrary in this Article SIXTH, except as set forth in Section 7 of this Article SIXTH, the Company shall not indemnify an Indemnitee pursuant to this Article SIXTH in connection with a proceeding (or part thereof) initiated by such Indemnitee unless the initiation thereof was approved by the Board of Directors of the Company. Notwithstanding anything to the contrary in this Article SIXTH, the Company shall not indemnify an Indemnitee to the extent such Indemnitee is reimbursed from the proceeds of insurance, and in the event the Company makes any indemnification payments to an Indemnitee and such Indemnitee is subsequently reimbursed from the proceeds of insurance, such Indemnitee shall promptly refund indemnification payments to the Company to the extent of such insurance reimbursement
- 9. <u>Subsequent Amendment</u>. No amendment, termination or repeal of this Article SIXTH or of the relevant provisions of the DGCL or any other applicable laws shall adversely affect or diminish in any way the rights of any Indemnitee to indemnification under the provisions hereof with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the final adoption of such amendment, termination or repeal.
- 10. Other Rights. The indemnification and advancement of expenses provided by this Article SIXTH shall not be deemed exclusive of any other rights to which an Indemnitee seeking indemnification or advancement of expenses may be entitled under any law (common or statutory), agreement or vote of stockholders or disinterested directors or otherwise, both as to action in Indemnitee's official capacity and as to action in any other capacity while holding office for the Company, and shall continue as to an Indemnitee who has ceased to be a director or officer, and shall inure to the benefit of the estate, heirs, executors and administrators of Indemnitee. Nothing contained in this Article SIXTH shall be deemed to prohibit, and the Company is specifically authorized to enter into, agreements with officers and directors providing indemnification rights and procedures different from those set forth in this Article SIXTH. In addition, the Company may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Company or other persons serving the Company and such rights may be equivalent to, or greater or less than, those set forth in this Article SIXTH.
- 11. Partial Indemnification. If an Indemnitee is entitled under any provision of this Article SIXTH to indemnification by the Company for some or a portion of the expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) or amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee in connection with any action, suit, proceeding or investigation and any appeal therefrom but not, however, for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion of such expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) or amounts paid in settlement to which Indemnitee is entitled.
- 12. <u>Insurance</u>. The Company may purchase and maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Company or another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) against any expense, liability or loss incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Company would have the power to indemnify such person against such expense, liability or loss under the DGCL.

- 13. Savings Clause. If this Article SIXTH or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Company shall nevertheless indemnify each Indemnitee as to any expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) and amounts paid in settlement in connection with any action, suit, proceeding or investigation, whether civil, criminal or administrative, including an action by or in the right of the Company, to the fullest extent permitted by any applicable portion of this Article SIXTH that shall not have been invalidated and to the fullest extent permitted by applicable law.
- 14. <u>Definitions</u>. Terms used herein and defined in Section 145(h) and Section 145(i) of the DGCL shall have the respective meanings assigned to such terms in such Section 145(h) and Section 145(i).

SEVENTH

Meetings of the stockholders may be held within or outside the State of Delaware, as the Bylaws may provide. The books of the Company may be kept (subject to any provision contained in the DGCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or imposed by other provisions of this Certificate of Incorporation, the Bylaws, the DGCL, or other applicable law, or by any contract or agreement to which the Company is or may become a party.

EIGHTH

The Company reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this express reservation.

[Signature Page Follows]

IN WITNESS WHEREOF, this Amended and Res [] day of [], 2023.	stated Certificate of Incorporation has been executed by a duly authorized officer of this corporation this
	IVERIC BIO, INC.
	BY:
	Name: Title:
	[Signature Page to Certificate of Incorporation]

AMENDMENT TO

AMENDED AND RESTATED BY-LAWS

OF

IVERIC BIO, INC.

ARTICLE V

GENERAL PROVISIONS

5.9 Forum for Adjudication of Certain Disputes. Unless the corporation consents in writing to the selection of an alternative forum (an "Alternative Forum Consent"), the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the corporation, (ii) any action asserting a claim of breach of a duty (including any fiduciary duty) owed by any current or former director, officer, stockholder, employee or agent of the corporation to the corporation or the corporation's stockholders, (iii) any action asserting a claim against the corporation or any current or former director, officer, stockholder, employee or agent of the corporation arising out of or relating to any provision of the General Corporation Law of the State of Delaware, the Certificate of Incorporation or these By-laws (each, as in effect from time to time), or (iv) any action asserting a claim against the corporation or any current or former director, officer, stockholder, employee or agent of the corporation governed by the internal affairs doctrine of the State of Delaware; provided, however, that, in the event that the Court of Chancery of the State of Delaware lacks subject matter jurisdiction over any such action or proceeding, the sole and exclusive forum for such action or proceeding shall be another state or federal court located within the State of Delaware, in each such case, unless the Court of Chancery (or such other state or federal court located within the State of Delaware, as applicable) has dismissed a prior action by the same plaintiff asserting the same claims because such court lacked personal jurisdiction over an indispensable party named as a defendant therein. If any action the subject matter of which is within the scope of the preceding sentence is filed in a court other than a court located within the State of Delaware (a "Foreign Action") in the name of any stockholder, such stockholder shall be deemed to have consented to (x) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce the preceding sentence and (y) having service of process made upon such stockholder in any such action by service upon such stockholder's counsel in such Foreign Action as agent for such stockholder. Unless the corporation gives an Alternative Forum Consent, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the sole and exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended. Failure to enforce the foregoing provisions would cause the corporation irreparable harm and the corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions. To the fullest extent permitted by law, any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the corporation shall be deemed to have notice of and consented to the provisions of this Section 5.9. The existence of any prior Alternative Forum Consent shall not act as a waiver of the corporation's ongoing consent right as set forth above in this Section 5.9 with respect to any current or future actions or claims.





Press Release

Astellas Enters into Definitive Agreement to Acquire Iveric Bio

-Companies to Create World-Class Ophthalmology Entity-Lead Program, Avacincaptad Pegol for the Potential Treatment of Geographic Atrophy with PDUFA Goal Date of August 19, 2023-Acquisition advances Astellas'
Primary Focus on "Blindness & Regeneration"-Acquisition price of US\$40 per share in cash,
representing a total equity value of approximately US\$5.9 billion-

Tokyo and Parsippany, New Jersey, April 30, 2023 - Astellas Pharma Inc. (TSE: 4503, President and CEO: Naoki Okamura, "Astellas") and IVERIC bio, Inc. (NASDAQ: ISEE, CEO: Glenn P. Sblendorio, "Iveric Bio") today announced that on April 29, 2023 (Japan time), the Companies have entered into a definitive agreement under which Astellas through Berry Merger Sub, Inc., a wholly-owned subsidiary of Astellas US Holding, Inc., has agreed to acquire 100% of the outstanding shares of Iveric Bio for US\$40.00 per share in cash for a total equity value of approximately US\$5.9 billion (the "Acquisition"). In the Acquisition, Iveric Bio will become an indirectly wholly-owned subsidiary of Astellas. The total equity value of Iveric Bio in the Acquisition assumes that there are approximately 148.2 million outstanding shares of Iveric Bio common stock on a fully diluted basis. The purchase price represents a premium of 64% to Iveric Bio's unaffected closing share price of US\$24.33 on March 31, 2023, and a premium of 75% to Iveric Bio's 30 trading day volume weighted average price as of March 31, 2023. The Boards of Directors of both companies have unanimously approved the transaction.

"We are pleased to reach an agreement with Iveric Bio, a company with exceptional expertise in the R&D of innovative therapeutics in the ophthalmology field." said Naoki Okamura, President and CEO, Astellas. "Iveric Bio has promising programs including Avacincaptad Pegol ("ACP"), an important program for Geographic Atrophy ("GA") secondary to Age-Related Macular Degeneration ("AMD"), and capabilities across the entire value chain in the ophthalmology field. We believe that this acquisition will enable us to deliver greater VALUE to patients with ocular diseases at high risk of blindness."

"This transaction with Astellas, a highly respected pharmaceutical company, demonstrates the significant value that we have built for our stockholders and recognizes the tremendous work by our dedicated team at Iveric Bio," said Glenn P. Sblendorio, Chief Executive Officer of Iveric Bio.

"The opportunity to create a world-class entity with the ophthalmology expertise and capabilities of Iveric Bio and the global reach and resources of Astellas is unique and has the potential to benefit patients worldwide suffering from blinding retinal diseases, including GA," said Pravin U. Dugel, MD, President of Iveric Bio.

1. Strategic Objectives of the Acquisition

Astellas aims to become a cutting-edge, VALUE-driven life science innovator to realize its VISION to be "on the forefront of healthcare change to turn innovative science into VALUE for patients." Through Astellas' R&D strategy, <u>Focus Area Approach</u>, it is working to create innovative drugs for diseases with high unmet medical need by identifying unique combinations of biology and therapeutic modality / technology from multiple perspectives. Currently, Astellas has identified five Primary Focuses, including "Blindness & Regeneration", and is prioritizing investment resources in these areas. As such, this transaction is a key step in building Astellas' product portfolio in this important area.

Iveric Bio focuses on the discovery and development of novel treatments in the field of ophthalmology. The company announced in February 2023 that the U.S. Food and Drug Administration ("FDA") accepted for filing a New Drug Application ("NDA") for ACP for the treatment of GA secondary to AMD. The NDA has been granted priority review with a Prescription Drug User Fee Act ("PDUFA") goal date of August 19, 2023.

ACP, a complement C5 inhibitor, is an investigational drug for GA secondary to AMD and has significant potential to deliver value to a large and underserved patient base. ACP met its primary efficacy endpoint (reduction of the rate of GA progression) with statistical significance across two pivotal clinical trials (GATHER Clinical Trials) and has received breakthrough therapy designation*1 from the FDA for this indication.

Astellas expects that the acquisition of Iveric Bio will not only contribute to Astellas' FY2025 revenue targets set in its Corporate Strategic Plan 2021, but also, that ACP in conjunction with fezolinetant and PADCEV, is anticipated to be a revenue-generating pillar to help compensate for the decline in sales of XTANDI due to anticipated patent expiration later this decade.

In addition, the acquisition of Iveric Bio will provide a foundation of ophthalmology focused capabilities, including a multi-faceted commercial team, expansive network of experts in the ophthalmology field, established relationships with medical institutions, and the infrastructure and experience to drive our combined ophthalmology business going forward. Furthermore, through acquired capabilities, Astellas will accelerate pre-clinical and clinical development and commercialization activities to positively contribute to the goals of Primary Focus, "Blindness & Regeneration."

Funds for the acquisition consist of newly procured funds from bank loans and issuing of commercial paper totaling approximately 800 billion yen and existing cash on hand. Astellas expects to repay this debt within the next five to seven years. The completion of the Acquisition is not subject to a financing condition. Astellas does not anticipate making any change in its dividend policy following the Acquisition.

The closing of the proposed Acquisition is subject to approval by Iveric Bio's stockholders and other customary closing conditions, including receipt of required regulatory approvals. The companies expect to complete the Acquisition in the second quarter of Astellas' fiscal year 2023 (third calendar quarter of 2023). A copy of the merger agreement regarding the proposed Acquisition will be filed with the U.S. Securities and Exchange Commission ("SEC") and will be publicly available on the SEC's website at (http://www.sec.gov).

2. Overview of Iveric Bio

① Company		IVERIC bio, Inc.				
② Address		8 Sylvan Way Parsippany, NJ 07054, US				
③ Representative's Title and Name		CEO, Glenn P. Sblendorio				
Business Description		R&D of pharmaceuticals				
⑤ Stated Capital		US\$ 137 thousand (as of December 31, 2022)				
© Consolidated Stockholders' equity		US\$ 534,657 thousand (as of December 31, 2022)				
① Year of Establishment		2007				
® Number of Shares Outstanding		137,616,082 (as of April 27, 2023)				
Major shareholders and ownership ratio (*)		Vanguard Group, Inc.: 7.4% BlackRock, Inc.: 7.1% Deep Track Capital, LP: 5.7%				
Relationship between	n Astellas and Iveric Bio					
	Capital Relationship:		There is no capital relationship between Astellas and Iveric Bio required to be disclosed.			
	Personal Relationship: Business Relationship: Status of A Related Party		There is no personal relationship between Astellas and Iveric Bio required to be disclosed.			
			There is no business relationship between Astellas and Iveric Bio required to be disclosed.			
			Iveric Bio is not an affiliated party of Astellas.			
① Iveric Bio's consolic	lated operating results and	consolidated fi	nancial position	n for the past three years (**)		
Accounting Period (in thousands except per share data)		Fiscal year ended December, 2020		Fiscal year ended December, 2021	Fiscal year ended December, 2022	
Total Equity			191,563	360,528	534,657	
Total Assets			216,754	389,358	666,823	
Equity per share (US \$)		2.58		3.54	4.42	
Revenue (***)		-	-	-		
Operating Loss		(88,736)	(114,757)	(189,906)		
Loss before taxes		(88,242)	(114,522)	(185,211)		
Net Loss			(84,547)	(114,522)	(185,211)	
Net Loss per share (US \$)		(1.14)	(1.12)	(1.53)		
Dividend per share (US \$)		-	-	-		

^{*} Includes holdings of its subsidiaries and affiliates; based on information from Iveric Bio's definitive proxy statement filed on April 5, 2023, with the SEC in connection with Iveric Bio's planned annual stockholder meeting for 2023

^{**} Excerpt from Iveric Bio's Annual Report on Form 10-K for the fiscal year ended December 31, 2022, filed by Iveric Bio with the SEC on March 1, 2023

^{***} There were no sales since these were before the product launch

3. Number of Iveric Bio shares to be acquired by Astellas, acquisition price, and status of shareholdings before and after the Acquisition

① Number of Iveric Bio shares held by Astellas before	0 (as of April 27, 2023)
the Acquisition	(Number of voting rights: 0)
	(Ownership ratio of voting rights: 0%)
② Number of Iveric Bio Shares Outstanding as of	137,616,082 shares
April 27, 2023	
3 Acquisition amount (****)	US\$40 per share (approximately US\$5.9 billion in the aggregate)
Percentage of Iveric Bio shares to be held after the	100%
Acquisition	(Number of voting rights: 100%)
	(Ownership ratio of voting rights: 100%)

^{****} Acquisition amount includes the full amount required to purchase all outstanding options and restricted stock units

4 Schedule

① Date of Definitive Agreement	April 29, 2023 (Japan time)
② Estimated Date of stockholder meeting to be held by Iveric Bio	second quarter of Astellas' fiscal year 2023 (third calendar quarter of 2023)
③ Estimated Date of Closing (****)	second quarter of Astellas' fiscal year 2023 (third calendar quarter of 2023)

^{****} Subject to applicable regulatory approvals

5. Financial Impact of the Acquisition

The impact of the consummation of the Acquisition on Astellas' financial results is not reflected in Astellas' consolidated financial forecasts for the fiscal year ending March 31, 2024, that were announced on April 27, 2023. Astellas is still reviewing the impact and will promptly announce any events that are to be publicly reported.

6. Financial and Legal Advisors

J.P. Morgan Securities LLC, acting through its affiliate JPMorgan Securities Japan Co., Ltd., is acting as exclusive financial advisor to Astellas and Jones Day is acting as Astellas' legal counsel. BofA Securities, Inc. and Centerview Partners LLC are serving as Iveric Bio's exclusive financial advisors and Skadden, Arps, Slate, Meagher & Flom LLP is serving as Iveric Bio's legal counsel, with Wilmer Cutler Pickering Hale and Dorr LLP advising on general corporate and licensing matters.

*1: The FDA's breakthrough therapy designation is intended to expedite the development and review of drugs for serious or life-threatening conditions. The criteria for breakthrough therapy designation require preliminary clinical evidence that demonstrates the drug may have substantial improvement on at least one clinically significant endpoint over available therapy. FDA will review the full data submitted to support approval of drugs designated as breakthrough therapies to determine whether the drugs are safe and effective for their intended use before they are approved for marketing.

About Astellas' Primary Focus "Blindness & Regeneration"

The Primary Focus' mission is to identify, develop and deliver next generation treatments to restore sight for patients with retinal diseases. Astellas is passionate about R&D to increase productivity and creativity and deliver innovative treatments to patients as quickly as possible by combining optimal internal and external capabilities. For more information, please visit our website at https://www.astellas.com/en/innovation/primary-focus-blindness-regeneration.

About Geographic Atrophy (GA) secondary to Age-related Macular Degeneration (AMD)

Age-related macular degeneration (AMD) is the major cause of moderate and severe loss of central vision in aging adults, affecting both eyes in the majority of patients. The macula is a small area in the central portion of the retina responsible for central vision. As AMD progresses, the loss of retinal cells and the underlying blood vessels in the macula results in marked thinning and/or atrophy of retinal tissue. Geographic atrophy secondary to AMD, leads to irreversible loss of vision in patients and has a high unmet medical need. It is estimated that approximately 1.6 million people in the U.S. have GA in at least one eye¹.

About Avacincaptad Pegol (ACP)

Avacincaptad pegol (ACP) is an investigational drug that is currently under evaluation for safety and efficacy by the U.S. FDA. ACP is a novel complement C5 protein inhibitor. Overactivity of the complement system and the C5 protein are suspected to play a critical role in the development and growth of scarring and vision loss associated with geographic atrophy (GA) secondary to age-related macular degeneration (AMD). By targeting C5, ACP has the potential to decrease activity of the complement system that causes the degeneration of retinal cells and potentially slow the progression of GA.

About GATHER Clinical Trials

ACP met its primary endpoint in the completed GATHER1 clinical trial and the ongoing GATHER2 clinical trial both of which are randomized, double-masked, sham-controlled, multicenter Phase 3 clinical trials. These clinical trials evaluated the safety and efficacy of monthly 2 mg intravitreal administration of ACP in patients with GA secondary to AMD. For the first 12 months in both trials, patients were randomized to receive either ACP 2 mg or sham monthly. There were 286 participants enrolled in GATHER1 and 448 participants enrolled in GATHER2. The primary efficacy endpoints in both pivotal studies were based on GA area measured by fundus autofluorescence at three time points: Baseline, Month 6, and Month 12. The mean rate of growth (slope) in GA area from baseline to month 12 using observed data was 35% in GATHER 1 and 18% in GATHER2. In GATHER1 and GATHER2 combined, the most frequently reported treatment emergent adverse events in the 2 mg recommended dose were related to injection procedure. The most common adverse reactions (≥ 5% and greater than sham) reported in patients who received avacincaptad pegol 2 mg were conjunctival hemorrhage (13%), increased IOP (9%), and CNV (7%). After 18 months of treatment in GATHER1 and 12 months of treatment in GATHER2, there were no events of serious intraocular inflammation, vasculitis, or endophthalmitis.

About Astellas

Astellas Pharma Inc. is a pharmaceutical company conducting business in more than 70 countries around the world. We are promoting the Focus Area Approach that is designed to identify opportunities for the continuous creation of new drugs to address diseases with high unmet medical needs by focusing on Biology and Modality. Furthermore, we are also looking beyond our foundational Rx focus to create Rx^{+} healthcare solutions that combine our expertise and knowledge with cutting-edge technology in different fields of external partners. Through these efforts, Astellas stands on the forefront of healthcare change to turn innovative science into VALUE for patients. For more information, please visit our website at https://www.astellas.com/en.

About Iveric Bio

Iveric Bio is a science-driven biopharmaceutical company focused on the discovery and development of novel treatments for retinal diseases with significant unmet medical needs. The Company is committed to having a positive impact on patients' lives by delivering high-quality, safe, and effective treatments designed to address debilitating retinal diseases including earlier stages of age-related macular degeneration. For more information on the Company, please visit www.ivericbio.com.

Cautionary Notice Regarding Forward-Looking Statements

All statements in this press release, other than statements of historical fact, are statements that could be deemed "forward-looking statements." In some cases, forward-looking statements may be identified by terminology such as "believe," "may," "will," "should", "predict", "goal", "strategy", "potentially," "estimate," "continue," "anticipate," "intend," "could," "would," "project," "plan," "expect," "seek" and similar expressions and variations thereof. Iveric Bio intends these forward-looking statements to be covered by the safe harbor provisions for forward-looking statements in the U.S. Private Securities Litigation Reform Act of 1995.

This press release contains "forward-looking statements" relating to, among other things, the proposed acquisition of Iveric Bio by Astellas and the objectives of such proposed acquisition, Astellas' and Iveric Bio's beliefs and expectations regarding the potential benefits sought to be achieved by Astellas' proposed acquisition of Iveric Bio, the potential effects of the proposed acquisition on both Astellas and Iveric Bio, the expected benefits and success of Iveric Bio's product candidates, the potential for and anticipated timing for approval of ACP, the anticipated financing of the proposed acquisition, and the anticipated timing of completion of the proposed acquisition, each of which involves substantial risks and uncertainties that could cause actual results to differ materially from those expressed or implied by such statements.

Risks and uncertainties include, among other things, risks related to the ability of Iveric Bio and Astellas to complete the transactions contemplated by the merger agreement; the satisfaction or waiver of the conditions to closing the proposed acquisition set forth in the merger agreement (including the failure to obtain necessary regulatory approvals and failure to obtain the requisite vote by Iveric Bio stockholders) in the anticipated timeframe or at all, including the possibility that the proposed acquisition does not close; the timing and nature of regulatory filings for Iveric Bio's product candidates, and the possibility of a termination of the merger agreement; the possibility that competing offers to acquire Iveric Bio may be made; risks related to the ability to realize the anticipated benefits of the proposed acquisition, including the possibility that the expected benefits from the acquisition will not be realized or will not be realized within the expected time period; the risk that Iveric Bio's business and products will not be integrated with those of Astellas successfully; the effects of disruption from the transactions contemplated by the merger agreement on Iveric Bio's business and the fact that the announcement and pendency of the transactions may make it more difficult to establish or maintain relationships with employees, suppliers and other business partners; negative effects of this announcement or the consummation of the proposed acquisition on the market price of Astellas' or Iveric Bio's common stock and/or operating results; significant transaction costs; unknown liabilities; the risk of litigation and/or regulatory actions related to the proposed acquisition or Iveric Bio's business; risks related to the financing of the acquisition; other business effects and uncertainties, including the effects of industry, market, business, economic, political or regulatory conditions; future exchange and interest rates; changes in tax and other laws, regulations, rates and policies; future business combinations or disposals; the uncertainties inherent in research and development, including the ability to meet anticipated clinical endpoints, commencement and/or completion dates for clinical trials, regulatory submission dates, regulatory approval dates and/or launch dates, as well as the possibility of unfavorable new clinical data and further analyses of existing clinical data; risks associated with interim data; the risk that clinical trial data is subject to differing interpretations and assessments by regulatory authorities; whether regulatory authorities will be satisfied with the design of and results from the clinical studies; whether and when drug applications may be filed in any jurisdictions for Iveric Bio's pipeline products; whether and when any such applications may be approved by regulatory authorities, which will depend on myriad factors, including making a determination as to whether the product's benefits outweigh its known risks and determination of the product's efficacy and, if approved, whether any such products will be commercially successful; decisions by regulatory authorities impacting labeling, manufacturing processes, safety or other matters that could affect the availability or commercial potential of such products; expectations regarding personnel and human capital matters; and competitive developments.

Moreover, Astellas and Iveric Bio operate in very competitive and rapidly changing environments, and new risks emerge from time to time. Astellas and Iveric Bio have based these forward-looking statements on their current expectations and projections about future events and trends that they believe may affect the financial condition, results of operations, business strategy, short-term and long-term business operations and objectives and financial needs of Astellas and Iveric Bio, but they cannot guarantee future events, results, actions, levels of activity, performance or achievements, business and market conditions, the timing and results of biotechnology development and potential regulatory approval. The foregoing factors are not exhaustive. You should also carefully consider other risks and uncertainties that may affect the business of Iveric Bio, including those described in the "Forward-Looking Statements", "Summary of Principal Risk Factors", and "Risk Factors" sections of Iveric Bio's Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and other documents filed from time to time with the SEC, all of which are available on the SEC's website at www.sec.gov. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements. Forward-looking statements speak only as of the date they are made. Readers are cautioned not to put undue reliance on forward-looking statements and Astellas and Iveric Bio assume no obligation to, and do not intend to, update or revise these forward-looking statements, whether as a result of new information, future events, or otherwise, unless required by applicable law.

Additional Information and Where to Find It

In connection with the proposed acquisition, Iveric Bio will be filing documents with the SEC, including preliminary and definitive proxy statements relating to the proposed acquisition. This press release is not a substitute for the proxy statement or any other document which Iveric Bio may file with the SEC. The definitive proxy statement will be mailed to Iveric Bio's stockholders in connection with the proposed acquisition. BEFORE MAKING ANY VOTING DECISION, IVERIC BIO'S INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE PRELIMINARY AND DEFINITIVE PROXY STATEMENTS AND ANY OTHER DOCUMENTS TO BE FILED WITH THE SEC IN CONNECTION WITH THE PROPOSED TRANSACTION OR INCORPORATED BY REFERENCE IN THE PROXY STATEMENT WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED ACQUISITION. Any vote in respect of resolutions to be proposed at Iveric Bio's stockholder meeting to approve the proposed transaction or other responses in relation to the proposed transaction should be made only on the basis of the information contained in Iveric Bio's proxy statement. Investors and security holders may obtain free copies of these documents (when they are available) and other related documents filed with the SEC at the SEC's web site at www.sec.gov, and all documents filed by Iveric Bio with the SEC are available to all stockholders of Iveric Bio free of charge at https://investors.ivericbio.com/financial-information/sec-filings.

Participants in the Solicitation

Iveric Bio, and its directors, executive officers and other members of management and certain other people may be deemed to be participants in the solicitation of proxies in connection with the proposed acquisition. Information about Iveric Bio's directors and executive officers is included in the proxy statement for Iveric Bio's annual meeting of stockholders for 2023, filed with the SEC on April 5, 2023. Additional information regarding these persons and their interests in the merger will be included in the proxy statement relating to the proposed acquisition when it is filed with the SEC. These documents, when available, can be obtained free of charge from the sources indicated above.

Important Additional Information

This communication is for informational purposes only and is not intended to and does not constitute, or form part of, an offer, invitation or the solicitation of an offer or invitation to purchase, otherwise acquire, subscribe for, sell or otherwise dispose of Iveric Bio common stock or any other securities, or the solicitation of any vote or approval in any jurisdiction, pursuant to the proposed acquisition or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law.

###

ISEE-G

Contacts for inquiries or additional information:

Astellas Pharma Inc. Corporate Advocacy & Relations +81-3-3244-3201

Iveric Bio

Investor Contact:

Kathy Galante Senior Vice President, Investor Relations <u>kathy.galante@ivericbio.com</u>

or

Media Contact:

Jeannie Neufeld Senior Director, Public Relations & Communications <u>jeannie.neufeld@ivericbio.com</u>

Reference

¹ Klein, et al. JAMA Ophthalmology. 2011.